# RAMCO AVIATION SOLUTION VERSION 5.9 USER GUIDE MAINTENANCE PROGRAMS

©2021 Ramco Systems Limited. All rights reserved. All trademarks acknowledged.

This document is published by Ramco Systems Ltd. without any warranty. No part of this document may be reproduced or transmitted in any form or by any means, electronic or mechanical, for any purpose without the written permission of Ramco Systems Limited.

Improvements and changes to this text necessitated by typographical errors, inaccuracies of current information or improvements to software programs and/or equipment, may be made by Ramco Systems Limited, at any time and without notice. Such changes will, however, be incorporated into new editions of this document. Any hard copies of this document are to be regarded as temporary reference copies only.

The documentation has been provided for the entire Aviation solution, although only a part of the entire solution may be deployed at the customer site, in accordance with the license agreement between the customer and Ramco Systems Limited. Therefore, the documentation made available to the customer may refer to features that are not present in the solution purchased / deployed at the customer site.

# About this manual

This manual briefly describes the basic processes and functions in Ramco Aviation Solution.

### Who Should Read This Manual

This manual is intended for users who are managing the Aviation industry processes and are new to Ramco Aviation Solution.

This manual assumes that the user is familiar with the Aviation Industry nomenclatures and systems based software.

#### How To Use This Manual

Ramco Aviation Solution provides extensive Online Help that contains detailed instructions on how to use the application. Users are suggested to use this manual for specific references, along with the Online Help. This manual contains enough information to help the users perform the basic tasks and points toward the Online Help for more detailed information.

#### How This Manual is organized

The User Guide is divided into 3 chapters and index. Given below is a brief run-through of what each chapter consists of.

Chapter 1 provides an overview of the entire **Maintenance Programs** business process. The sub processes are explained in the remaining chapters.

Chapter 2 focuses on the Maintenance Task sub process.

Chapter 3 focuses on the Component Maintenance Program sub process.

Chapter 4 focuses on the Aircraft Maintenance Program sub process.

The Index offers a quick reference to selected words used in the manual.

#### **Document Conventions**

- The data entry has been explained taking into account the "Create" business activity. Specific references (if any) to any other business activity such as "Modify" and "View" are given as "Note" at the appropriate places.
- ▶ Boldface is used to denote commands and user interface labels. Example: Enter Company Code and click the Get Details pushbutton.
- Italics used for references.

Example: See Figure 1.1.

The <sup>>></sup> icon is used for Notes, to convey additional information.

# **Reference Documentation**

This User Guide is part of the documentation set that comes with Ramco Aviation Solution.

The documentation is generally provided in two forms:

- ▶ The Documentation CD in Adobe® Systems' Portable Document Format (PDF).
- Context-sensitive Online Help information accessible from the application screens.

#### Whom To Contact For Queries

Please locate the nearest office for your geographical area from www.ramco.com for assistance.

# content

1	INTRODU	ICTION	6
2	MAINTEN	IANCE TASK	
		HORING A TASK	
2	2.1.1	Defining work areas and zones	
	2.1.1	Creating access panel tasks and access panels	
	2.1.2	Specifying parts for the access panel task	
	2.1.3	Specifying resources for the access panel task	
	2.1.4	Specifying model effectivity details for the access panel task	
	2.1.5	Associating access panels to access panel task	
	2.1.7	Creating an access panel	
	2.1.7	Defining quick codes	
	2.1.9	Setting options	
	2.1.10	Creating a task	
	2.1.10	Creating standard task from non-standard task	
	2.1.12	Specifying parts for the task	
	2.1.12	Specifying resources/sign-off information for the task	
	2.1.13	Specifying schedule information for the task	
	2.1.15	Specifying model effectivity details for the task	
	2.1.16	Editing aircraft mod # details	
	2.1.17	Specifying component effectivity details for the task	
	2.1.18	Editing the access panel information	
	2.1.19	Specifying the task references	
	2.1.20	Entering work area or zone information for the task	
	2.1.21	Specifying parameter details for the task / sub-task	
	2.1.22	Creating dependent systems condition list	
	2.1.23	Authorizing tasks	
	2.1.24	Returning a task	
	2.1.25	Managing unified task sequence	
	2.1.26	Managing unified task sequence	
	2.1.27	Mapping parts to sequence control #	
	2.1.28	Mapping tasks to sequence control #	
	2.1.29	Sequencing tasks	
	2.1.30	Recording Task Part modification details	
	2.1.31	Maintaining activated tasks	
	2.1.32	Revising an activated task with a new revision	
	2.1.33	Deactivating a task	
	2.1.34	Reactivating a task	
	2.1.35	Maintaining task relationship	
	2.1.36	Managing repair scheme definition	
	2.1.37	Maintaining repair scheme	
	2.1.38	Managing task effectivity	
	2.1.39	Managing task file attachment	
	2.1.40	Authoring repair procedure	
	2.1.41	Maintaining Standard Cost of task	50
3	COMPON	IENT MAINTENANCE PROGRAM	52
3	8.1 Con	FIGURING COMPONENT MAINTENANCE PROGRAM	53
	3.1.1	Defining the quick codes	
	3.1.2	Setting options	
	3.1.3	Creating part maintenance program	
	3.1.4	Defining date based schedule	

# content

ra	m	C	0

3.1.5	Defining usage based schedule	57
3.1.6	Copying part maintenance program to multiple parts	58
3.1.7	Authorizing a part maintenance process	59
3.1.8	Creating component maintenance program	59
3.1.9	Associate work units on opportunity check	62
3.1.10	Authorizing a component maintenance program	62
3.1.11	Creating position based schedule	63
3.1.12	Creating date based schedule	63
3.1.13	Create usage based schedule	64
3.1.14	Define aircraft effectiveness	
3.1.15	Activating a position based schedule	65
3.1.16	Updating work units in existing maintenance program	66
3.1.17	Maintaining restoration work units	67
3.1.18	Managing Schedule Adjustment for Components	68
4 AIRCRAFT	MAINTENANCE PROGRAM	71
4.1 CON	FIGURING THE MAINTENANCE PROGRAM FOR AIRCRAFT MODELS	73
4.1.1	Creating quick codes	73
4.1.2	Setting options for aircraft maintenance program	73
4.1.3	Maintaining maintenance events	
4.2 CREA	TING AN AIRCRAFT MAINTENANCE PROGRAM	76
4.2.1	Specifying schedules for carrying out the maintenance program	80
4.2.2	Identifying the maintenance events to be carried out for the program	82
4.2.3	Capturing document references for the maintenance program	
4.2.4	Defining model effectivity for the maintenance program	
4.3 Asso	CIATING AIRCRAFT TO MAINTENANCE PROGRAM	
4.3.1	Associating multiple aircraft to the maintenance program	85
4.3.2	Associating multiple programs to an aircraft	86
4.3.3	Modifying the maintenance program of an aircraft	87
4.3.4	Modifying the schedules for an aircraft specific maintenance program	
4.3.5	Modifying the scheduled dates and the values for the work units in the mainte	nance program 93
4.3.6	Inactivating the association of an aircraft to a maintenance program	
4.4 ACTI	VATING A MAINTENANCE PROGRAM	
4.4.1	Comparing the revisions of a maintenance program	
4.5 Revis	SING A MAINTENANCE PROGRAM	100
4.5.1	Revising the program	100
INDEX		101

# **1INTRODUCTION**

Aircraft and its components undergo regular maintenance inspections and repairs throughout their life cycle. Maintenance activities to be performed on aircraft and components are provided by the OEMs. Maintenance time limits and intervals for these maintenance activities are also provided by the OEMs based on the standard operating conditions. Schedule for the maintenance activities stipulating what, how and when to perform the maintenance according to the operating conditions is formulated by the respective airline operators in the maintenance program. A maintenance program provides details of maintenance tasks and their time limits and intervals adhered to by the airline operator. The customized scheduled maintenance program is then authorized by the regulatory authorities for execution.

The Maintenance Programs business process addresses the above-mentioned business need and facilitates users to define maintenance tasks for aircraft as well as components, and then devise maintenance programs for aircraft and components by grouping these tasks based on the requirement. Finally, the time intervals, the relationship and the sequence between the tasks in the maintenance programs are computed to arrive at the scheduled date of execution of the task on the aircraft/component. The Maintenance Programs process involves the Maintenance Task, Component Maintenance Program and Aircraft Maintenance Program business components

# 2 MAINTENANCE TASK

Aircraft and its components undergo regular maintenance inspections and repairs throughout their life cycle. Maintenance activities to be performed on aircraft and components are provided by the OEMs. Maintenance time limits and intervals for these maintenance activities are also provided by the OEMs based on the standard operating conditions. Schedule for the maintenance activities stipulating what, how and when to perform the maintenance according to the operating conditions is formulated by the respective airline operators in the maintenance program. A maintenance program provides details of maintenance tasks and their time limits and intervals adhered to by the airline operator. The customized scheduled maintenance program is then authorized by the regulatory authorities for execution.

The Maintenance Programs business process addresses the above-mentioned business need and facilitates users to define maintenance tasks for aircraft as well as components, and then devise maintenance programs for aircraft and components by grouping these tasks based on the requirement. Finally, the time intervals, the relationship and the sequence between the tasks in the maintenance programs are computed to arrive at the scheduled date of execution of the task on the aircraft/component.

The Maintenance Programs process involves the Maintenance Task, Component Maintenance Program and Aircraft Maintenance Program business components.

Authoring of aircraft and component maintenance tasks and managing the changes to a task are handled as part of this sub process.

The **Maintenance Task** business component enables you to identify the basic maintenance tasks, which will be used in maintaining the aircraft and components.

# **2.1 Authoring a task**

# 2.1.1 Defining work areas and zones

#### **Creating a work area**

Work area is the physical location in the aircraft where the maintenance activity must be carried out.

1. Select Create Work Area under Maintenance Task business component. The Create Work Area Information page appears. *See Figure 2.1.* 

★ 🗎 Create Work Area Information		
Model Details     Aircraft Model #	F 101-00	
Work Area Details		
44 4 1 -1/1 > >> + - 0 - 0 - 0 - 0 - 0		Q <b>v</b>
# 🖻 Work Area # Work Area D	Description	
1 🖾 WA310 Work area for	A310 model aircraft	
2		
	Create Work Area	

Figure 2.1 Creating work area information

- 2. Select the Aircraft Model # for which the work area is being defined.
- 3. Enter the Work Area # and Work Area Description in the multiline, for the selected aircraft model.
- 4. Click the Create Work Area pushbutton.

#### 2.1.2 Creating access panel tasks and access panels

#### **Creating an access panel task**

Access panels provide work area access for performing maintenance activities. A maintenance task can have corollary tasks relating to the opening and closing of the required access panels.

1. Select Create Access Panel Task under the Maintenance Task business component. The Create Access Panel Task Information page appears. *See Figure 2.2.* 

Create Access Panel Task Information					≣ ≭ ≣		• ? 🗔
- Access Panel Task Details				•			
Access Panel Task #	AP201501	Entor the depende	nt ovotom	Base Aircraft Model # 👂 190-100AR			
AP Task Description A		Enter the depende		÷			
AP Task Type 0	Dpen 💌 🚽	condition to which	the access				
Est. Elapsed Time		nanel task must be	associated	Est. Man Hrs.	8.00		
Maint. Manual Ref. # A				System Reference #			
DSC # P D	DSC_1			DSC Description dsc_1			
Copy From     Access Panel Task # P	[1-333			Base Aircraft Model # 🔎			
	All Parts Del	tails					
	Sub Task Resource						
- Sub-Task Details							
	τ.				•		Q
# 🖻 Seq # Sub Tasks Resource Group	Document ID	File Name Maint, Manual Ref. #	DSC # P	DSC Description			
1 Inspector	✓ ST 1D						
2							
Specify the sequence in v	which						
the sub task is to be perfe	ormed			quence sub-tasks			
		in multiples of	5 to enable you	u to enter		_	
				n existing sequence			>
Re - Number							
		Create AP T	ask				
Edit Part Requirements	Edit Resource	e / Sign-Off Requirements		Edit Aircraft Model Effectivity			
Associate Access Panels							

Figure 2.2 Creating access panel task

2. Enter the Access Panel Task # to uniquely identify the access panel task.

- 3. Identify the **Base Aircraft Model #** for the access panel task.
- 4. Select the access panel task type in the **AP Task Type** drop-down list box. The access panel task could be of type "Open" or "Close".
- 5. Enter the details such as AP Task Description, Est. Elapsed Time and Est. Man Hrs.
- 6. To copy details from an existing access panel task, enter the Access Panel Task # in the Copy From group box and specify the Copy Options.
- 7. Enter the sub tasks to be performed for carrying out the access panel task, in the **Sub Tasks** field in the multiline.
- 8. Select the **Resource Group**, which is responsible for executing the sub-task.
- 9. Click the Create AP Task pushbutton.

#### To proceed,

- $\tau$  Select the Edit Parts Requirement link to specify parts for the access panel task.
- $\tau$  Select the Edit Resource / Sign-Off Requirements link to specify resources for the access panel task.
- $\tau$  Select the Edit Aircraft Model Effectivity link to specify access panel task applicability to aircraft model.
- $\tau$  Select the Associate Access Panels link to associate access panels to access panel task.

#### 2.1.3 Specifying parts for the access panel task

You can add or modify the details of parts required for executing the access panel task.

- 1. Select the Edit Parts Requirement link in the Create Access Panel Task Information page.
- 2. Enter the **Part #** that is required for carrying out the access panel task.
- 3. Enter the **UOM** and **Est. Quantity** of the part that is required.
- 4. Click the Edit Part Requirements pushbutton.

# 2.1.4 Specifying resources for the access panel task

You can add or modify the details of resources required for executing the access panel task.

- 1. Select the Edit Resources Requirement link in the Create Access Panel Task Information page.
- 2. Select the type of the resource to be specified for the task, in the Resource Type drop-down list box. The resource can be of type "Skill", "Tools", "Equipment" or "Others".
- 3. Enter the resource in the Resource # field in the multiline.
- 4. Select "Yes" for "Skill" type of resources in the Approval Reqd.? field, if you want the system to verify the license or certification availability for the employee who executes the task.
- 5. Select "Yes" for "Skill" type of resources in the Sign-Off? field, if you want the system to ensure sign-off for the task by an employee who possesses the skill, during work execution.
- 6. Click the Edit Resources pushbutton.
- 7. Click the Edit Sign-Off Details pushbutton to update the sign-off details

# 2.1.5 Specifying model effectivity details for the access panel task

You can define the model effectivity, to indicate the applicability of the access panel task for the aircraft models.

- 1. Select the Edit Aircraft Model Effectivity link in the Create Access Panel Task Information page.
- 2. Enter the Aircraft Model # in the multiline.
- 3. Click the Edit Model Effectivity pushbutton.

#### 2.1.6 Associating access panels to access panel task

You can identify the access panels on which the access panel task can be performed.

- 1. Select the Associate Access Panels link in the Create Access Panel Task Information page.
- 2. Enter the default aircraft model number that you wish to specify for the access panels in the **Def.** Aircraft Model field.
- 3. Enter the Aircraft Model # and Access Panel # in the multiline.
- 4. Click the Associate Access Panels pushbutton.

# 2.1.7 Creating an access panel

Access panels provide work area access for performing maintenance activities. Executing a maintenance activity may involve opening of one or more panels to access a component or the area inside the component.

1. Select Create Access Panel under the Maintenance Task business component. The Create Access Panel Information page appears. *See Figure 2.3.* 

Create Access Panel	Information				32\$	₽ ·	<b>+</b> 3	
- Access Panel Details								
	Aircraft Model #	190-100AR	v	Access Panel #				
	Access Panel Description	Access panel for	opening an engine					
Access Panel Task Details	Zone # 🔎	1	Identify the aircraft model for which the					
	Open Access Panel Task # 👂	SM-001	access panel is defined	Open AP Task Description sm test				
	Close Access Panel Task # 👂			Close AP Task Description				
			Create Access Panel					

Figure 2.3 Creating access panel

2. Enter the Access Panel # as a unique identifier of the access panel.

Enter the Access Panel Description.

- 3. Enter the Zone number identifying the zone to which the access panel must be associated.
- 4. Specify the Open Access Panel Task # to open the access panel.
- 5. Specify the Close Access Panel Task # to close the access panel.
- 6. Click the **Create Access Panel** pushbutton.

#### 2.1.8 Defining quick codes

#### What are quick codes?

Quick codes can be used to classify the task based on the type of maintenance activity to be carried out. Some examples of quick codes for task are "Lubrication" and "Inspection".

1. Select the **Create Quick Codes** link under the **Maintenance Task** business component. The **Create Quick Codes** page appears. *See Figure 2.4.* 



*		Create Quick Codes			≣ 7\$	-	⇒ ←	?	Q K
		, i a i d	Quick Code Type 🛛 🔻						
1		1 - 1 / 1 → → + - □	- 0 0 T T			Ŧ		_	Q
#	1	Quick Code	Description						
1			Washing parks				×	1	
2									
				Create Quick Codes					

Figure 2.4 Defining quick codes

- 2. Enter the unique identifier of the quick code in the **Quick Code** field in the multiline.
- 3. Enter the **Description** of the quick code.
- 4. Click the **Create Quick Codes** pushbutton.

#### 2.1.9 Setting options

You can set default options for generating a non-standard task. You can specify whether a nonstandard task can be generated and the numbering series to be followed while generating them.

1. Select the **Set Options** link under the **Maintenance Task** business component. The **Set Options** page appears. *See Figure 2.5*.

*	D) (	Set Options													ב¢			- '	2 🗔
	otion	Setting Details									Date	ronnar IIII	-uu-yyyy						
				G	enerate	Non-S	andaro	Tasks res 🔻											
			Numbe					d Tasks NST 🔻											
								ectivity Allowed											
								fication Allowed											
								n Mode Manual	•										
			5	equer															
							umberin		•										
		Repair Scheme ce Group For Repair Proce		Re-Si	equence	e Multip	olication	Factor	100										
												_		_	_		_	_	
44	1	1 - 4 / 4 > >> + 6		T	<b>▼</b> <sub>x</sub>						▶ 늘 등 곳 문 븝 ∞ 박		All			•		_	Q
#		Execution Document	Mechanic	c	Inspec	tor	Mecha	anic & Inspector	Not Requir										
1		Shop Work Order	No	*	No	~	Yes	~	Yes										*
2		A/C Maint. Exe. Ref #	Yes	*	Yes	~	Yes	*	Yes										*
3		Central Discrepancy	No	~	No	~	No	*	Yes										*
4		Others	No	•	No	~	No	•	Yes										*
5				~		~		*											¥
_																			
										Set Option									
	ecord	Statistics																	
			Last	Modif	ied By	DMUS	ER				Last Modifie	ed Date 03	-16-2016						

Figure 2.5 Setting options for task

- 2. Specify whether non-standard tasks can be generated in the **Generate Non-Standard Tasks** dropdown list box.
- The system retrieves the numbering types that have been defined for the task in the Numbering Type for Non-Standard Tasks drop-down list box. Select the numbering type for the non-standard task.
- 4. Indicate whether **Multiple Model Effectivity** is allowed for task, standard procedure and nonstandard task by selecting the appropriate option.
- 5. Use the **Maint. Program Modification** drop-down list box to indicate whether component/aircraft maintenance programs can be modified. From the drop-down list box, select,
  - "Allowed", to enable modification of maintenance programs.
  - "Not Allowed", to prevent modification of maintenance programs.
  - "With Revision", to allow modification of maintenance programs with a new revision number

- 6. Use the **Sequence Control Generation Mode** drop-down list box to indicate whether the generation of sequence control number must automatic or manual. Select,
  - Automatic, to enable the system to automatically generate the sequence control number on the basis of the default numbering type.
  - Manual, to enable users to specify the identification number for a sequence control.
- 7. Use the Default Numbering Type drop-own list box to select the numbering type for the automatic generation of sequence control numbers in the Manage Unified Task Sequence activity. The drop-down list box displays all numbering types in Active status defined for the transaction Maintain Task Seq #. It is mandatory to specify the numbering type, if the sequence control generation mode is "Automatic". Conversely, the drop-own list box displays no values, if the sequence control generation mode is set as "Manual".
- Enter the Re-Sequence Multiplication Factor for re-sequencing of sequence control numbers in the Manage Unified Task Sequence activity, Mandatory. The factor you specify here must be a positive integer.
- 9. Click the **Set Options** pushbutton.

# 2.1.10 Creating a task

You can define the task, which is the basic unit of the maintenance work performed on the equipment. You can create various revisions of task. Revisions represent the number of changes the task has undergone.

- 1. Select **Create Task** under the **Maintenance Task** business component. The **Create Task Information** page appears. *See Figure 2.6.*
- 2. Indicate whether the task is applicable for an "Aircraft", "Engine" or "Component" n the **Task** Applicability drop-down list box.
- 3. Select the base aircraft model on which the task can be performed in the **Base Aircraft Model #** field.
- 4. Enter the **Task #** as a unique identifier of the task.
- 5. Specify **Revision #** and **Revision Date** for the task.
- 6. Select the type of the task in Task Type.
- 7. Enter Task Description.
- 8. Select the category of the task in Task Category field.
- 9. Select the **Operations Type** of the maintenance task. Specify the work center in which the maintenance task must be executed in the **Work Center #** field.
- 10. Use the **WBS Code** drop-down list box to specify the Work Breakdown Structure (WBS) Code indicating the attributes that identify whether the task is executed for planning purpose or execution purpose.
- 11. To copy details from an existing task, enter the Task #, Revision # and Base Aircraft Model # in the Copy From group box and specify the Copy Options.
- 12. Check the following boxes to copy specific details of the task to the new task being created by you. All: To copy all the details pertaining to the task. Details include reference, schedule, sub tasks, access panel, work area or zone, effectivity, reference details, parts and resources.
  - Schedule: To copy the schedule details
  - Sub Task: To copy the sub tasks.



- Task Requirements: To copy the task requirement details such as spare part details and resource details.
- Effectivity: To copy the effectivity details
- Access Panel /Work Area /Zone: To copy the access panel, work area or zone details
- Task References: To copy the reference document details
- 13. In the Task Additional Details group box, specify the Est. Elapsed Time, Insp. Man Hrs. and Est. Man Hrs. required for executing the task.

* 8	Cre	ate Task	Information							RAMCO OU-F	tamco Role 👻 🛛	× 0 0	← 🖽	? 🗔
									Date Format	yyyy/mm/d	d			
- Tas	k Details	5	Task Applicability	Aircraft 💌					Base Aircraft Model #			Q		
			Task #		Enter the task				Revision #					
			Task Type						Revision Date	2021/02/03		<b></b>		
			Task Description		description									
			Long Description											
			Task Category	-					Operations Type	FLIGHT OP	5 💌			
			Work Center #	Q					WBS Code	1-PME	N			
- Cop	y From-													
				0000-0000007	Q				Revision #	1				
			Base Aircraft Model #	008-200	Q								1	
						Periodicity     Task Require			Specif	iy the V	Vork			
				Copy Options	Effectivity	Access Panel		7000	Drock	down C	tructure /			
					Task References	Access Faller	/ WOINAIG	ca / 2011e	Dreak	uown a	Structure (	(VVDS)		
					rust references				code t	o ident	ify the tas	sk		
- Tas	k Additio	onal Details	ATA #				1		System ATA # Description					
			DSC #		Enter the ATA	chanter			DSC Description					
			Est. Man Hrs.			chapter			Insp. Man Hrs.					
			Est. Elapsed Time	Hou					Output Qty. Type	Fixed	-			
			Wrench Time						Ideal Time					
			Emp. Loading for	Elapsed time					Exec. Phase	Regular		-		
_	-Task D		Default Exec. Comments	_	Enter defau comments									
•		no rows to d		Q % Y %						∔ % All		<ul> <li>Search</li> </ul>		Q
#		Seq #	Sub Tasks			Resource Grou	ıp		Document ID		File Name			
1						Not Required		~						
			Maint, Manual Ref. # Source Document # Manufacturer # Work Location Ref. # Nandatory? Eng. Doc. Class User Defined 2 EO User Defined 5		0 • •				MPD Item # Source Document Type Manufacturer Name Eng. Doc. Eff. Date Reliability Related? User Defined 1 EO User Defined 4 EO User Defined 6	Others	•			
- Doc	ument /	Attachment	Details File Name		O View File									
			File Name		D view File	Create Task								
Edit Par	Require	ments		Edit	Resource/Sign-Off Requirement				Edit Schedule Inform	ation				
	lel Effecti				Component Effectivity	-			Edit Access Panel Del					
		Zone details			Notes				Edit Task / Sub-Task					
	raft Effec				Task Details				Edit Parameter Readi		m			
	Repair S	Scheme andard Cost		Man	age Task Effectivity				Manage Task File Att	acnment				
- iumodii	. uak 3te	andere cost												I

Figure 2.6 Creating a task

- 14. Use the **Output Qty. Type** drop-down list box to select the type of output quantity from the task. The drop-down list box displays Fixed and Variable.
- 15. Enter the total time estimated for the actual execution of the task in the Wrench Time field.
- 16. Enter the Ideal Time recommended by the OEMs for the completion of the task.
- 17. Use the **Emp. Loading** drop-down list box to select the time for employee loading for the task. The drop-down list box displays Elapsed time and Wrench time.
- 18. Enter the execution comments to be defaulted in the shop work order or aircraft package at the time of assigning the task in the **Default Exe. Comments** field.
- 19. Select the **Execution Phase** of the task. The execution phase can be one of the following:

- Preparatory Select this option if the task must be performed as a part of preliminary activities before aircraft is grounded for maintenance.
- Regular Select this option if the task must be performed as a part of normal maintenance activities carried out in hangar.
- Post Flight Select this option if the task must be performed as a part of post test flight activities.
- 20. Enter the **Sub Tasks** that must be performed while executing the maintenance task, in the **Sub-Task Details** multiline.
- 21. Enter the Maintenance Planning Document item number of the task, in the MPD Item #.
- 22. Specify the dependent system condition to which the sub-task must be associated in the **DSC #** field.
- 23. Enter the projected time for executing/completing the sub-task in the Est. Elapsed Time field.
- 24. Use the **Time UOM** drop-down list box to select the unit of measurement for the estimated elapsed time.
- 25. Use the **Output Qty. Type** drop-down list box to select the type of output quantity from the subtask.
- 26. Use the Emp. Loading drop-down list box to select the time for employee loading for the sub-task.
- In the Reference Details group box,
- 27. Specify the Aircraft Maintenance Manual or the Component Maintenance Manual item number of the task in the **Maint. Manual Ref. #** field.
- 28. Specify the Maintenance Planning Document item number of the task in the MPD Item # field.
- 29. Identify the **Source Document Type** and enter the **Source Document #** for the task.
- 30. Enter Manufacturer # associated with the task.
- 31. Enter the Work Location Ref. # for the task.
- 32. Enter the date till which the engineering document is valid in the Eng. Doc Eff. Date field.
- 33. Use the Mandatory drop-down list box to indicate whether the task is mandatory for the part #.
- 34. Use the **Reliability Related** drop-down list box to indicate whether the task executed on the part will be evaluated for the component reliability analysis.
- 35. Use the **Eng. Doc. Class** drop-down list box to select the engineering document class to be associated with the task.
- 36. Click the Create Task pushbutton.

#### Mapping of new task to sequence control

If you have accessed the "Create Task Information" page from the "Manage Repair Scheme Definition" page and if you have specified a sequence control for the new task in the later page, the system on authorization of the task:

- Maps the new task to the sequence control
- Makes all those parts that are mapped to the sequence control effective for the new task.

To enable the system to automatically perform the above-explained jobs, the new task must possess WBS code with process parameters defined in the Common Masters component in the following way:

- ▶ WBS level? as "Root/Intermediate"
- Allow Master Sequence Control definition?' as "Yes".

#### **Creating repair scheme task**

If the newly created task has a parent task, the system sets the new task as a repair scheme task in the repair scheme of the parent task.

To proceed,

- $\tau$  Select the **Edit Part Requirements** link to specify the materials required for carrying out the task.
- $\tau$  Select the **Edit Resource/Sign-Off Requirements** link to specify resources and sign-off information for carrying out the task.
- $\tau$  Select the **Edit Schedule Details** link to specify the schedule details for the task.
- $\tau$  Select the **Edit Model Effectivity** link to specify the task applicability to aircraft model.
- $\tau$  Select the **Edit Component Effectivity** link to specify the task applicability to the component.
- $\tau$  Select the **Edit Access Panel Details** link to specify the access panels that are to be used for executing the task.
- $\tau$  Select the **Edit Task / Sub-Task References** link to specify the reference document details for the task.
- $\tau$  Select the **Edit Work Area / Zone Details** link to specify the work area and the aircraft zone where the task has to be executed.
- $\tau$  Select the **Edit Parameter Reading Requirements** link to add or modify the parameter details of the task/sub-task.
- $\tau$  Select the **Edit Aircraft Effectivity** link to specify the task applicability to the aircraft.
- $\tau$  Select the Maintain Repair Scheme link to create a repair scheme for the tasks.
- $\tau$  Select the Manage Task Effectivity link to manage task effectivity details.
- $\tau$  Select the **Manage Task File Attachment** link to associate multiple files to a task for printing.

# 2.1.11 Creating standard task from non-standard task

You can create standard task from non-standard task Non-Standard tasks are the unscheduled, unplanned tasks carried out during the non-standard kind of maintenance. The non-standard tasks can be generated by the system while creating the work orders. Using this activity you can convert non-standard tasks to standard tasks.

- 1. Select Create Non-Standard Based Standard Task under the Maintenance Task business component. The Select Non-Standard Task page appears.
- 2. Specify search criteria and click the **Search** pushbutton, to search for non-standard tasks.



3. Click the hyperlinked non-standard task in the multiline. The **Create Non-Standard based Task Information** page appears. *Figure 2.7* 



Figure 2.7 Creating non-standard based standard task

Follow the instructions given under the topic "Creating a task" to create standard tasks from nonstandard tasks.

# 2.1.12 Specifying parts for the task

You can add or modify the details of parts required for executing the task.

- 1. Select the Edit Part Requirements link in the Create Task Information page. The Edit Part Requirements page appears. *See Figure 2.8.*
- 2. Enter the **Part #** field and other details in the multiline.

Or

- 3. Enter the Mfr. Part # and the Mfr. # of the part.
  - Note: The system displays the Mfr. Part # and Mfr. #fields for you to enter, and hides the "Part #" field above, if the parameter "Enable Manufacturer Part # control in transaction" under the Category 'Manufacturer Part #' is set as "Yes" in the "Set Inventory Process Parameters" activity of the "Logistics Common Master" business component.
- 4. Enter the **Customer #** of the part.



★ ■ Edit Part Requirements      Task Details      Task ≠ TSK500      Revision ≠      Base Aircraft Model ≠ 0613      Part Details      4 ← 1 - 1/1 ▷ ᠉ + - □ ◎ ※ ♀ ∅ ▼ ▼	Check this box if y part requirements be revised based consumption reco work orders	for this task to on the parts	Task Description TSK5	Task Applicability Aircraft Task Description TSK500 created Update based on Actuals				
# Part # O Customer # O Part Description	Part Type	UOM P	Est. Quantity	Need Frequency	- Oca	Probability	Q	
1 E :10973-PDC_3 1090000 :10973-PDC_3	Expendable	EA		Always	~			
2				Always	~			
'Part #' is hidden and 'Mfr. Part #' and 'Mfr. #' fields are displayed based on inventory options set in "Logistics Common Masters" business component	Edit Parts						•	

Figure 2.8 Specifying parts for the task

Enter the required quantity of the part to carry out the task in the Est. Quantity field.

- 5. Use the **Need Frequency** drop-down list box to indicate the requirements of the part.
- 6. Enter the **Probability** of requirement of the part. A positive value ranging between 0 and 1 (e.g. 0.1, 0.5, 0.9) is entered here if the 'Need Frequency' is set as "As Required" or the value '1' is entered if the "Need Frequency" is set "Always".
- 7. Click the Edit Parts pushbutton.

# 2.1.13 Specifying resources/sign-off information for the task

You can add or modify the details of resources required for executing the task.

- 1. Select the Edit Resource/Sign-Off Requirements link in the Create Task Information page. The Edit Resource/Sign-Off Requirements page appears. *See Figure 2.9.*
- 2. Select the Resource Type you wish to specify for the task.
- 3. Enter details such as the **Resource #, Time UOM, Customer #, Est. Time** and Est. Nos in the multiline, to specify the resources required for executing the task.
- 4. Select "Yes" for the "Skill" type of resources in the **Approval Reqd.?** field, if you want the system to verify the license and certification availability for the employee who executes the task.

*	E EC	lit Resourc	e / Sign-Off Requirements									1 🗲 ? 🗔
— Та	sk Deta	ils										
		Task Applicabilit	ty Component						Base A	ircraft Model a	#	
		Task	# TSK109							Revision a	#	
		Task Descriptio	n TSK109 created									
+ Re	source	Details										
*	•	1 - 1/1 🕨	» + - 🗆 % 🌣 🖉 🍸	7			<u> 노 비 명 값 오</u>	🖹 ×4 🔮	X # # III N %	All	▼ Search	Q
#		Resource #	Q	Resource Description	Time UOM		Customer #	Q	Est. Time		Est.Nos	Approval R
1		130-00-0230		CMC TECHNICAL SERVICES	Hours	~	1090000					No
2					Hours	~						No
_		-Off Inform Sign-Off Infor	"Resource #" drop-d		20	it Resource						
	Four	nd no rows to di	splay!!! 🕨 💌 🛨 🗆 🔀 😫	× T X				🖹 × 🗳	図 単 美 田 林 ※	All	▼ Search	Q
#		Seq #	Sub Tasks			Resource (	Group					
1						Not Require	ed					~
		4			Edit S	ign-Off Det	ails					ł

Figure 2.9 Specifying resources to the task

5. Click the Edit Resources pushbutton.

In the Resource Sign-Off Information group box,

- 6. Use the **Resource #** drop-down list box to specify the resource required for completing the task.
- 7. Click the **Get details** pushbutton to retrieve the sub-task details for the task number, aircraft model number and resource number combination.
- 8. Use the **Resource Group** drop-down list box to specify the resource group. The system provides the options "Not Required", "Mechanic", "Inspector" and "Insp & Mech".
  - Note: You can set the "Resource Group" only to "Inspector" or "Not Required", if the resource group of the sub-task is set as "Inspector" in the previous page. You can set the "Resource Group" only to "Mechanic" or "Not Required", if the resource group of the sub-task is set as "Mechanic" in the previous page.
  - Solution If resources are deleted in the "Resource Details" multiline, the system automatically removes the corresponding sub-task details from the multiline.
- 9. Click the Edit Sign-Off Details pushbutton to update the sign-off details.

# 2.1.14 Specifying schedule information for the task

You can enter the schedule information for the task. The schedule decides the optimum limits within which the task is to be performed. Schedule can be in terms of calendar duration or on the basis of one or more consumption parameters (like flight hours / flight cycles). This becomes very important especially for tasks of type MPD and CMM.

- 1. Select the Edit Schedule Details link in the Create Task Information page. The Edit Schedule Information page appears. *See Figure 2.10.*
- 2. Select the **Schedule Type** as "Recurring" or "One-Time".
- 3. Specify the **Threshold** and **Repeat Interval**, to indicate the duration after which the task must be performed and the interval within which the task must be repeated thereafter.
- 4. Enter the **Terminating Value** field to specify the terminating value in terms of days, for task execution.
- 5. Specify the Time Unit as 'Days', 'Months', 'Month End' and 'Years'.
- 6. Use the Limit Basis drop-down list box to select the limit basis for picking the Next Schedule Date (NSD) for a task that has multiple NSDs. This situation occurs either when a task has date-based and usage-based schedules or when the usage-based schedule comprises multiple parameters.
- 7. In the Usage Based Schedules multiline, enter the Parameter based on which the task must be performed.
- 8. Specify the Threshold Value, Repeat Interval and Terminating Value for the parameter.



Edit Schedule Information						≣ 7\$		*	? 🗔
Task Details				Date Format	yyyy-dd-mm				
	# 1-50C-0000-CMM-00005626	5		Revision #					
Task Applicabilit	y Component			Base Aircraft Model #					
Task Descriptio	n PME-2			Schedule Type	Recurring 💌				
Periodicity Revision	#			Revision Date		1			
<ul> <li>Date Based Schedules</li> </ul>									
	it Days 🔻				Whichever Is Earlier	*			
Threshol		20.00		Repeat Interval		20.00			
Terminating Valu	e								
(4) 4) 1 - 2/2 → → + - □ → ☆ ☆	T T.		٨		II AI		Ŧ	-	Q
# 🗉 Parameter 👂	Parameter Description	UOM	Threshold Value	Repeat Interval		Terminating	Value		
1 🗈 APUC 🖌	APU Cydes	CYC	1000.00	•	1000.00		100	0000000	.00
2 🖻 APUH 🖌	APU Hours	HRS	10000.00		50000.00		5000000	000000	.00
3									
Notes     Failure Effect	t			Failure Effect Description					
Special Frequency Note	~				^				
					~				
Access Note	s				$\sim$				
Aircraft Note	s				0				
					~				
Engine Note	S				$\sim$				
Interval Note	s				0				
Update Options									
Inherit Schedules to Program	? Yes 🔻			Inherit Schedules to Program Template?					
Remark	s				0				
			Edit Schedules						
			Ealt Schedules						
Initialize Maint. Prog. & Update Compliance									

#### Figure 2.10 Specifying schedule information of task

- 9. In the **Update Options** group box, use the **Inherit Schedules to Program?** drop-down list box to indicate whether task schedules must be updated in the maintenance programs associated with aircraft/components when schedules for the task are modified.
- 10. Use the **Inherit Schedules to Program Template?** drop-down list box to indicate whether task schedules must be updated in the maintenance programs of part/aircraft model when schedules for the task are modified.
- 11. Click the **Edit Schedule** pushbutton.

The system performs the following:

- The schedules of tasks are updated in the maintenance programs only if the Prog. Item Type of tasks is 'Block', 'Base', 'Non-Block' or 'To Be Decided'.
- When a new parameter is added for the task, the system updates the parameter details to the AMP, CMP, Model Program and Part Program, if the added parameter is defined for the effective aircraft, component, model and part, respectively.
- ▶ NSD / NSV computation Computes and updates the NSD / NSV for tasks. Refer to the flow diagram for NSD / NSV computation. *See Figure* 2.10a-g.



Figure 2.11 a Flow diagram for NSD / NSV computation - Updating maintenance program



Figure 2.10b NSD / NSV computation - Updating maintenance program



Figure 2.10c NSD / NSV computation - Updating maintenance program



Figure 2.10d NSD / NSV computation - Date Based Schedule Update



Figure 2.10e NSD / NSV computation - Date Based Schedule Update



Figure 2.10f NSD / NSV computation - Usage Based Schedule Update



Figure 2.10g NSD / NSV computation - Usage Based Schedule Update

To proceed,

τ Select the Initialize Maint. Prog. & Update Compliance link to update schedule details of tasks and record compliance information.

# 2.1.15 Specifying model effectivity details for the task

- You can define the model effectivity, to indicate the applicability of the task for the aircraft models. Note: This page is displayed only when the "Multiple Model Effectivity" option is set as "Allowed" in "Set Options" activity.
- 1. Select the Edit Model Effectivity link in the Create Task Information page. The Edit Model Effectivity page appears. *See Figure 2.11.*



*	) I	Edit Model Effectivity	1					<b>=</b> 7		+ '	? 🗔 🖪
							Date & Time Format y	/yyy-dd-mm	hh:mm:ss		
T	ask De	tails									
			Task #	1-A310-0000-MPD-00006396			Revision #				
			Task Description	AIRCRAFT INSPECTION AS A RESUL			Task Applicability A	Vircraft			
M	odel E	ffectivity Details									
44	•	1 -1/1 > >>	+ - 0 % 0 0	T T.				III AI	Ŧ		Q
#		Aircraft Model # 🔎	Model Type	Model Description	Aircraft Effectivity	Comments	Created By	Created Date/Time		Last Mo	dified by
1		737-200		Boeing 737-200	All		DMUSER	2014-10-03 12:53:58			
2											
		<									>
					Edit Effectivity						
Edit A	rcraft	Effectivity		Edit Mod # Effectivit	у		View Maint. Operator E	ffectivity			

Figure 2.12 Specifying model effectivity of task

- 2. Enter the Aircraft Model # in the multiline.
- 3. Click the **Edit Effectivity** pushbutton.
- 4. To specify aircraft effectivity details, select the aircraft model in the multiline and select the Edit Aircraft Effectivity link.
- 5. Select the Edit Mod # Effectivity link to modify the aircraft model details.
- 6. Select the View Maint. Operator Effectivity link to view the maintenance operator effectivity details.

# 2.1.16 Editing aircraft mod # details

Using this page you can edit the Mod # effectivity details for an aircraft that belong to a specific model.

1. Select the Edit Mod # Effectivity link provided at the bottom of the Edit Model Effectivity page. The Edit Mod # Effectivity page appears. *See Figure 2.12.* 

\star 🗎 Edit Mod #	# Effectivity				≭ 🖶 🛱 🗲 ? 🗔 🗖
- Task Details					
	Task # 1-A310-0000-MPD-00006396			Revision #	
	Task Description AIRCRAFT INSPECTION AS A	RESUL	Tas	k Applicability Aircraft	
	Aircraft Model # 737-200 ▼ Get Detai		Base Airo	craft Model # 737-200	
Mod # Effectivity D	etails Default Applicability				
44 4 1 -1/1	)			) 22 🐺 🖷 💷 🖬	۹ 🔻
# 🗏 Mod #	Mod Description	Applicability	Mod Classification	Mod Category	ATA #
1 🗆 afda	gasdgas	Not Applicable	~		
2			*		
<					>
		Edit Mod # Effect	ivity		

Figure 2.13 Editing Mod # effectivity details of task

- 2. Use the **Aircraft Model #** drop-down list box and specify the model number to which the aircraft belongs, in the **Task Details** multiline.
- 3. Click the **Get Details** pushbutton to retrieve Mod # Effectivity details pertaining to the aircraft.
- 4. In the **Mod # Effectivity Details** multiline, specify the **Default Applicability** and the **Applicability** of the Mod number on the aircraft. The system provides the options: "Pre", "Post", "No Impact" and "Not Applicable".
  - Note: The system loads the option "Not Applicable", when there is no applicability value defined for the Mod # Aircraft combination in the "Common Masters" business component.
- 5. Click the **Edit Mod # Effectivity Details** pushbutton, to modify the mod effectivity details of the aircraft.

# 2.1.17 Specifying component effectivity details for the task

You can define the component effectivity, to indicate the applicability of the task for the components.

1. Select the Edit Component Effectivity link in the Create Task Information page. The Edit Component Effectivity page appears. *See Figure 2.13.* 

) E	dit (	Componen	t Effectivit	y							≣ <i>7</i> \$	Ē		<b>⊢</b> ?	Γø
									Date & Time	Format yyyy-dd-mm		hh:m	im:ss		
Та	sk De	etails													
				Task # 1-COMP-0000	-9X-00000031					Revision #					
			Т	ask Description PME-1						Task Applicability Component	t				
– Pa	rt De	tails													
44 4		1 - 4 / 4 🕨	<b>&gt;&gt;</b> + •	- 0 % 0 0 T T					<b>7</b> 🗉 🖉 X 🖾 🖬 🛛	🗏 🖷 💷 🛛 🗛			Ŧ		_
#		Part # 🔎 🐂	-	Part Description	Prime Part Group Applicability		Comments	Created By	Created Date/Time	Last Modified by		Last M	Modified D	Date/Tim	e
1		WT104149:66	5503	<u> </u>	No	~									
2		WT 104270													
3		WT104316	'Part a	#' is hidden and	'Mfr. Part #' and										
4	۲	WT104316	1 Afr #	fielde ere dien	loved based on										
5					ayed based on										
		<	invent	ory options set i	in "Logistics										)
				· · · ·	siness component										
			Comm		Siness component		Edit	Effectivity							
ew Ma	iint. C	Operator Effect	ivity												

Figure 2.14 Specifying component effectivity of task

2. Enter the **Part #** in the multiline.

Or

- 3. Enter the Mfr. Part # and the Mfr. # of the part.
  - Note: The system displays the Mfr. Part # and Mfr. #fields for you to enter, and hides the "Part #" field above, if the parameter "Enable Manufacturer Part # control in transaction" under the Category 'Manufacturer Part #' is set as "Yes" in the "Set Inventory Process Parameters" activity of the "Logistics Common Master" business component.
- 4. Enter any additional **Comments** on the applicability of the task to the component.
- 5. Click the **Edit Effectivity** pushbutton.
- 6. Select the View Maint. Operator Effectivity link to view the maintenance operator effectivity details.
- 7. Select the **Record Task Part Mod Details** link at the bottom of the page to update part modification details for the effective tasks.

#### 2.1.18 Editing the access panel information

You can add access panels or modify the access panel details that have already been identified for the task.

1. Select the Edit Access Panel Details link in the Create Task Information page. The Edit Access Panel Information page appears. *See Figure 2.14.* 



Task Details     Task Applicabilit     Task 4     Task 4     Task 4     Task 4     Task 2     Task Descriptor     Task Descriptor	# 1-COMP-0000-9X-00000031	Select the aircraft model for which the access panels are being identified.	Revision # Aircraft Model # 000	Get Details
(i i 1 · 1 / 1 ) > + = □ ≤ ○ ○ 1	r T,		x 🗹 🖬 📽 🗣 🖷 💷 🗚	•
# Seq # Access Panel # P	Zone #	Access Panel Description		
1				
2 Indicate the sequence panels are to be oper				

#### Figure 2.15 Specifying access panel information for the task

- 2. Enter the access panel number in the **Access Panel #** field.
- 3. Click the Edit Access Panel pushbutton.

#### 2.1.19 Specifying the task references

You can enter the reference information for the task or sub task.

- 1. Select the Edit Task / Sub-Task References link in the Create Task Information page. The Edit Task References page appears.
- 2. Select the type of the reference document in the **Reference Doc Type** drop-down list box.
- 3. Enter the **Document ID** of the document containing details of the task.
- 4. Click the Edit References pushbutton.

# 2.1.20 Entering work area or zone information for the task

You can add or modify the work area or the zone where the task must be executed.

1. Select the Edit Work Area/ Zone details link in the Create Task Information page. The Edit Work Area / Zone Information page appears. *See Figure 2.15*.

	Edit	Work Area/ Zone Infor	rmation			1	7\$	-		+	?	0 K
-	ask I	Details		Select the entity, which you are	going to							
			Task Applicability Aircraft	define in the multiline, as "Work	Area" or							
			Task # 1-A310	"Zone".		Revision #						
			Task Description AIP									
			Work Area / Zone			Aircraft Model # 737-200 💌						
- w	ork	Area / Zone Details		Get Details								
		1 - 3 / 3 <b>&gt; &gt;&gt; + -</b>	0 * 0 0 T T			11 🗉 🗴 🖂 🖄 🗶 🖷 💷 🗛		•	•		-	Q
#		WorkArea / Zone # 🔎	Description	-			_					
1	E	1	Port Wing		Click this	pushbutton to display the						
2		1-H	1									
3		100	Lower half of fuselage t	o rear pressure bulkhead ( area below main cabin deck, including f	details air	eady defined for the Aircraft						
4	E				model for	the selected entity						
							J					
				Edit Work Area / Zo								
				Edit Work Area / 20	ine							

#### Figure 2.16 Entering work area or zone information for the task

- 2. Select the entity, which you are going to define in the multiline, as "Work Area" or "Zone".
- 3. Select the Aircraft Model # for which the work area or zone must be defined.
- 4. Enter the **Work Area / Zone #** field in the multiline, to specify the work area or zone where the task must be executed when performed on the selected Aircraft Model.
- 5. Click the Edit Work Area / Zone pushbutton.



# 2.1.21 Specifying parameter details for the task / sub-task

You can define or modify the parameter details for the task/sub-task.

1. Select the Edit Parameter Reading / Eval. Form link in the Create Task Information or Edit Task Information page. The Edit Parameter Reading / Conditional Maint. Eval. Requirements page appears. *See Figure 2.16.* 

Task / Sub-Task Details Task = 1+A310-00004PP-00005 Task Applicability Arcraft Task Description AIRCAFT INSPECTION A Sub Task Seq # 2 ♥ Get Details Parameter Reading Details Conditional Maint, Evaluation Conditional Maint, Evaluation		Eu	IIL Paralli	eter Reading		laint. Eval. Form								≣ ≭	특 다	-	?
#       B       Line #       Parameter       Parameter Recording?       Standard?       Parameter Description       LOW       Standard Value       Permitted Values       Min. Value       Max. Value         1       Image: Constraint of the standard value       FH       Mandatory       VES       Flight Hours       Ves       Select this link to retrieve technical and attribute parameter sector.         2       Image: Constraint of the standard value       Select this link to retrieve technical and attribute parameters       Select this link to retrieve technical and attribute parameters	,				Task Applicability Task Description Sub Task Seq #	Aircraft AIRCRAFT INSPECTION 2 Cet Details	Š	Std", if th	ne "S		is	Rev	ision #	GEAR CHECK			
1     FH     Mandatory     YES     Flight Hours       2     FH     Mandatory     YES     Flight Hours   Select this link to retrieve technical and attribute parameters	44	1	1 - 1 / 1	> > + =	0 % 0 0	T Tx	_			J	L II 🛛	X Z 🗎 X C 🖡 🕯	AI	Ŧ			Q
2 Mandatory YES Select this link to Select this link to attribute parameters	#		Line #		Parameter Type	Parameter Recording?		Standard?			UOM	Standard Value	Permitted Values	Min. Value	Max. Va	lue	
Select this link to retrieve technical and attribute parameters	1			FH					*	Flight Hours							
Help on Consumption & Range Parameters Help on Technical & Attribute Parameters			-	Range Parameters	retrie	ct this link to	tion		*	Help on Technica	& Attribut	e Parameters	retrieve tec	hnical and			>

Figure 2.17 Entering parameter details for the task / sub-task

- 2. Use the Sub Task Seq # drop-down list box to specify the sequence for the sub-task.
- 3. Click the **Get Details** pushbutton to retrieve the parameter details for the task number, base aircraft model number and sub task sequence number combination.
- 4. Select the Parameter Reading Details tab in the Edit Parameter Reading / Conditional Maint Eval. Form page. See *Figure 2.16*.
- 5. Enter the **Parameter** of the task/sub-task.
- 6. Use the **Parameter Reading?** drop-down list box to specify whether recording of parameter reading for the task/sub-task is mandatory or not.
- 7. Set the **Standard?** field to "Yes" if the parameter is a standard parameter. Select "No" otherwise.
- 8. Enter the Standard Value, Permitted Values, Min. Value and Max. Value of the parameter.
- 9. Select the Conditional Maint. Evaluation tab in the Edit Parameter Reading / Conditional Maint Eval. Form page. See *Figure 2.17*.

•	Paramet	er 🔪	Trigger Value	Permitted Values	Trigger Value (Min.)	Trigger Value (Max.)	Follow-up Action	Seq #	Follow-up Task # 🔎	Records Follow-up Instructions
		>					None		1-A310-0000-CMM-	
		*						•	2-00-B7-01	
		-						•	2-00-B7-03	
				e paramete						
	<		of the ta	isk / sub-ta	sk 🛛					

Figure 2.18 Specifying follow-up action for task

10. The name of the parameter in the usage-based schedule in the **Parameter** field.

Specify the following for the parameter in the multiline.

- 11. Trigger Value: The parameter value at which the follow-up action must be executed.
- 12. **Permitted Values:** The permitted values for the parameter. You can specify multiple values separate by a separator, such as",", ":" or "/".
- 13. Trigger Value (Min.): The minimum parameter value at which the follow-up action must be



executed.

- 14. **Trigger Value (Max.):** The maximum parameter value at which the follow-up action must be executed.
- 15. **Follow-up Action:** Use the drop-down list box to select the follow-up action for the task on attainment of one of the following:
  - The trigger value
  - Any of the values in the Trigger Value (Min.) Trigger Value (Max.) range.

The drop-own list box displays: None, Initiate Schedule, Terminate Schedule, Initiate Immediate Exec. and Initiate Records Follow-up.

- 16. Specify the task that must be executed as part of the follow-up action in the Follow-up Task # field.
- 17. Specify a set of actions, tasks or procedures that must be carried out as part of "Initiate Records Follow-up" follow-up action on attainment of one of the following in the **Records Follow-up Instructions** field
  - The trigger value
  - Any of the values in the Trigger Value (Min.) Trigger Value (Max.).
    - Note: This field is mandatory, if you have selected "Initiate Records Follow-up" as the Follow-up Action.
- 18. Specify the base model # of the follow-up task in the Follow-up Task Base Model # field.
- 19. Click the **Edit Parameter Readings** pushbutton to add or modify the parameter details for the task/sub-task.

#### 2.1.22 Creating dependent systems condition list

Dependent System Condition (DSC) list describes the condition or the required state of the various related systems for executing a maintenance task.

1. Select Create Dependent Systems Condition List under Maintenance Task business component. The Create Dependent System Condition Information page appears. *See Figure 2.18.* 

Create Dependent System Condition Information			≣ ≭ ∰ ⊄ <b>?</b> ©
- DSC Details			
DSC # 160515		Copy From P TEST4	
DSC Description Condition for Overhauling			
Dependent Systems Desired Condition Details			
<f +="" -="" 1="" 3="" <="" h="" q="" t<sub="" y="" →="" □="">k</f>			۵ 🔻
# 🖹 Seq # Dependent System	Desired Condition		
1 🗵 1 DS1	New		
2 🖻 2 DS2	Serviceable		
3 🗇 3 DS2	Unserviceable		
4 Re - Number		Describe the desired condition of the dependent system, before executing a maintenance task	
Reference Document Details     Ref. Document No. 1     Ref. Document No. 2		File Name 1 🖉	View File View File
	Create DSC		

Figure 2.19 Creating dependent system condition

- 2. Enter the **DSC #** to uniquely identify the DSC.
- 3. Enter the DSC Description.
- 4. Enter the dependent systems based on which task is performed, in the Dependent System field in

the multiline.

5. Click the **Create DSC** pushbutton.

### 2.1.23 Authorizing tasks

Tasks can be used in planning and execution documents only after they are activated. You can either activate or cancel the tasks that have been created and are in "Fresh" status.

1. Select Authorize Task under Maintenance Task business component. The Authorize Tasks page appears. *See Figure 2.19.* 

* [	D, I	Authorize Tasks									← ?	Co
50	arch	Criteria				criteria to retrieve st be activated or		Date Format yyyy-d	ld-mm			
			Task # Task Description Task Applicability ATA # Work Center # Created By		eled.	Search	Base Ai	Fask Category craft Model ≢ Task Type erations Type REPAI WBS Code	R STATION			
-) Se		Results 1 - 5 / 500 ►	» + 🗆 T T,				,		Al	Ŧ	_	_
÷		Task Applicability	Base Aircraft Model #	Task #	Revision #	Task Desc.	Operations Type	Status	Revision Date		Task T)	pe
1		Component		1-ROUTINE-0000-MPD-00005059		PME 2	REPAIR STATION	Fresh			MPD	
2		Component		1-ROUTINE-0000-MPD-00005060		PME 3	REPAIR STATION	Fresh			MPD	
3		Component		1-ROUTINE-0000-MPD-00005061		PME 4	REPAIR STATION	Fresh			MPD	
4		Component		1-ROUTINE-0000-MPD-00005062		PME 5	REPAIR STATION	Fresh			MPD	
5		Component		2-ROUTINE-0000-MPD-00006174		Intermediate 1	REPAIR STATION	Fresh			MPD	
		<										>
		ze Related Task		Maintain Rep		Authorize Task(s)	Return Task(s)	ze Maint. Prog. & Upda	te Compliance			

Figure 2.20 Authorizing a task

- 2. Enter the search criteria to retrieve the tasks that must be activated or canceled.
- 3. Enter the Authorization Comments, if you wish to authorize the task.
- 4. Select the **Authorize Related Task** check box to enable the system to authorize related tasks in addition to the parent task.
- 5. Select the tasks that must be activated and click the Authorize Task(s) pushbutton.
  - Note: The system performs the following:
  - a. Activates the selected tasks and sets the status of the tasks as "Active".
  - b. Inactivates all the previous revisions of the task, if the "Inactivate Previous Revisions?" check box is selected.
  - c. Consequent to authorization, the parts for which the task is effective are automatically mapped to the sequence control # to which the task has been mapped, if the WBS code for the task has the "Repair Scheme process parameter set to "1".
  - d. If you have selected the "Authorize Related Tasks" check box, the system authorizes related tasks in addition to the primary task.
  - e. When a new parameter is added for the task, the system updates the parameter details to the AMP, CMP, Model Program and Part Program, if the added parameter is defined for the effective aircraft, component, model and part, respectively.
  - *f.* Computes and updates NSD / NSV for the tasks. Refer to the flow diagram in the "<u>NSD / NSV Computation</u>" section for more details.

# 2.1.24 Returning a task

You can return the tasks that you do not wish to authorize, back to the initiator.

- 1. Repeat steps 1 and 2 under Authorizing a task.
- 2. Select the tasks that must be returned and click the Return Task(s) pushbutton.
  - Note: The system returns the selected tasks and sets the status of the tasks as "Returned".

To proceed,

- $\tau$  Select the **Edit Task Information** link to modify the task details.
- $\tau$  Select the **Maintain Repair Scheme** link to create a repair scheme for the tasks.

#### 2.1.25 Managing unified task sequence

This process enables you to create/update the definition of a sequence control. A sequence control is identified by sequence control # and defines the execution order for a set of component tasks. More precisely, a number of child tasks related to several tasks can be sequenced to be executed in a specific order on a maintenance object. The sequence control # essentially acts as a clue for a shop mechanic in the maintenance. A mechanic can identify the exact order in which a set of tasks must be executed on a part/component by its sequence control #.

To create sequence control #, you must the following,

- Associate parts to the sequence control #
- Associate tasks to a sequence control #. (These tasks are called root tasks for which repair schemes must have been defined using the Maintain Repair Scheme activity.)

Sequence related tasks of all the root tasks in a specific order for execution on parts/components in a shop. You can also modify the above details for an existing sequence control in future. However, a part and task can be associated with only one sequence control # that is in "Active" status at any point of time. Note that this activity does not allow for sequencing of tasks applicable to aircraft.

# 2.1.26 Managing unified task sequence

1. Select the Manage Unified Task Sequence activity in Maintenance Task component. The Manage Unified Task Sequence page appears. See *Figure 2.20*.

Ma	nag	je Unified Task Se	equence						= c		+ ? 🗔
		isk Seq. Control #							Date Format yyyy-dd	-mm Time Forma	at hh:mm:ss
Denn	ie Ta	Seq. Control # Seq. Control #	# P 10010	×	Seq. Control Des	с.			Status Active	Ŧ	
					Oreate New Sequence Control #	C Edit Existing Sequence	ence Control #				
Мар	Par	ts to Seq. Control	# Map Tasks To S	eq. Control # S	equence Tasks						
-	•	1 -3/3 • •	+ - 0 +	0 0 T T,		<b>ж</b> ш		AL III AL	T		Q
#		Part # 🔎	Remarks 📝	Prime Part ?	Part Description	Created Date	Created By	Last Modified Date	Li	ast Modified by	
1		WC256-2:35895									
2		WC256-5:35895									
3		WC256-6:35895									
4											
		<									>
					Map Parts to	Seq. Control #					
View As	socia	ted Repair Schemes									



#### **Creating sequence control #**

- 2. In the Define Task Seq. Control # group box, enter the sequence control that you want to create and record details, Mandatory. If the "Sequence Control Generation Mode" is set as "Auto" in the Set Options of the Maintenance Task component, the system automatically generates the sequence control # for the new sequence on successful mapping of the parts to the sequence control. However, it is mandatory for the user to specify the sequence control #, if the "Sequence Control Generation Mode" is set as "Manual". Further, enter the following.
- 3. Enter the **Seq. Control Desc.** of the sequence control for which you want to record details. Mandatory.
- 4. Use the **Status** drop-down list box to select the status of the sequence control # that you wish to modify. The drop-own list box displays Active and Inactive.
- 5. Select Create New Sequence Control # radio button.
- 6. Select the Map Parts To Seq. Control # tab to record.
- 7. Select the Map Tasks to Seq. Control # tab.
- 8. Select the **Sequence Tasks** tab.

#### 2.1.27 Mapping parts to sequence control #

1. Select the Map Parts To Seq. Control # tab in the Manage Unified Task Sequence page. See *Figure 2.21*.

Мар	Part	ts to Seq. Control #	Map Tasks To Se	q. Control # Sequ	ence Tasks					
44	4	1 - 3 / 3 > >>	+-0*	0 0 T T,		人间(		AL III AL	<b>v</b>	Q
ŧ		Part # 🔎	Remarks 📝	Prime Part ?	Part Description	Created Date	Created By	Last Modified Date	Last Modified by	
		WC256-2:35895								
		WC256-5:35895								
		WC256-6:35895								
		<								>
		•								-
					Map Parts	to Seq. Control #				
w As	socia	ted Repair Schemes								

Figure 2.22 Mapping parts to sequence control

- 2. In the multiline, enter the Part # to which you want to associate the sequence control #. The part you specify must be Active and valid as defined in the Part Administration component. The part must be effective to at least one of the tasks mapped to the sequence control # in the Map Tasks To Seq. Control # tab. Further, enter the following.
- 3. The Mfr. Part # and the Mfr. # of the part to which you want to associate the sequence control number.

#### To proceed,

 $\tau$  Select the **View Associated Repair Schemes** link to see details of repair schemes associated with the sequence control #.

#### 2.1.28 Mapping tasks to sequence control #

1. Select the Map Tasks to Seq. Control # tab in the Manage Unified Task Sequence page. See Figure 2.22.



· ·	•	1 -3/3 > >>	+-0%04	Tx		1	トーロメロ目	328 👎 🖷 💷 🗛	•	
ŧ		Task # 🔎	Task Description	WBS Code	Remarks 📝	Created Date	Created By	Last Modified Date	Last Modified by	
		1-A330-3450-ACC-	ACCEPT CHECK	1-PME		11/11/2011 9:31:36 PM	SCHELLAMUTHU			
		1-A330-3450-OV-	OUT VENDOR (OH)	1-PME		11/11/2011 9:31:36 PM	SCHELLAMUTHU			
		1-A330-3450-OV-	OUT VENDOR (RP)	1-PME		11/11/2011 9:31:36 PM	SCHELLAMUTHU			

#### Figure 2.23 Mapping tasks to sequence control

- 2. In the multiline, the Task # of the task that you wish to associate with the sequence control #.
- 3. Select the Map Tasks To Seq. Control # pushbutton.

#### Deleting task from the sequence control #

- 4. Select the box from the multiline for the task that you wish to delete.
- 5. Select the 🖃 icon above the multiline

The task as well as its related tasks (listed in the Sequence Tasks tab is disassociated from the sequence control #.

To proceed,

- τ Select the **Maintain Repair Scheme** link to record/modify details of the repair scheme associated with the task that you have selected in the multiline.
- $\tau$  Select the **Manage Task Effectivity** link to modify effectivity definition for tasks mapped to the sequence control #.
- $\tau$  Select the **View Associated Repair Schemes** link to see details of repair schemes associated with the sequence control #.

# 2.1.29 Sequencing tasks

1. Select the Sequence Tasks tab in the Manage Unified Task Sequence page. See Figure 2.23.

44 4		1 - 5 / 5 🕠	» + □ ¢	🔅 T T,					▶ ๒ ७ ४ ๔ 首 ∞ е	📮 🕸 💷 🗛	•	
#		Rep. Seq. #	Prev. Rep. Seq. #	Task #	Task Desc.	Task WBS Code	Remarks 📝	Operator #	Parent Rep. Scheme #	Parent WBS Code	Root Rep. Scheme #	Root WBS Code
		400	400	3-00034847	COMPLETE	3-OPERATION		Generic	2-A330-3450-STD-00000021	2-TASK	1-A330-3450-OV-00000006	1-PME
		200	200	3-00034848	PREPARE	3-OPERATION		Generic	2-A330-3450-STD-00000021	2-TASK	1-A330-3450-OV-00000006	1-PME
		100	100	3-00181910	ACCEPTCK	3-OPERATION		Generic	2-A330-3450-STD-00000012	2-TASK	1-A330-3450-ACC-00000001	1-PME
		300	300	3-00442111	PREPARE	3-OPERATION		Generic	2-A330-3450-STD-00000011	2-TASK	1-A330-3450-OV-00000001	1-PME
	E	500	500	3-00442112	COMPLETE	3-OPERATION		Generic	2-A330-3450-STD-00000011	2-TASK	1-A330-3450-OV-00000001	1-PME
		<										

#### Figure 2.24 Sequencing tasks

2. Use the **Display Option** drop-down list box to select the display option for records in the multiline. The drop-down list box displays the following: By Repair Sequence # and By Task. Select
- ➤ "By Task" to display all related tasks of the tasks mapped to the sequence control # in the "Map Tasks To Seq. Control #" tab. in the order of the task #.
- ➤ "By Repair Sequence #" to display all repair sequence #s for the sequence control # in the order of the sequence number.
- 3. Select the **Eliminate Duplicate?** check box to disallow repetition of same Task # Rep. Seq. # in multiline. This check box is enabled only if you have selected "By Repair Sequence #" as the display option. Select the **Eliminate Duplicate?** check box to display only one instance of the "Rep. Seq # and "Task #" combination against the sequence control #. To retrieve all repair sequence #s for the sequence control #, do not select the check box.
- 4. Select the **Get Details** pushbutton to displays details of the sequence control # in the multiline.
- 5. In the multiline, enter the **Rep. Seq. #** to indicate the order of execution of the task on the part, Mandatory. Further, enter the following.
- Use the Seq. Control? drop-down list box to indicate the requirement of sequence control for tasks. The drop-down list box displays the following: Required", "Not Required", "Across Phases" and "Within Phases".
  - Note: If the WBS code of the task has the "Sequence Control definition allowed?" process parameter set to "Yes", you can set this field to "Required", "Across Phases" or "Within Phases". On the contrary, if this process parameter set to "No", you must set this field to only "Not Required".
- 7. Use the **Seq. Modification** drop-down list box to indicate whether sequence modification for tasks is allowed. The drop-down list box displays the following: "Allowed" and "Not Allowed".
- 8. Select the **Re-Sequence** pushbutton.

Based on the multiplication factor specified in the Set Options activity, the system generates new repair sequence numbers in the **Rep. Seq. #** column. The system generates repair sequence numbers for some or all of the tasks.

9. Select the **Update Seq. Number** pushbutton.

The system saves the repair sequence numbers against the sequence control #. It also updates the repair sequence #s, if modified. If modified for the first time, the old repair sequence #s are saved in the Orig. Seq. # field. The Orig. Seq. # field retains these first time values and, further changes to the repair sequence # do not impact Orig. Seq. #s. The repair sequence numbers for tasks are also updated in the repair schemes of the root tasks.

#### **Modifying sequence control #**

- 1. In the **Define Task Seq. Control #** group box, enter the **Seq. Control #** and the **Seq. Control Desc.** for which you want to modify details, Mandatory. Further the of the sequence control for which you want to record details. Mandatory. Further, enter the following.
- 2. Use the **Status** drop-down list box to select the status of the sequence control # that you wish to modify. The drop-own list box displays Active and Inactive.
- 3. Select **Edit Existing Sequence Control #** radio button. The available details of the sequence control # are retrieved by the system.
- 4. To modify these details, follow the procedure explained for creation of sequence control #.

## 2.1.30 Recording Task Part modification details

Part modification involves upgrade of parts with superior properties on the basis of Mod # that is provided in Service Bulletins and CMM. The Maintain Task Part Mod Details page can be used to



create, modify and view Mod # details against the effective parts for the given Task. In this activity you can map Mod # for the part with the effective task. In other words, you specify the task that must be executed on the part in order to comply with Mod #.

1. Select the Maintain Task Part Mod Details link under the Maintenance Task business component. The Maintain Task Part Mod Details page appears. *See Figure 2.24.* 

						1011		5				
* 1	4aintain Ta	sk Part Mod D	etails								States 🗸 🎞 🖶	; ⇔
	earch Criteria											
		Task #/Rev. # 👂				Part # 👂				Source Doc. #		
						Get Details						
	ask Part Mod [											
44 4	1 - 10 / 3	.0 → → ==								All	<b>v</b>	Q
#	🗖 🛛 Task # 🛔	D Rev. #	Task Description	Part # 🔎	Mfr. Part # 🔎	Mfr. # 🔎	Part Description	MSN - From 🔎	MSN - To 🔎	New Mod #	Conditional Effectivity	
1												
2												
3												
4	8											
5												
6												
7												
8												
9												
10												
	<											>
						Save						
						Save						

Figure 2.25 Recording Task Part Modification details

- 2. In the Task Part Mod Details multiline, enter Task # and Rev. # that is effective to the part.
- 3. Enter Part #, Mfr. Part #, Mfr. # for the part on which modification must be carried out.
- 4. Enter MSN From and MSN To for the range of parts on which the modification process must be executed.
- 5. Enter New Mod # that identifies the modification transaction on the part.
- 6. Enter **Conditional Effectivity** that explains the user-defined effectivity for the part modification.
- 7. Click the Save pushbutton.
- 8. Note: The **Save** pushbutton is not available, if the page has been accessed by clicking the **View Task Part Mod Details** link.

## 2.1.31 Maintaining activated tasks

#### Modifying a task without creating a new revision

You can modify the contents of an activated task without creating a new revision.

- 1. Select the Maintain Activated Tasks link under the Maintenance Task business component. The Select Task page appears.
- 2. Enter the task directly and select the Edit Task Information link.

Or

- 3. Select a task by carrying out a search based on filter criteria. Click the hyperlinked task in the multiline. The **Edit Task Information** page appears. See *Figure 2.25*.
- 4. Enter Task Description.
- 5. Use the **WBS Code** drop-down list box to specify the Work Breakdown Structure (WBS) Code indicating the attributes that identify whether the task is executed for planning purpose or execution purpose.
- 6. Specify ATA #, Est. Man Hrs., and Est. Elapsed Time.
- 7. Use the **Output Qty. Type** drop-down list box to select the type of output quantity from the task. The drop-down list box displays Fixed and Variable.
- 8. Use the **Emp. Loading** drop-down list box to select the time for employee loading for the task. The drop-down list box displays Elapsed time and Wrench time.

			Date Format	vvvv/mm/dd
Task Details				
Task Details	Component Ente	r the task	Base Aircraft Model #	
	1-50C-0000-CMM-0000645 desc	ription	Current Revision #	
				A structure
			Status	
Task Category	50C 🔻		Change Status To	Active 💌
TaskDesc	-			
Task Lng Desc1				
			•	
Work Center #	Q	Specify the Work Breakdown	Operations Type	FLIGHT OPS -
Model Effectivity		Chrysters (MDC) and to	WBS Code	1-PME -
Revision Details		Structure (WBS) code to		· · · · · · · · · · · · · · · · · · ·
Current Revision Type		identify the task	Current Revision Date	
New Revision #			New Revision Type	<b>v</b>
New Revision Date			Inactivate Previous Revisions?	
				_
Comments				
Task Additional Details				
Non Standard Task #			Task Desc.	
# ATA	00-00		System ATA # Description	GENERAL - AIRCRAFT
DSC #	Q	Enter the ATA chapter	DSC Description	
Est. Man Hrs.	3.00		Insp. Man Hrs.	
4				•
				•
Re - Number		View File		
Reference Details				
Maint. Manual Ref. #			MPD Item #	
Source Document #	3345		Source Document Type	Others 💌
Manufacturer #	۵		Manufacturer Name	
Work Location Ref. #			Eng. Doc. Eff. Date	
Mandatory?	<b></b>		Reliability Related?	
Eng. Doc. Class	▼ ▼		User Defined 1	<b>▼</b>
User Defined 2	•		EO User defined 4	<b></b>
EO User defined 5			EO User defined 4 EO User defined 6	
			EU User defined 6	<b>•</b>
+ Document Attachment Details				
		Edit Task		
11-1-7-5-				
	PrtReg/CEval/			
Edit Parts Requirement		n-Off Requirements	Edit Access Panel Det	
Edit Model Effectivity	Edit Component I	Effectivity	Edit Schedule Informa	
Edit Work Area / Zone Edit Aircraft Effectivity	Edit Notes	eading / Eval. Form	Edit Task / Sub-Task Maintain Repair Scher	
Edit Aircraft Effectivity Manage Task Effectivity	Edit Parameter R View Customer E		Maintain Repair Schei Manage Task File Atta	
	view customer c	in contrary	manage rask the Ada	ACTIVITY IN THE REPORT OF A DECEMBER OF A

9. Click the Edit Task pushbutton.

Figure 2.26 Maintaining activated tasks

## 2.1.32 Revising an activated task with a new revision

You can modify the contents of an active task and create a new revision.

- 1. Use the Edit Task Information page of the Edit Task or the Maintain Activated Tasks activity to modify the task details.
- 2. Enter the **Revision Details** for the task.
- 3. Click the **Edit Task** pushbutton.
  - Note: The system creates a new revision for the task and automatically activates the same.

To provide further details,

- τ Select the **Edit Parts Requirement** link to specify parts for carrying out the task.
- $\tau$  Select the **Edit Resource/Sign-Off Requirements** link to specify resources and sign-off information for carrying out the task.
- $\tau$  Select the **Edit Schedule Details** link to specify the schedule details for the task.
- τ Select the Edit Model Effectivity link to specify the task applicability to the model.
- $\tau$  Select the **Edit Component Effectivity** link to specify the task applicability to the component.
- $\tau$  Select the **Edit Access Panel Details** link to specify the access panel information for the task.
- $\tau$  Select the **Edit Task / Sub-Task References** link to specify the reference document details for the task.
- $\tau$  Select the **Edit Work Area / Zone Details** link to specify the work area and the zone where the task has to be executed.
- $\tau$  Select the Edit Access Panel Details to specify access panel details for the task.
- $\tau$  Select the **Edit Parameter Reading / Eval. Form** link the page to add/modify the parameter details that are required for the task/sub-task.

Refer to the topic, Creating tasks, for more details.

- $\tau$  Select the **Manage Task Effectivity** link to record/modify effectivity definition for the task.
- τ Select the Maintain Repair Scheme link to create/modify repair scheme for the task.
- τ Select the Manage Task File Attachment link to associate multiple files to a task for printing.
- $\tau$  Select the **Record Task Part Mod Details** link at the bottom of the page to update part modification details for the task.

#### 2.1.33 Deactivating a task

- 4. Use the Edit Task Information page of the Maintain Activated Tasks activity and set the Change Status To field to "Inactive".
  - Note: The system inactivates the specific revision of the task.

#### 2.1.34 Reactivating a task

You can re-activate an inactive task.



1. Use the Edit Task Information page of the Maintain Activated Tasks activity and set the Change Status To field to "Active".

>>> Note: The system activates the specific revision of the task

## 2.1.35 Maintaining task relationship

Through this activity, you can define relationship between a primary task and the related tasks. The relationship is defined between two tasks depending on the significance of a specific stage of a primary task with respect to the other related tasks. A relationship can be classified as "Regular" or "Schedule" to simplify the retrieval and viewing the relationships.

1. Select the Maintain Task Relationship link under the Maintenance Task business component. The Maintain Task Relationship page appears. See *Figure 2.26*.

<b>k</b>		Maintain Task Relationsh	lip					1	] ≍ = ¤ ← ?	C¢.
							Date	& Time Format yyyy-dd-mm hh:mm:ss		
-) 56	earch	n Criteria								
		Search	On Primary Task #			Additional Sear	ch On	<b>v</b>		
		Display Opt	ion Regular Relationship	•		Relatio	nship Planning ×	▼ Block Schedule ▼		
					Searc	h				
Та	ask R	lelationship Details								
44	4	1 - 10 / 50 > >> + -	0 × 0 0 T T				人口回义区首	24 C 👎 🖷 💷 Al	T	
#		Primary Task # 🔎	Task Description	Relationship		Seq #	Rel. Task # 🔎	Rel. Task Description	Relationship Categor	nry
1		1-00-86	Inspection 1	Block Schedule	~	1	3-00-10	Inspection Operational 1		
2	E	1-00-86	Inspection 1	Block Schedule	~	2	3-00-11	Inspection Operational 2		
3	E	1-00-86	Inspection 1	Block Schedule	*	3	01-PME-034534500001	test		
4	E	1-00-87	Inspection 1	Block Schedule	*	1	3-00-13	Inspection Operational 3		
5	E	1-00-87	Inspection 1	Block Schedule	×	2	3-00-15	Inspection Operational 4		
6	E	1-00-88	Inspection 1	Block Schedule	*	1	3-00-16	Inspection Operational 5		
7	E	1-00-88	Inspection 1	Block Schedule	*	2	3-00-17	Inspection Operational 6		
8	E	1-50C-0000-CMM-00000536	PME-1	Block Schedule	~	1	3-10000359	Engine Check 1		
9	E	1-50C-0000-CMM-00000536	PME-1	Block Schedule	<b>~</b>	3	3-10000361	Engine Check 4		
10	E	1-50C-0000-CMM-00000536	PME-1	Block Schedule	~	4	3-10000362	Engine Check 3		
		<								>
_										_
					Maintain Task R	elationship				
ainta	in Re	pair Scheme			M	anage Task File	Attachment			

Figure 2.27 Maintaining task relationship

#### **Creating new relationship between tasks**

Enter the following in the Task Relationship Details multiline:

- 2. Enter a unique identifier of the primary task in the Primary Task # field.
- 3. Select a value from the Relationship drop-down list box or Schedule Relationship drop-down list box.
- 4. Enter a unique identifier of the related task in the **Rel. Task #**. This field is mandatory if any Relationship or Schedule Relationship has been selected.
- 5. Use the **Sch. Rel. with Predecessor Task** drop-down list box to set the schedule relationship of the task with its predecessor in a work order.
- 6. Enter any follow-up instructions required in the **Follow-Up Instructions** field. This field is mandatory if you selected the relationship as "Initiate Records Follow Up".
- 7. Click on the Maintain Task Relationship pushbutton.

#### Retrieving / editing existing relationship between tasks

- 1. Provide **Search Criteria** to retrieve the relationship that has been defined between primary task and the related tasks.
- 2. Click on the **Search** pushbutton.
- 3. Based on the search criteria, the system retrieves the relationship details in the **Task** Relationship Details multiline,



- 4. You can edit the details, if required. Note that you cannot edit the primary task # while modifying the details.
- 5. Click on the Maintain Task Relationship pushbutton.

## 2.1.36 Scheduling relationships for execution phases

The sub process allows you to define schedule relationships between two execution (primary and secondary) phases of aircraft maintenance or component maintenance process.

1. Select the **Execution Phase Schedule Relationships** link under the **Maintenance Task** business component. The **Execution Phase Schedule Relationships** page appears. See *Figure 2.27*.

* 8	Ex	ecution Phase Schedule Relationships			RAMCO OU-ra	amco role 🔻 💢 🖨 🛱 🗲 ? 🗔 🗷
- Exe	ution I	Details				
			Execution Type Aircraft Maintenance	<ul> <li>Get Details</li> </ul>	Package Type Log card 💌	
- Exe	ution I	hase Schedule Relationships				
•		1 - 4/4 🕨 🕨 🛨 🗖 🗇 🛠 🇭 🏋 🏹			人 血 ⑤ 図 図 図 単 ※ 図 単 ※ III 14 % AII	<ul> <li>Search</li> </ul>
#		Primary Execution Phase	Secondary Execution Phase		Schedule Relationship	
1		C0/Pre-Induction	✓ C1/NDT	~	Finish-Start	~
2		Post Flight	✓ Post Flight	~	Start-Start	*
3		Post Flight	<ul> <li>Preparatory</li> </ul>	~	Start-Finish	*
4		Post Flight	V C1/NDT	~	Finish-Finish	~
5			~	~		~
		4				,
				Save		



- 2. In the **Execution Phase Relationships** multiline, use the **Primary Execution Phase** drop-down list box to select the primary execution phase for the schedule relationship.
- 3. Use the **Secondary Execution Phase** drop-down list box to select the secondary execution phase for the schedule relationship.
- 4. Use the **Schedule Relationship** drop-down list box to select the type of schedule relationship between execution phases.
- 5. Click the **Save** pushbutton to update the schedule relationship details.

## 2.1.37 Managing repair scheme definition

1. Select the Manage Repair Scheme Definition link under the Maintenance Task business component. The Manage Repair Scheme Definition page appears. *See 2.28.* 

★ 🗎 Manage Repai	r Scheme Definition										⊢ ?	
Search Criteria	Primary Ta Maint, Obje WBS ( Task Cate	ect #	to crea map th	his data link te a task and e task to the nce control #.	Click th to edit t	is data link he task.	Task Description Seq. Control # Rep. Sch. Op. # Task Type	CMM	<b>v</b>			
1 - 7 / 7					<b>`</b>	- \ / A		₽ AI	•			Q
Repair Task #	Task Description	Rep. Seq. #	Part #	Rep. Sch. Op. #	New?	Child Node? Edit		Authorization?	W/C #	Seq. Ctl	.?	
CMP-1	Restoration and Completion					Create Child Node Edit	Task Create Unit		YUL-500-04	ł		
		Click this that woul	d be a r	k to create a tas elated task in the the repair task.		Greate Child Nodi Edit Greate Child Nodi Edit Greate Child Nodi Edit	Task Create Unifi	Click this				y
<												>

#### Figure 2.29. Managing repair scheme definition

- 2. Specify fields in the Search Criteria group box to retrieve repair tasks.
- 3. Click the **Search** pushbutton to retrieve details of tasks as well as their repair schemes in the **Primary Task Details** multiline displays.
- 4. Click on the data link in the **Repair Task #** column to create/modify details of the repair scheme for the task.
- 5. Select the **Create New** link in the **New?** column to create a new task and map the task to the sequence control.
  - Note: The data link appears in this column for blank records or if you have retrieved a specific sequence control # through search. Now you can proceed to create a task and automatically map the task to the sequence control.
- 6. Select the **New Child Node** link in the **Child Node** column to create a task, which would be a related task in the repair scheme of the repair task.
  - Note: The data link is available only if a repair scheme exists for the repair task. Typically, you use this link to create related tasks and map these to repair schemes.
- 7. Select the **Edit Task** link in the **Edit** column to modify information on the task.
- 8. Click the data link in the **New Seq. Ctl #** to modify sequence control for the repair task. However, if the task is not associated with a sequence control, the **Create Seq. control #** link appears here.
- 9. Click the data link in the **Authorization?** column, if you wish to authorize the repair task. On authorization, the task is automatically mapped to the sequence control #.

#### Tree structure in the multiline and its purpose

In the **Primary Task Details** multiline, if a repair scheme is available for the task in the Repair Task # column, data hyperlink appears in the column. You can click the link to access the **Maintain Repair** 

Scheme page and update its details. Conversely, click the B icon the appearing on the left to display the related tasks in the repair scheme. Each of the following rows displays a related task in the repair scheme. A related task in turn could be associated with a repair scheme, which can be identified by the B icon. The entire hierarchy of the maintain scheme of the repair task is represented in the multiline using a tree structure as shown in the picture below. The topmost node holds the repair task/root task while the child nodes display the related tasks. A child node may be expanded further to display related tasks and so on. However, if no repair scheme is available for the task, the tree is not displayed in the multiline. *See 2.29*.



Figure 2.30 Managing repair scheme definition

2.1.38 Maintaining repair scheme



This page allows you to create a repair scheme for the tasks by grouping the tasks based on the need of users, MRO, operator, etc. Grouping of the tasks can be done at model level or part level. You can specify the sequence in which the tasks must be executed. Using this page you can perform the following:

- Create a repair scheme by grouping the tasks based on need of users / MRO need / operator need.
- Group tasks at part level / model level.
- Perform master sequencing of tasks.
- Automation of the tasks.
- Eliminate duplication of sequence number of task.

You can insert tasks in the multiline and assign sequence number for them, by clicking the "Re Number" pushbutton to rearrange the existing rows in the ascending order of the sequence number. The system regenerates the sequence number in multiples of five, thereby enabling you to specify the sequence number for the newly added task.

1. Select the Maintain Repair Scheme link in the Maintain Task Relationship, Edit Task or Maintain Activated Tasks page of the Maintenance Task business component. The Maintain Repair Scheme page appears. See *Figure 2.30*.

Tak # 2008705     Description     Acheck (MPD 6787-200)        Description       Beschiption acheck (MPD 6787-200)    Effectivity Control Specific Table arrant Model #  Get Details  Get Details  Get Details  Control of the related task number  Repair Classification  Repair Scheme details  Control of the related task number  Repair Scheme details  Click this pushbutton to  Terrange the existing rows  In the ascending order of the sequence number.  Click this pushbutton to  retrieve child tasks	Maintain Repair Scheme								그: 틈		+	? 🗔
Tak # 2-08-705       Base Model # 8757-200         Description A-check (MPD-8767-200)       Operator # 12 ▼         Effectivity Control Speedic ▼       Operator # 12 ▼         Maint. Object # Avraft Model #       Get Details         Catal Control Speedic ▼       Repair Classification         Repair Agency # P       Pinne Part?         Mandatory?       No         Repair Agency # P       Enter the related task number         Repair Scheme details       Task # P         Task # P       Task # P         Click this pushbutton to rearrange the existing rows in the ascending order of the sequence number.       On-Wing ×         Click this pushbutton to retrieve child tasks       Click this pushbutton to retrieve child tasks	Primary Task Details				D	ate Format	vyyy-dd-mm					
Work Center # YUL-100-05       Repair Agency # P         Repair Agency # P       Repair Agency # P         Repair Agency # P       Enter the related task number         Mandatory?       No         Remarks       Fine Part?         TAT(Cah. Days)       TAT(Cah. Days)         Repair Scheme details       Im III AI         Image: Repair Seq. # Task # P Task #	Task ≠ Description Effectivity/Control Maint. Object ≠	A-Check (MPD-B767-200) Specific 🔍	Get Details									
Image: Seq. #       Prev. Rep. Seq. #       Task #       P       Child Position #	Work Center ≠ Repair Agency ≠ ₽ Repair Basis Mandatory? Remarks	Enter the related task	k number		Repair Pro	ocess Code Prime Part?		T		Ŧ		
#       Rep. Seq. # Prev. Rep. Seq. # Task # P       Task       Rebainship       Child Poston # Child Part # P       Job Type       Exec. Doc. Type       Prime       Repair       Execution       Work Center       Repair       Std.       Repair         1       Image: Click this pushbutton to rearrange the existing rows in the ascending order of the sequence number.       On-Wing       Image: Click this pushbutton to retrieve child tasks       Image: Click this pushbutton to retrieve child tasks				<b>.</b>	人民首义也		All		Ŧ			Q
Re-Number Click this pushbutton to retrieve child tasks	Rep. Seq. # Prev. Rep. Seq. # Task # P Click thi Click thi rearrange	s pushbutton to ge the existing rows	On-Wing V On-Wing V		Prime Repair	Execution	Work Center		~	•	ir	* * *
Maintain Repair Scheme	Re-Number		retrieve chilo	tasks								>

Figure 2.31 Maintaining task relationship

- 2. Select the **Effectivity Control** as "None" or "Specific" to specify whether the repair scheme is applicable for all the operators, parts or aircraft models or a specific operator, part or aircraft model. If this field is set as "Specific", you can do the following:
- 3. Specify the **Operator Code** identifying the airline operator,

Or

- 4. In the **Maint. Object #** field, enter the "Aircraft Model #", if the Applicability of the task in the header is "Aircraft", or "Part #" if the Applicability of the task is other than "Aircraft".
- 5. Use the Effective for Alternates drop-down list box to specify whether the repair scheme is

applicable for parent part alternates. The system lists the values 'yes' and 'No'.

- Note: This field is enabled only if the 'Effectivity Control' field is chosen as 'Specific'.
- 6. Specify the Additional Repair Details and click the Get Details pushbutton.

#### In the Repair Scheme Details multiline,

- 7. Enter the Task Control # indicating the master sequence number for the execution of the tasks and the Related Task #.
- 8. Use the **Effective for Alternates** drop-down list box to specify whether the repair scheme is applicable for parent part alternates. The system lists the values 'yes' and 'No'. The system displays the value 'No', by default.
- 9. Enter the Child Position # and the Child Part #.
- 10. Enter the **Child Mfr. Part #** (number assigned to the child part, by the manufacturer of the part) and the **Mfr. #** indicating the manufacturer of the part.
  - Note: The system displays the Child Mfr. Part # and Mfr. #fields for you to enter, if the parameter "Enable Manufacturer Part # control in transaction" for the Category 'Manufacturer Part' is set as "Yes" in the "Set Inventory Process Parameters" activity of the "Logistics Common Master" business component.
- 11. Use the **Prime Part Group Appl.?** to specify whether the repair scheme is applicable for the parts having the same prime part number or not. The system lists the values "Yes" and No".
- 12. Select the **Maint. Type** as "Overhaul", "Repair", "Inspection", "Other" or "Retire" to specify the type of maintenance work to be carried out on the part.
- 13. Select the **Execution Facility** to specify the location where the part can be repaired, as "In-House", "Out-Source" or "In-House & Out-Source".
- 14. Select the **Work Center** in which the task must be executed and enter the **Repair Agency #** to which the part must be sent for repair.
- 15. Use the **Std. Disposition** drop-down list to specify the type of work performed on the part. The system list the values "Repair", "Replace", "Exchange" and "Ext. Repair".
- 16. Select the **Repair Classification** in order to differentiate the tasks which are over and above the contract (COA Contract Over and Above) between the operator and the MRO.
- 17. Select the **Routing Req.?** field as "Yes" or "No" to specify whether shop work order must be created for the execution of the tasks.
- 18. If the above field is set as "Yes", you must select the option "Automatic" or "Manual" in the **Routing Sheet Updates** drop-down list, to specify whether the shop work order is created automatically or manually.
- 19. Enter the Rel. Task Lag (Days) indicating the time lag (in days) in completing the task.
- 20. Use the **Schedule Relationship** drop-down list box and select the value "Start-Start", "Start-Finish", "Finish-Start" or "Finish-Finish" to specify the sequence in which the related tasks must be executed.
  - Note: For example, if the relationship between the task T1 and T2 is "Finish– Start", this indicates that T2 can commence only if the primary task T1 has been completed.
- 21. Enter the Base Model # of the aircraft on which the task is performed.
- 22. Click the Get Related Tasks pushbutton to retrieve the following:



- If no task is selected in the multiline, the system retrieves all the child related tasks.
- If a task is selected in the multiline, the system retrieves the child related tasks specific to the selected task, with Execution relationship defined as "Concurrent Group", "Concurrent Specific" and "Concurrent Conditional".
  - Note: The "Predecessor Constrained" tasks will not be retrieved in the multiline. The count of the child related tasks is displayed as "Unprocessed Rel. Task".
- 23. Click the Maintain Repair Scheme pushbutton to maintain repair scheme for the tasks.
  - Note: The system generates Task Control #, on addition of new tasks. The system creates a repair scheme for the tasks, if the task relationship with the primary task is already defined in the "Maintain Task Relationship" page. If the relationship does not exist with the primary task, the system creates "Concurrent – Specific" relationship between the primary task and the related task for creating the repair scheme.

To proceed,

τ Select the **Maintain Task Effectivity** link to record/modify effectivity definition for a task.

## 2.1.39 Managing task effectivity

The activity allows you to perform bulk update of task effectivity. The effectivity of the task can be updated against the maintenance objects such as aircraft registration number, part number and aircraft model number. You can change the effectivity of the task against the maintenance object and update the modified effectivity details.

You can also update the effectivity changes to the maintenance program and the maintenance program template.

1. Select the Manage Task Effectivity link under the Maintenance Task business component. The Manage Task Effectivity page appears. See *Figure 2.31*.

	ject	Details										
				Maintenance Item #	* Task #	▼ 2-00-B7-05	Get Details					
-) De	fault	Details		Default Effectivity	Select				Comme	-		
Eff	ectiv	rity Details —		Default Effectivity	Select *				Comme	115		
(1)	Γ	1 - 5 / 5 🕠		-0*001	r T,					al 💷	T	
ŧ		Task # 🔎	Rev #	Model # 🔎	Aircraft Reg # 🔎	Part # 🔎	Maint. Operator # 🔎	Effectivity Change		Current Effectivity	Comments	
		2-00-87-05	1	A320-211	C-FDQV			Select	*	Restricted-Aircraft		
		2-00-B7-05	1	B767-200				Select	~	Restricted Model		
		2-00-B7-05		CSNA 172				Select	*	Restricted Model		
	E1	2-00-B7-05						Select	~	All-Maint.Operator		
			Enter	the task num	ber			Select	*			
			Enter	the task num	nber			Select Select				
		<	Enter	the task num	nber 🦷				*			
		< Options	Enter					Select	*			
			Enter	the task num					*	ie? No 🔻		

Figure 2.32 Managing task effectivity

In the Object Details group box, use the Maintenance Item # drop-down list box and select the type of the maintenance item as "Aircraft Reg #", "Part # / Mfr. Part #", "Model #", "Task #", "Maint. Operator #" and Seq. Control. #. Enter the maintenance item # corresponding to the type of the maintenance item selected, in the box provided alongside.

- 3. Click the Get Details pushbutton to retrieve the effectivity details in the multiline.
- 4. In the **Default Details** group box, select the **Default Effectivity** as "Effective" or Not Effective" and enter any **Comments** related to the task effectivity.
- 5. In the Effectivity Details multiline, enter the Task #, Rev #, Model #, Aircraft Reg #.
- 6. Enter the **Part #** for which the task effectivity details need to be updated.

Or

- 7. Enter the Mfr. Part # and the Mfr. # of the part.
  - Note: The system displays the Mfr. Part # and Mfr. #fields for you to enter, and hides the "Part #" field above, if the parameter "Enable Manufacturer Part # control in transaction" under the Category 'Manufacturer Part #' is set as "Yes" in the "Set Inventory Process Parameters" activity of the "Logistics Common Master" business component.
- 8. Select "Effective" or "Not Effective" in the **Effectivity Change** drop-down list box, to change the effectivity of the task as effective or nor effective.

The system displays the **Current Effectivity** of the task as "Effective", "Not Effective", "Restricted" or "All".

- 9. Enter any **Comments** related to the effectivity and the **Base Model #** of the aircraft.
- 10. In the **Update Options** group box, select "Yes" in the **Update Maint. Program?** drop-down list box if you wish to update the maintenance program. Else select "No".
- 11. Select "Yes" in the **"Update Maint. Program Template?** drop-down list box if you wish to update the maintenance program template.
- 12. Click the Update Effectivity pushbutton to update the task effectivity details.

To proceed,

- $\tau$  Select the **Edit Task Details** link to modify details of the task.
- τ Select the Manage Task File Attachment link to associate multiple files to a task for printing.
- $\tau$  Select the **Record Task Part Mod Details** link at the bottom of the page to update part modification details for the task.

#### 2.1.40 Managing task file attachment

You can associate multiple files to a task for printing. For an aircraft model # and for a range of manufacturer serial number of aircraft specified, you can enter the file mapping details or retrieve the already existing task-file mapping details against the Task #-Revision # combination. You can specify the sequence in which the files must be printed.

File attachment may be broadly classified into two types as follows:

*Specific File Attachment*: The file attachment in which From/To Aircraft MSN, Aircraft Ownership, Owning Agency and Maint. Operator # are defined.

*Generic File Attachment*: The file attachment in which From/To Aircraft MSN, Aircraft Ownership, Owning Agency and Maint. Operator # are not defined.

Whenever Package Print / Selective Print / Task Card Print is initiated, if specific file attachment(s) is printed for a Task, then the generic file attachment(s) are also



printed along with it, if the process parameter 'Print List Selection for Task Card Files with Specific Attributes' under the Entity 'Task' of the Entity Type 'Maintenance Task' is set as "Matched & Generic Files" in the "Common Master" business component. If this process parameter is set as "Matched Files Only", specific file attachment(s) overrides generic file attachment(s). (i.e. only specific file attachment(s) is printed for a Task).

1. Select the Manage Task File Attachment link under the Maintenance Task business component. The Manage Task File Attachment page appears. See *Figure 2.32*.

							Da	e Format yyyy-o		■ ☆ ←	? [
		Criteria Task Details		•	Ado	l. Search On Search	V		Task Status A	ctive 💌	
)-Sei	_	Results 1 - 10 / 59 → →→ +						# # II	Al	T	
_		Task # $\wp$	Revision #	Aircraft Model # 🔎	From Aircraft	MSN	To Aircraft MSN	Aircraft Owners		Owning Agency # J	ρ
		3-A31-00-MPD-08952	1	A310				Owned	*		
		3-A31-00-MPD-08952	1	A310				Owned	*		
		3-A31-00-MPD-08952	1	A310				Leased	*		
		3-A31-00-MPD-08952	1	A310					*		
	6	T-1	1						*		
		7-1	1						*		
		T123-V							*		
		T123-V Enter th	ne task num	ber					*		
		1-50C-0							*		
0		1-50C-0000-9X-00005092	1	A550					*		
		<									
		Check File Availability									
						Save					
						5075					
		View	File		View Task			Upload Docume	nte		

#### Figure 2.33 Managing task file attachment

- 2. Specify the Search Criteria and click the Search pushbutton..
- 3. Enter a valid Task # to which the file is to be mapped.
- 4. Enter the Seq # in which the files must be printed.
- 5. Enter the File Reference #.
- 6. Enter either Task Reference # or File Name.
- 7. Click the **Check File Availability** pushbutton to determine the availability of file in the specified folder in the FTP path.
- 8. Click the Save pushbutton to save the task-file mapping details.

To proceed further,

- $\tau$  ~ Select the View File link to view the file associated to the task.
- $\tau$  Select the **View Task** link to view the task details.
- $\tau$  Select the **Upload Documents** link to upload the files.

#### 2.1.41 Authoring repair procedure

- 1. Select Author Repair Procedure link from any of the following pages. The Author Repair Procedure page appears. See *Figure 2.33*.
  - Identify Resolution Procedure page of Discrepancy Processing business component.
  - Plan Work Order page or Record Shop Execution Details page of Shop Work Order business component.
  - Record Aircraft Maintenance Execution Details page of Aircraft Maintenance Execution business component.



- Review Fleet Maintenance Plan page of Aircraft Maintenance Planning component.
- Manage Work Assignments and Reporting page of Work Monitoring and Control component.
  - Note: You cannot launch this page, if the following are true: (a) the status of the Reference Document is "Terminated", (b) the task is a standard task or (c) from the "Identify Resolution Procedure" page, if the Reference Document is "Maintenance Report" and the processing status "Analyzed".

Enter or modify the following fields in the **Task Details** group box:

- 2. Enter the Task Description and the Task Category.
- 3. Use the **Exec. Phase** drop-down list box to select the execution phase of the task. The system provides the following options:
  - "Preparatory" Select this option, if the task must be performed as a part of preliminary activities before the aircraft is grounded for maintenance.
  - "Regular" Select this option, if the task must be performed as a part of normal maintenance activities carried out in the hangar.
  - "Post Flight" Select this option, if the task must be performed as a part of post test flight activities.
  - In addition to the above, the drop-own list box displays all quick codes defined under the quick code type "Exe. Phase", which are in Active status.
- 4. Enter the Estimated Man Hours in the Task Details group box.
  - Note: You can modify the details of a sub-task, only if the job status of the task is 'Fresh' and the sign-off' status of the sub-task is "Pending".
- 5. Enter the Seq. # in which the sub task must be performed while executing the maintenance task.

Author Repair Procedure							4	? 🗔
Reference Details								
Ref. Doc #		Ref. Doc Type						
Associated Discrepancy#	Discrepancy De	scription						
Part #		Part # / Serial #			Component #			
- Task Details								
Task # VP-000546-	-2014/1	Previous Task Cou	nt 0(0) Next					
Task Description sdfgfsdg		Long Description	dfgfsdg					
Task Type 🛛 🔻		Task Type	-		Repair Classification	1	Ŧ	
Exec. Phase Regular	T	Est. Elapsed Time	1.00 Hours 🔻		Est. Man Hrs.	1.00		
Work Center #		ATA #						
Sub-Task Details								
				🗄 🔤 🖷	AI	•		Q
# 🗏 Şeq # Sub Tasks	Resource Group	Skill # 🔎	Document Id		File Name 🔎			
1 🗉 1 sdfgfsdg	Not Required 🗸	01						
2	Not Required 🗸	00						
3	Not Required							
Enter the sequence in which	Required 🗸 🗸							
the sub task must be executed	Required 🗸							
the sub task must be executed	<b>^</b>							
<								>
- Additional Details								
Notes								
Document Attachment Details								
File Name Ø		View File						
File Name 👂								
		Author Repair	Procedure					

Figure 2.34 Author Repair Procedure

- 6. Use the **Resource Group** drop-down list box to select the resource group for performing the sign-off of the sub task. The system provides the following options:
  - "Not Required" Select this option if sign-off is not required.
  - "Mechanic" Select this option if the sign-off is to be performed by the mechanic.
  - "Inspector" Select this option if the sign-off is to be performed by the inspector.
  - "Insp. & Mech." Select this option if the sign-off is to be performed by both the inspector and mechanic.
    - Note: Addition of new subtasks for a non-standard task is allowed till the last subtask associated to the non-standard task is signed-off.
- 7. Click the Author Repair Procedure pushbutton.
  - Note: Ensure that (a) the Reference Document is in "Active" status for the aircraft registration number, (b) the status of the Reference Document Number is not "Hold" and (c) the status of the task is "Fresh" or "Scheduled".

## 2.1.42 Maintaining Standard Cost of task

- 1. Select the Maintain Task Standard Cost link under the Maintenance Task business component. The Maintain Standard Cost page appears.
- 2. In the Task Details multiline, enter Task # for which you want to record standard cost details.
- 3. Click the Get Task Details pushbutton to retrieve details of the specified task.
- 4. In the **Task Details** multiline, enter **Material Cost, Resources Cost** and **Other Cost** incurred on execution of the task.

- 5. Enter Total Cost required for completing the task (Total Cost = Material Cost + Resources Cost + Other Cost).
- 6. Enter **Effective From** and **Effective To** dates to specify the period during which the cost remains valid for the task.
- 7. Click the Save Std. Cost pushbutton to save the recorded cost details.

## **3COMPONENT MAINTENANCE PROGRAM**

The component maintenance program contains all the scheduled maintenance related details pertaining to the component. It has the work units that need to be performed on the component, along with the schedules for execution of the work unit. It can either be created directly at the component level, or copied from the part program in bulk at the part level. In either case, the program details like work units and their schedules can be modified.

The component maintenance planning cell for component assesses all the upcoming maintenance activities that need to be executed for the component sets, which are due for maintenance.

The scheduled maintenance of components is driven through pre-defined maintenance programs. With the arising of these scheduled maintenance events, the component maintenance planner forecasts for a planning horizon and propagates execution planning. The planner finalizes the work scope; allocates the component removals and on-wing inspections to the visit package and creates work scope along with the work order.

The **Component Maintenance Program** business component enables you to group all the scheduled maintenance activities to be performed on the component and provides a single lookup for all the maintenance-related information for a component.

## **3.1 Configuring component maintenance program**

## 3.1.1 Defining the quick codes

#### What are quick codes?

Quick Codes are user-defined values, used to categorize a set of details of identified behavior. These quick codes are later used in the process of retrieving or addressing the details by referring to the attached quick code.

Quick codes act as additional qualifiers for a business entity or document. Quick codes can assume user provided values which can be used to categorize/group an entity/document record to satisfy specific needs in a user organization's internal processes, especially with respect to unique reporting requirements.

The "User-Status" quick code type is predefined by the system. You can define values for this quick code type. For example, the "User-Status" quick code type can contain the "Material Not Available" quick code.

1. Select Add Quick Codes in the Component Maintenance Program business component. The Create Quick Codes page appears. *See Figure 3.1* 

*		Create Quick Codes					<u></u>		+ 2	Ø K
-	)uick (	Qu ode Details	ick Code Type Program Group							
	•	1 -1/1 > >> + - O	4 0 0 T T		▶ ๒ ◙ ४ छ 首 ∞	# # III A	1	T		Q
#		Quick Code	Description							
1		CMP	Component maintenance program							
2										
										_
										_
_										
_			Crea	te Quick Codes						
_										

Figure 3.1 Creating quick codes

- 2. Use the **Quick Code Type** drop-down list box to select the type of quick code to be created. You can define quick codes of the type "User Status" and "Program Group".
- 3. Enter the **Description** for the quick code.
- 4. Click the Create Quick Codes pushbutton.
  - Note: The system assigns the "Active" status to the quick codes entered in the multiline.

#### 3.1.2 Setting options

You can define system parameters in order to update the schedules of all the tasks in a work order. Based on the option settings, the system updates the last performed dates for the tasks in the work order.

- 1. Select the **Set Options** link under the **Component Maintenance Program** business component. The **Set Options** page appears.
- 2. In the **Option Setting Details** group box, use the **Compliance Updation Basis** drop-down list box to select either "Actuals" or "Time Window", for executing the work unit.
  - Actuals Select this option if the next due date for carrying out the work units must be calculated based on its previous completion.



- Time Window Select this option if the following conditions are applicable:
  - Within the Tolerance limits: If the option is selected as "Time Window", the NSD (Next Schedule Date /NSV (Next Schedule will be calculated based on a scheduled fixed interval)
  - Beyond the Tolerance limits: If the option is selected as "Time Window", the NSD / NSV will be calculated by adding the Repeat Interval value to the date or value at which the work unit is actually executed.
- 3. Use the **Planner Group Security d**rop-down list box to specify whether planner group security is required or not.
- 4. Use the **Impact on Sch. Adj.** drop- down list box to stipulate whether schedule adjustments must influence existing schedules for the scheduled adjusted task in component maintenance program. Select:
  - "Retain Existing Schedules", to retain the existing schedules for the scheduled adjusted tasks in component maintenance program.
  - "Reset to Template Schedules", to reset the existing schedules for the schedule adjusted tasks in the component program based on the part program.
- 5. Use **Rules based Inheritance of As Required tasks to Execution documents** the drop-down list to indicate whether the "As Required" tasks must be inducted into the execution document based on rules.
- 6. Use the Alert Value based inheritance of Scheduled tasks into Execution Documents drop-down list box to indicate whether the scheduled tasks must be inducted into the execution document based on the alert value.
- 7. Click the **Set Options** pushbutton to set the values.

## 3.1.3 Creating part maintenance program

#### **Creating part program**

You can create a maintenance program based on parts.

- 1. Select Maintain Part Program under the Component Maintenance Program business component. The Select Part # page appears.
- 2. Enter the **Part #** and click the **Maintain Program** link alongside. Or, search for a part, click the **Search** pushbutton and select the hyperlinked **Part #** in the **Part Details** multiline. The **Edit Part Program Information** page appears. *See Figure 3.2* 
  - Note: The system retrieves only the 'Active' parts in the search results.



Figure 3.2 Creating a part maintenance program

- 3. Enter Parent Removal and Specific Part # in the Removal details group box.
- 4. Specify the **Program Group** of the maintenance program planned for the part.
- 5. Use the **Maintenance Type** drop down list box in the **Maintenance Details** multiline and select the maintenance type, which could be "Overhaul", "Inspection", "Repair", "Retire" or "Others".
  - Note: The "Retire" option will be available only if the "Lifed?" field is set as "Yes" for the part number, in the "Maintain Maintenance Info. for Part" page of the "Aircraft" business component. Only one work unit can exist with maintenance type "Retire" for the part program.
- 6. Enter the Work Unit #.
- 7. Specify the **Program Item Type** maintenance task. The drop-down list box displays the following values: Block, Base, Perpetual, As Required, Non-Block and To be Decided.
- 8. Use the **Default Exe. Priority** drop-down list box to specify the priority for execution of the task with reference to other tasks in the maintenance program planned for the part.
- 9. Use the **Initiated/Reset By** drop-down list box to specify the manner in which the Next Scheduled Value (NSV) for the task in the maintenance program must be computed. When you select,
  - Self-Compliance, the system sets the Next Schedule Value/Next Schedule Date on compliance of the task.
  - Related Task Compliance, the system sets the Next Schedule Value/Next Schedule Date on compliance of all related tasks in additional to the compliance of main task.
- 10. Use the Reset on Attachment? drop-down list box and select,
  - 'Not Required' to compute NSD / NSV based on Last Performed Date / Value.
  - 'First Attachment' to compute NSD / NSV on first attachment of the component after the off-wing compliance of the task.

ramco



- 'Every Attachment' to re-compute NSD / NSV on every attachment of the component to the Aircraft.
- 11. Specify the parent task associated with the work units/tasks planned to be executed on the part, in the **Parent Item #** field.
  - > Note: This field is applicable only if the Program Item Type of the task is "Base".
- 12. Indicate the **Schedule Status** of the work unit task, which could be "Active" or "Inactive" or Terminated.
- 13. Set the **Execution Facility** to "In-house" or "Out-source" to specify the location where the work unit has to be performed.
  - Note: The "In-house" option alone will be listed in this field, if the execution facility is set as "In-house" for the part number in the "Maintain Maintenance Info. for Part" page of the "Aircraft" business component. The "Out-source" option alone will be listed in this field, if the execution facility is set as "Out-source" for the part number in the "Maintain Maintenance Info. for Part" page of the "Aircraft" business component.
- 14. Select the Work Center # and enter the Est. Duration.
- 15. Use the **On Wing?** drop down list box to select whether the work unit has to be performed on-wing or off-wing.
- 16. Use the **Package Type** drop-down list box to select the package type of the work unit.
- 17. Enter the Ref. Document Type, Ref Document #, Reference EO # and other details.
- 18. Use the **Shelf Life Expiry** drop-down list box to select whether the given maintenance requirement is applicable to the shelf life expiry of the component.
- 19. Use the **Expense Type** drop-down list box to select the expense type of the work unit. The system provides the options "Revenue" and "Capital".
  - Note: Ensure that this field is set to "Revenue", if the expense type for the part is identified as "Revenue" in the "Part Administration" business component.
- 20. Select the Get Base Tasks pushbutton to display base tasks of a task in the multiline.
  - Note: The program item type of the task for which you want to view the base tasks must be "Block".
- 21. Use the **Change Work Unit Status To** drop-down list box to change the status for the work unit to "Active" or "Inactive".
- 22. Select the **Copy to Components** check box to automatically copy changes made in part program to component programs upon authorization of part program.
- 23. Click the **Update Program** pushbutton to create the part maintenance program.
  - Note: The system ensures that there is at least one work unit with maintenance type "Retire", if the "Lifed?" field is set as "Yes" for the part number in the "Maintain Maintenance Info. for Part" page of the "Aircraft" business component.

To proceed,

- $\tau$  Define date based schedule
- $\tau$  Define usage based schedule
- $\tau \quad \ \ \text{Associate estimated skill requirement}$
- τ Associate estimated facilities requirement

- τ Associate estimated material requirement.
- τ Copy part maintenance program to multiple parts
- τ Authorize part program
- $\tau$  Upload documents associated with the part maintenance program
- τ Maintain component maintenance programs

## 3.1.4 Defining date based schedule

1. Select the Edit Date Based Schedule link in the Edit Part Program Information page. The Edit Date Based Schedule Information page appears. *See Figure 3.3* 

*		Edit Date Based Sch	edule	Information												-		+	?	to K
- Pa	art De	tails				_				-										
			Main	Part #		4-0001:36361 dition			work			Part Des	cription ENGIN	NE						
				Program Status	Authoriz	ed	unit n	um	ber			Rev	ision # 1							
-M	ainte	nance Activity Details		Work Unit #	TSK-1-5	50C-0000-CMM-	00000053			-										
			Work	Unit Description	PME-1							Maintenanc	eType Inspec	ction						
	ata D	ased Schedule Details —		Work Center #							Work	Center Des	cription							
	_	1 -1/1 → → +	- 8		Ŧ							1 -			_		Ŧ	_	_	Q
#		Schedule Type	- 0	Time Unit		Threshold			Interval		Alert Value	Positive T		All	Nega	tive Tol	-		-	2
1		Recurring	~	Days	~			12.00		10.00					-					
2		Recurring	*	Days	*															
		<																		>
-									Edit Det	e Based Schedu	4-									
_									Edit Date	e Baseo Schedu	lie									

Figure 3.3 Defining date based schedule for part program

- 2. Use the **Work Unit #** drop down list box to select the work unit to be performed. The system displays all the work units created for the part except those for which **Shelf Life Expiry** has been set to "Yes". Click the **Get Details** pushbutton to get the work unit information.
- 3. Select the Schedule Type, which could be "Recurring" or "One-Time".
  - Note: This field should be set as "One-Time", if the *Maintenance Type* of the work unit is "Retire".
- 4. Select the Time Unit, which could be "Days", "Weeks", "Months", "Month End" or "Years".
- 5. Enter the starting value of performing the work unit in the **Threshold** field.
- 6. Enter the value after which an alert need to be given in the Alert Value field.
- 7. Enter the Positive Tolerance and Negative Tolerance.
- 8. Enter the Terminating Date and Terminating Value.
- 9. Select "Last Performed Date" or "Installation Date" in the **Update Basis** field, to specify the basis for updating the work units of the part program.
- 10. Enter any comments or supplementary information on the maintenance program planned for the part in the **Remarks** field.
- 11. Click the Edit Date Based Schedule pushbutton to record the schedule details.

#### 3.1.5 Defining usage based schedule

1. Select the Edit Usage Based Schedule link in the Edit Part Program Information page. The Edit Usage Based Schedule Information page appears. *See Figure 3.4* 



Edit Usage Based Schedul	le Information					44 4 1 2	3 ▶ ₩ 1 /	3 🗐 🕮 🖷	₽ ← '	? 🗟 🖪
- Part Details										
	Part # 0-0440-4-0	001:36361				Part Descript	tion ENGINE			
	Maintenance Process On-Condition	n								
	Program Status Fresh					Revisio	n#2			
Maintenance Activity Details	Work Unit # TSK-1-50C Work Unit Description PME-1	-0000-CMM-00000053	Get Details			Maintenance T	/pe Inspection			
	Work Center #					Work Center Descript				
- Usage Based Schedule Details	Work Center #					work center beschp	5011			
(i)	-040011				人口		Al III Al		<b>v</b>	Q
# 🖻 Parameter 👂 UOM	Schedule Type	Threshold	Interval		Alert Value	Positive Tolerance	Negative Tolerance	Terminating Value	Remarks 📝	
1 🗉 FC	Recurring	1000	0.00	5000.00	4500.00	200.00	2002.00	1000000.00		
2 E FH Enter the	parameter	Select "Recurrin or "One-Time" as schedule type								
	E		Edit Usag	e Based Sche	dule					

Figure 3.4 Defining usage based schedule for part program

- 2. Use the **Work Unit #** drop down list box to select the work unit to be performed. The system displays all the work units created for the part except those for which "Shelf Life Expiry" has been set to "Yes".
- 3. Click the **Get Details** pushbutton to get the work unit information.
- 4. Enter **Parameter**, the number identifying the consumption parameter in the aircraft.
- 5. Select the **Schedule Type**, which could be "One Time" or "Recurring".
  - Note: This field should be set as "One-Time", if the Maintenance Type of the work unit is "Retire".
- 6. Enter the value after which an alert need to be given in the **Alert Value** field.
- 7. Enter the Positive Tolerance and Negative Tolerance.
- 8. Enter the Terminating value.
  - Note: The system displays the parameter values in the multiline based on the option "Decimal Format" or "HHMM Format" selected for the parameter in "Time Display Option" drop-down list box in the "Aircraft" business component while parameter creation. For e.g. If the Time Display Option is selected as "HHMM Format" and the value stored is 2.50 in the data base, the system displays 2.30 in the multiline.
- 9. Click the Edit Usage Based Schedule pushbutton to record the schedule details.

#### 3.1.6 Copying part maintenance program to multiple parts

You can copy the maintenance program created for a part to other parts.

1. Select the **Copy Program To Part #** link in the **Edit Part Program Information** page appears. *See Figure* 3.5



Copy Program to Part #				=	ē (	+	?	¢ K		
Copy From	Part # 0-0440-4-0001:36361 ntenance Process On-Condition		Part Description ENGINE							
Mail → Associate to (( ( 1 - 1/1 ) )) + - # Part # 0		Click this pushbutton to copy program to the			V		_	Q		
1 35895 COST	test	selected part								
		Copy Program								

Figure 3.5 Copying maintenance program to other parts

- 2. Enter the number identifying the part to which the maintenance program must be copied, in the **Part #** field.
- 3. Click the Copy Program pushbutton to copy the maintenance program details.

#### 3.1.7 Authorizing a part maintenance process

- 1. Select Authorize Part Program under Component Maintenance Program business component. The Authorize Part Program page appears. *See Figure 3.6.*
- 2. Enter the desired search criteria in the **Search Criteria** group box and click the **Search** pushbutton. The system retrieves all the details of all the parts associated with the maintenance, matching the search criteria.
- 3. Select the part program to be authorized, in the multiline.

		Part # Part # P	naintenance rocess for	Date Format yyyy-dd-mm Part Description ATA # Search								
	1 - 10 / 11 🕨 🕨 🝸	T,				🖆 👎 🖮 🖬	T					
1	Part #	Part Description	Maintenance Process	ATA #	Program Revision #	Updated by	Updated Date					
	0-001-368-016:35895	REPAIRABLE CAT3 STARTER		20-00	0	DMUSER	2014-02-04					
	0-0440-4-0001:36361	ENGINE	On-Condition	72-00	2	DMUSER	2016-12-05					
	0-0440-4-0005:36361	AUX. POWER UNIT	On-Condition	49-00	3	DMUSER	2015-25-09					
	0-0440-4-0015:36361	PS9323 CARRIER	On-Condition	25-30	0	DMUSER	2014-30-09					
	0-100-1	3".DIA.0-100PSI GAUGE	On-Condition	00-00	1	DMUSER	2014-26-03					
	0-3000-25LBDIV:08664	0-3000 PSI GAUGE	On-Condition	138-70	0	DMUSER	2014-02-04					
	0-3200-0-0002:3001	PART COMP	Hard-Time	137-32	0	DMUSER	2014-03-04					
	1-00-558548	component swo	On-Condition	130-10	0	DMUSER	2016-15-04					
	1134: 1PL57	PISTOL GREASE GUN	On-Condition	00-00	0	DMUSER	2014-26-03					
	161T1100-91:36361	MAIN LANDING L/H GEAR	On-Condition	32-10	0	DMUSER	2015-25-09					
	<b>De</b>		Art #         Part # </th <th>Part # Part # Process for the part  Process for the part Process for the</th> <th>A Criteria         Part #         maintenance process for the part         Search           1         10/11         11         17         7         Search           Part #         Part Description         Maintenance Process         ATA #           0         001-584-016:5895         REPAIRABLE CAT3 STARTER         20-00           0         0-0440-4005:58361         PKINE         On-Condition         72-00           0         0-0440-4005:58361         PKINE         On-Condition         25-30           0         0-0440-4005:58361         PKINE         On-Condition         25-30           0         0-0440-4005:58361         PS9323 CARRIER         On-Condition         00-00           0         0-0400-4005:58361         PS9323 CARRIER         On-Condition         00-00           0         0-3000-258.001/08664         0-3000 PSI GAUGE         On-Condition         138-70           1         10-3000-258.901/08664         0-3000 PSI GAUGE         On-Condition         139-70           1         10-05954         component stwo         On-Condition         130-10           1         103-15954%         component stwo         On-Condition         100-00           1         134:19.757         PISTU. GREASE GUN<th>A Criteria         maintenance process for the part         Search         Part #         P</th><th>A criteria       maintenance process for the part       Search         Part #       Part Description       ATA #         Part #       Part Description       Maintenance Process         0001:36301       ENGINE       On-Condition       20-00       0         00400-4001:36361       ENGINE       On-Condition       72-00       2       MulsER         00400-4001:36361       ENGINE       On-Condition       72-00       2       MulsER         0-0400-4001:36361       ENGINE       On-Condition       72-00       2       MulsER         0-0400-4001:36361       ENGINE       On-Condition       25-30       0       MulsER         0-0300-25LBCIV:08664       0-3000 PSI GAUGE       On-Condition       0:38-70       0       MulsER         0-0300-002:301       PART COMP       Hard-Time       137-</th></th>	Part # Part # Process for the part  Process for the part Process for the	A Criteria         Part #         maintenance process for the part         Search           1         10/11         11         17         7         Search           Part #         Part Description         Maintenance Process         ATA #           0         001-584-016:5895         REPAIRABLE CAT3 STARTER         20-00           0         0-0440-4005:58361         PKINE         On-Condition         72-00           0         0-0440-4005:58361         PKINE         On-Condition         25-30           0         0-0440-4005:58361         PKINE         On-Condition         25-30           0         0-0440-4005:58361         PS9323 CARRIER         On-Condition         00-00           0         0-0400-4005:58361         PS9323 CARRIER         On-Condition         00-00           0         0-3000-258.001/08664         0-3000 PSI GAUGE         On-Condition         138-70           1         10-3000-258.901/08664         0-3000 PSI GAUGE         On-Condition         139-70           1         10-05954         component stwo         On-Condition         130-10           1         103-15954%         component stwo         On-Condition         100-00           1         134:19.757         PISTU. GREASE GUN <th>A Criteria         maintenance process for the part         Search         Part #         P</th> <th>A criteria       maintenance process for the part       Search         Part #       Part Description       ATA #         Part #       Part Description       Maintenance Process         0001:36301       ENGINE       On-Condition       20-00       0         00400-4001:36361       ENGINE       On-Condition       72-00       2       MulsER         00400-4001:36361       ENGINE       On-Condition       72-00       2       MulsER         0-0400-4001:36361       ENGINE       On-Condition       72-00       2       MulsER         0-0400-4001:36361       ENGINE       On-Condition       25-30       0       MulsER         0-0300-25LBCIV:08664       0-3000 PSI GAUGE       On-Condition       0:38-70       0       MulsER         0-0300-002:301       PART COMP       Hard-Time       137-</th>	A Criteria         maintenance process for the part         Search         Part #         P	A criteria       maintenance process for the part       Search         Part #       Part Description       ATA #         Part #       Part Description       Maintenance Process         0001:36301       ENGINE       On-Condition       20-00       0         00400-4001:36361       ENGINE       On-Condition       72-00       2       MulsER         00400-4001:36361       ENGINE       On-Condition       72-00       2       MulsER         0-0400-4001:36361       ENGINE       On-Condition       72-00       2       MulsER         0-0400-4001:36361       ENGINE       On-Condition       25-30       0       MulsER         0-0300-25LBCIV:08664       0-3000 PSI GAUGE       On-Condition       0:38-70       0       MulsER         0-0300-002:301       PART COMP       Hard-Time       137-					

4. You can also use the **Select All** box if you wish to authorize all the part programs at the same time.

Figure 3.6 Authorizing a part maintenance program

- 5. Click the **Authorize Program** pushbutton to authorize the maintenance program of the selected parts.
  - Note: The status of the part maintenance program is set to "Authorized" and is available for forecasting. The system also updates this maintenance program for all parts of the same type with different serial number.

#### 3.1.8 Creating component maintenance program

You can create a maintenance program based on component. A component is a combination of a part number and its serial number.

You can create a maintenance program based on parts.

- 1. Select the **Maintain Component Maintenance Program** under the Component Maintenance Program business component. The Select Component # page appears.
- 2. Enter the **Component #** and click the **Maintain CMP** link provided alongside. Or, search for a part, click the **Search** pushbutton and select the hyperlinked component number in the multiline. The **Edit Component Maintenance Program Information** page appears. *See Figure 3.7.*
- 3. Select the type of parent removal in the **Parent Removal** drop-down list box and enter the **Specific Part #** fields in the **Part Removal details** group box.

* 8	Edit	Component Main	tenance	Program Informatio	on										4 1 2 3 4 5	5 🕨 🛪 🖨	🛱 🗲 ? 🗔
													Date Format	yyyy/mm/da	d hh:mm:ss		
- Comp		Details		Part # Serial # Maintenance Process Attached to : Aircraft Reg # Replacement Type CMP Status Revision #	A74700 On-Cor SRU Author	ndition					Airline operator mapped to component		Component #	000014 131-10 00-00	ANGLE DRILL		
+ Part I				Remarks									` <u> </u>	<		a user-de to the prog	
			-0	× • # T 7								人血目		;∓∌00	1. 14 % All	▼ Search	Q
#		Program Group		Maintenance Type		Work Unit #		Q	Prog. Item Type		Default Exe. Priority		Initiated/ Reset by		Reset on Attachment	2	
1			~	Inspection	~	1-50C-0000-9X-000	05057		Non-Block	~		~	Self Compliance		V Not Required		
2				Repair		1-50C-0000-CMM-0	0006120		Non-Block	~			Self Compliance		<ul> <li>Not Required</li> </ul>		
3				Others		TASK3			Non-Block	~	AOG		Self Compliance		<ul> <li>Not Required</li> </ul>		
4				Others		TASK1							Self Compliance		<ul> <li>Not Required</li> </ul>		
2			~	Others	Ň	TASK2			Non-Block	Sp Sp	ecify the	~	Self Compliance		<ul> <li>Not Required</li> </ul>		•
Get Bas	e Task										be of parent			1			,
Edit Date Maintain T							Edit Usage Ba <u>nitial</u> ize Main			rer	noval		Airling anor				
+ Actio		· ·	0										Airline opera				
T ACtio	n		Clic	ck here to re	ecor	d							mapped to t	ask			
			opr	portunity che	eck	details								u.u.u		1	
		Check Details Noc. Attachments	7			dotano	rize Con	mponent I	laintenance Program				Upload Documents				
			7														
					Fig	gure 3.7	7 Cr	eat	ing a co	тро	nent maintei	iai	Select 'Rese "Not Require				

- 4. Specify the **Program Group** of the maintenance program
- 5. Use the **Maintenance Type** drop down list box in the **Maintenance Details** multiline to select the type of maintenance type which could be "Overhaul", "Inspection", "Repair", "Retire" or "Others".
  - Note: The "Retire" option will be available only if the "Lifed?" field is set as "Yes" for the part number, in the "Maintain Maintenance Info. for Part" page of the "Aircraft" business component. Note: Only one work unit can exist with maintenance type "Retire" for the part program.

or "Every Attachment

- 6. Enter the Work Unit #.
- 7. Specify the **Program Item Type** maintenance task. The drop-down list box displays the following values: Block, Base, Perpetual, As Required, Non-Block and To be Decided.
- 8. Use the **Default Exe. Priority** drop-down list box to specify the priority for execution of the task with reference to other tasks in the maintenance program planned for the part.
- 9. Use the **Initiated/Reset By** drop-down list box to specify the manner in which the Next Scheduled Value (NSV) for the task in the maintenance program must be computed. When you select,
  - Self-Compliance, the system sets the Next Schedule Value/Next Schedule Date on compliance of the task.
  - Related Task Compliance, the system sets the Next Schedule Value/Next Schedule Date on compliance of all related tasks in additional to the compliance of main task.
- 10. Use the Reset on Attachment? drop-down list box and select,
  - 'Not Required' to compute NSD / NSV based on Last Performed Date / Value.

- 'First Attachment' to compute NSD / NSV on first attachment of the component after the off-wing compliance of the task.
- 'Every Attachment' to re-compute NSD / NSV on every attachment of the component to the Aircraft.
- 11. Specify the parent task associated with the work units/tasks planned to be executed on the part, in the **Parent Item** # field.

> Note: This field is applicable only if the Program Item Type of the task is "Base".

- 12. Indicate the **Schedule Status** of the work unit task, which could be "Active" or "Inactive" or Terminated.
- 13. Set the **Execution Facility** to "In-house" or "Out-source" to specify the location where the work unit has to be performed.
  - Note: The "In-house" option alone will be listed in this field, if the execution facility is set as "In-house" for the part number in the "Maintain Maintenance Info. for Part" page of the "Aircraft" business component. The "Out-source" option alone will be listed in this field, if the execution facility is set as "Out-source" for the part number in the "Maintain Maintenance Info. for Part" page of the "Aircraft" business component.
- 14. Select the Work Center # and enter the Est. Duration.
- 15. Select "Yes" in the **On Wing?** drop down list box to specify that the work unit must be performed on-wing. Select "No" if the work unit must be performed off-wing.
- 16. Select the Package Type and enter the Ref Document Type, Ref Document #, Reference EO # and other details.
- 17. Use the **Shelf Life Expiry** drop-down list box to specify whether the given maintenance requirement is for shelf life expiry of the component.
  - Note: If Shelf Life expiry is specified as "Yes", you cannot define schedules (date based as well as usage based) for the relevant work unit.
- 18. Use the **Expense Type** drop-down list box to specify the expense type of the work unit. The system provides the options "Revenue" and "Capital".

  - a. If the expense type for the part is identified as "Revenue" in the "Part Administration" business component.
  - *b.* If the component ownership is other than "Owned" in the "Aircraft" business component.
- 19. Select the Get Base Tasks pushbutton to display base tasks of a task in the multiline.
  - Note: The program item type of the task for which you want to view the base tasks must be "Block".
- 20. Use the **Change Work Unit Status To** drop-down list box to change the status for the work unit to "Active" or "Inactive".
- 21. Click the Update CMP pushbutton to create the component maintenance program.
  - Note: The system ensures that there is at least one work unit with maintenance type "Retire", if the "Lifed?" field is set as "Yes" for the part number, in the "Maintain Maintenance Info. for Part" page of the "Aircraft" business component.
  - Note: On creation of component maintenance program, the system copies "Revenue" as the expense type for the work units in the component maintenance program, irrespective of the expense type identified for the work units in the part



program, if the ownership is identified as "Customer" for the component number. Else, the system copies the expense type identified for the work units in the part program.

22. Click <u>the **Cancel CMP**</u> pushbutton to cancel the specific revision of the component maintenance program.

#### To proceed,

- $\tau \qquad \text{Define usage based schedule} \\$
- τ Define date based schedule
- τ Associate estimated skill requirement
- τ Associate estimated facilities requirement
- τ Associate estimated material requirement
- τ Associate work units on opportunity check
- τ Authorize part maintenance program
- τ Set Execution Inheritance rules for tasks in the maintenance program
- $\tau$  Upload documents associated with the component maintenance program

## 3.1.9 Associate work units on opportunity check

You can carry out some additional work units along with the next maintenance arising of the component.

- 1. Select the Edit Opportunity Check Details link in the Edit Component Maintenance Program Information page. The Edit Opportunity Check Information page appears. *See Figure 3.8.*
- Use the Execute With drop down list box to indicate when the work unit is to be performed on the component. The system lists the options "Next Overhaul", "Next Repair", "Next Inspection" and "Next Shop-Visit".
- 3. Enter the Work Unit Type, which could be "Task" or "Standard Procedure".

4. Enter the Work unit # which is to be performed. Edit Opportunity Check Information Date Format yyyy-dd-mm - Component Details Part # FRH280002:U1918 Part Description ATR RELEASE VALVE Serial # FRTR293-AD Component # 000019 ance Process On-Cond CMP Status Fresh Revision # 0 Work Unit Details (4 4 1 - 3 / 3 → → + - □ % Ø Ø T T<sub>x</sub> Work Unit Des Execute With Work Unit # 0 Everation Lin 1-00-90 Barcoding Check - 2 1 Next Inspection ✓ 1-00-91 2 3 Select the 1-00-92 Enter the work work unit type unit number Edit Opportunity Checks

Figure 3.8 Recording of opportunity check details

5. Click the Edit Opportunity Checks pushbutton to record the details.

#### 3.1.10 Authorizing a component maintenance program

1. Select Authorize Component Maintenance Program under Component Maintenance Program business component. The **Authorize Component Maintenance Program** page appears.

- 2. Enter **Search Criteria** and click the **Search** pushbutton.
  - Note: The system retrieves all the components, which have maintenance programs in the "Fresh" status. The system will not retrieve the components that are attached to the aircraft with record status "Frozen".
- 3. Select the component for which the maintenance program must be authorized, in the multiline.
- 4. To authorize all the programs, check the Select All box below the multiline.
- 5. Click the Authorize CMP pushbutton, to authorize the component maintenance program.
  - Note: The system changes the status of the program to "Authorized" and the program is made available for forecasting.
- 6. Click <u>the</u> **Cancel CMP** pushbutton to cancel the specific revision of the component maintenance program.

## 3.1.11 Creating position based schedule

The maintenance schedule information can be defined for a position code and part number. Schedule values (in the CMP and in the position schedule) will be taken for the forecast when the particular part number is attached to the specific position code.

- 1. Select Maintain Position Based Schedule under Component Maintenance Program business component. The Select Position Part # page appears.
- 2. Enter the Search Criteria to search for a position code and part number combination.
- 3. Select the position code part number combination, for which schedule must be defined, in the multiline.
- 4. Enter the Aircraft Model #, Part #, Level Code and Position Code in the Position Details group box to maintain position-based schedule for alternate parts of type "Component".

To proceed,

- $\tau$  Create date based schedule
- $\tau$  Create usage based schedule
- $\tau$  Define aircraft effectivity

## 3.1.12 Creating date based schedule

- 1. Select the Edit Date Based Schedules link in the Select Position Part # page. The Edit Date Based Schedule Information page appears. *See Figure 3.9.*
- 2. Use the **Work Unit #** drop down list box to select the work unit to be performed. The system displays all the work units defined in the component maintenance program for the part except those with Program Item Type-Perpetual and As Required; prefixes it with "TSK", "SP" or "WP" based on the type of the work unit.
- 3. Click the **Get Details** pushbutton to get the work unit information.
- 4. Select the **Schedule Type, which** could be "Recurring" or "One Time".
- 5. Select the Time Unit, which could be "Days", "Weeks", "Months", "Month End" or "Years".
- 6. Enter the starting value of performing the work unit in the **Threshold** field.
- 7. Enter the value after which an alert need to be given in the **Alert Value** field.



*		Edit Date Based Sch	edule I	nformation						III 7.	: 8		⊢'	? 🗟
- Pa	art De	tails												
				Part # 0-044	0-4-0001:36361				Part Description ENGINE					
			Maint	enance Process On-Co	ndition									
				Program Status Autho	rized				Revision # 1					
- M	ainte	nance Activity Details –		Work Unit # TSK-	1-50C-0000-CMM-000	000053	Get Details							
			Work	Unit Description PME-1					Maintenance Type Inspection					
				Work Center #				Work	Center Description					
- D	ate Ba	ased Schedule Details												
44	4	1 -1/1 🕨 🗰 🕂	- 0	* © © T T.					🖹 🎞 😫 📮 🖷 💷 🗛			v		
#		Schedule Type		Time Unit	Threshold		Interval	Alert Value	Positive Tolerance	Neg	ative Tol	erance		
1		Recurring	~	Days 🗸		12.00	10.00							
2		Recurring												
				Select the s										
				type as "Or	e-Time"									
				or "Recurrin										
		<		or Recurri	iy									>
_														
						_	Edit Date Based Sched	ule						

Figure 3.9 Creating date based schedule

- 8. Enter the Positive Tolerance and Negative Tolerance.
- 9. Click the Edit Date Based Schedule pushbutton to record the schedule details.

## 3.1.13 Create usage based schedule

You can create schedules depending on the usage of the component, based on the consumption parameters of the component

1. Select the Edit Usage Based Schedules link in the Select Position - Part # page. The Edit Usage Based Schedule Information page appears. *See Figure 3.10* 

	Edit	Usage Based So	chedule Info	ormation				44 4 16 17 18 19 20	▶ ₩ 18 /500	=	4	<b>+</b> 3			
-P	ositio	n Details													
				Aircraft Model # 100-00				Configuration Class ABC							
				Part # 0-0440	-4-0001:36361			Position Code 001							
- M	ainte	nance Activity Det	ails												
				Work Unit # TSK-1	-50C-0000-CMM-	00000053 🔹 Get Details			Get Part Progra	am					
			Work	Unit Description PME-1				Maintenance Type Inspection							
				Work Center #				Work Center Description							
	sage	Based Schedule De	tails												
44	4	1 - 1 / 1 🕨 🕨	+ - 0	* © © T T,			А.	<u>⊨</u> 5 x 2 i ∞ € ∓	HI AI		T		Q		
#		Parameter 🔎	UOM	Schedule Type		Threshold	Interval	Alert Value	Positive Tolerance	Negative Tolera	nce				
1		FC	CYC	Recurring	*	10000.00	5000.00	4500.00	200.00			200.00			
2				Recurring	*										
						Edit U	sage Based Schedule								
-															

Figure 3.10 Creating usage based schedule

- 2. Use the **Work Unit #** drop down list box to select the work unit to be performed. The system displays all the work units defined in the component maintenance program for the part except those with Program Item Type-Perpetual and As Required; prefixes it with "TSK", "SP" or "WP" based on the type of the work unit.
- 3. . Click the **Get Details** pushbutton to get the work unit information.
- 4. Enter **Parameter**, the number identifying the consumption parameter in the aircraft.
- 5. Select the Schedule Type, which could be "One Time" or "Recurring".
- 6. Enter the starting value of performing the work unit in the **Threshold** field.
- 7. Enter the value after which an alert need to be given, in the Alert Value field.
- 8. Enter the Positive Tolerance and Negative Tolerance.
  - > Note: The system displays the parameter values in the multiline based on the



option "Decimal Format" or "HHMM Format" selected for the parameter in "Time Display Option" drop-down list box in the "Aircraft" business component while parameter creation. For e.g. If the Time Display Option is selected as "HHMM Format" and the value stored is 2.50 in the data base, the system displays 2.30 in the multiline.

9. Click the Edit Usage Based Schedule pushbutton to record the schedule details.

## 3.1.14 Define aircraft effectiveness

You can specify the aircraft for which the position-based schedule is applicable. If no aircraft is specified, the system considers all aircraft under the model and configuration class.

1. Select Define Aircraft Effectivity link in the Select Position - Part # page. The Define Aircraft Effectivity Information page appears. *See Figure 3.11.* 

Define Aircraft Effectivity Information				44 4 16 17 18 19	9 20 🕨 👐	18 /500	<b>□</b> <i>1</i> 4		4	<b>€</b> 3	0	ĸ
Position Details				Date	Format yyyy-dd-r	mm						
Aircraft Model # 100-00 Part # 0-0440-	4-0001:36361			-	on Class ABC on Code 001							
Aircraft Details									_			
(( ( 1 - 1/1 ) )) + - □ % ○ (1 Y T)       #     Aircraft Reg # ○       Manufacturer #	Aircraft Model #	Manufacturer Serial #	Variable Tab #	▶ 匝 〒 X 図 首 ☆ Nose #	Associated By	All	Asso	viated Da				Q
1 FASS 2 Enter the aircraft				Click this pushbutton to								
< registration number		Edit Effectivit		modify aircraft effectivity							>	

Figure 3.11 Defining aircraft effectivity

- 2. Enter the Aircraft Reg #, the number identifying the aircraft to which the schedule is applicable.
- 3. Click Edit Effectivity pushbutton to record the aircraft effectivity details.

## 3.1.15 Activating a position based schedule

You can authorize a position based maintenance program in the "Fresh" status.

1. Select Activate Position Based Schedule under Component Maintenance Program business component. *See Figure 3.12* 



								Date Format y	/yy-dd-mm		
-) Sea	arch (	Criteria	Aircraft Mod	1.1.0				Conferentian Class			
								Configuration Class		Ŧ	
				art #				Part Description			
			Position (	Code				ATA #			
			Schedule St	tatus 🔻							
						Search					
- Po:	sition	- Part # Details									
44 4		1 - 10 / 12 🕨 🗰	T T.					X 🛛 🗎 🗏 🗐 🛛	Al 🔻		
#		Aircraft Model #	Configuration Class	Position Code	Part #	ATA #	Schedule Status	Schedule Info	Part Description	Updated by	
1		A320-211	AI-707	1	0-0440-4-0001:36361	72-00	Active	DT	ENGINE	DMUSER	
2		A310	AVEOS	POS6	0-0440-4-0001:56561	72-00	Active	DT / USG	IP TURBINE ROTOR BLADE	DMUSER	
3		A310	AI-707	POS-1	0-0440-4-0005:36361	49-00	Active	DT / USG	AUX. POWER UNIT	DMUSER	
4		B767-200	AVEOS	P1	103257:08393	72-00	Fresh	DT	ENGINE	DMUSER	
5		B767-200	ECONOMY	ENG-LH	103257:08393	72-00	Active	DT	ENGINE	DMUSER	
6		B767-200	ECONOMY	ENG-RH	103257:08393	72-00	Active	DT	ENGINE	DMUSER	
7		B767-200	SPEC	NLG	162T0000-24:81205	32-20	Active	DT	NOSE LANDING GEAR ASSEMBLY	DMUSER	
8		B767-200	SPEC	ENG-LH	8590062D:08393	72-00	Active	DT / USG	ENGINE	DMUSER	
9		B767-200	SPEC	ENG-RH	8590062D:08393	72-00	Active	DT / USG	ENGINE	DMUSER	
		B767-200	SPEC	ENG-LH	916393R91:09508	72-00	Active	USG	ENGINE	DMUSER	
10		<									3

Figure 3.12 Activating position based schedule

- 2. Enter **Search Criteria** to search for position based schedule information and click the **Search** pushbutton.
- 3. Check the **Inherit changes to attached Components** box to update maintenance programs of all attached components based on the changes made in the position based schedule. Conversely, leave the box unchecked, if you do not want any change in the maintenance programs of attached components.
- 4. Select the schedule that you want to make active/inactive, in the multiline.
- 5. To activate position based schedules, click the Activate Schedules pushbutton.
- 6. To inactivate position based schedules, click the Inactivate Schedules pushbutton.

The system changes the status of the program to "Authorized" and the program is made available for forecasting.

#### 3.1.16 Updating work units in existing maintenance program

You can add, modify or delete the work unit in the part program or component maintenance program.

- 1. Select the Update Work Units in Programs link under Component Maintenance Program. The Select Work Unit # page appears.
- 2. Provide search criteria to retrieve the work unit that you want to update in Active maintenance programs.
- 3. Click the hyperlinked **Work Unit #** in the **Search Results** multiline. The **Edit Work Unit Information in Programs** page appears. *See Figure 3.13.*
- 4. The system displays the details of the work unit in the **Work Unit Details** group box, the execution details of the program in the **Execution Details** group box and the transaction details in the **Transaction Details** group box. You can modify the existing details.
- 5. In the "Program Details" group box, specify the **Program Group** of the maintenance program planned for the part.
- 6. Specify the **Program Item Type** of the work unit.
- 7. Specify the manner in which the Next Scheduled Date /Next Schedule Date for the work unit in the

#### maintenance program must be computed in the Initiated/Reset By field.

8. Modify the Part #, Serial # and the Component #, if required, in the Part Applicability Details multiline.



9. In the Execution Details group box, select the Expense Type of the work unit/task.



- 10. Use the **Default Exe. Priority** drop-down list box to specify the priority for execution of the work unit with reference to other work units in the maintenance program planned for the part.
- 11. You can add, delete or modify the existing work unit details for a selected part. Select "Append" in the **Action** drop-down list box, to add the work unit details. Select "Delete" to remove the work unit details from the component. Select "Overwrite" to modify the existing work unit details.
  - Note: If the "Action" is set as "Delete", the system deletes only those work units having the same maintenance type as the selected work unit, from the part program or the component maintenance program.
- 12. Click the Edit Work unit Changes pushbutton to record the modified details.
- 13. Click the Cancel Update pushbutton to cancel the modification

#### 3.1.17 Maintaining restoration work units

In this activity, you can identify the work unit that has to be performed for removal and installation of the part. You can identify the restoration task for each part.

1. Select Maintain Restoration work units under Component Maintenance Program business component. The Maintain Restoration Work Unit Information page appears. *See Figure 3.14* 



								Date Format yyyy-	id-mm		
-) 5	earc	h Criteria									
			Part #				Pi	art Description			
			Work Unit Type Task 🔻					Work Unit #			
			ATA #								
					Se	arch					
- R	esto	oration Work Unit Details									
44	•	1 - 10 / 121 🕨 🕨 +	- 0 * ° ° T T						All	•	 
#	E	Part # 🔎	Work Unit Type		Work Unit # 🔎	Comments	Part Description	ATA #	Work Unit Desc		
1	E	0-0440-4-0001:36361	Task	*	1-50C-0000-CMM-00000053		ENGINE	72-00	PME-1		
2	E	0-0440-4-0001:36361 RO	Task	*	1-50C-0000-CMM-00000057		test	72-00	PME-5		
3	E	0-0440-4-0001:36361 RO SUB	Task	~	1-50C-0000-CMM-00000057		test	72-00	PME-5		
4	E	0-0440-4-0001:36361 TEST	Task	*	1-50C-0000-CMM-00000057		test	72-00	PME-5		
5	E	0-0440-4-0001:56561	Task	~	1-50C-0000-CMM-00000053		IP TURBINE ROTOR BLADE	72-00	PME-1		
6	E	0-0440-4-0005:36361	Task	*	3-00000007		AUX. POWER UNIT	49-00	Restoration Task		
7	E	0-0440-4-0006:36361	Task	*	1-50C-0000-9X-00005057		ATLAS, (LO-COST) CARRIER	20-00	Repair - 01		
8	E	0-0440-4-0011:36361	Task	~	1-50C-0000-CMM-00000369		APU	49-00	PME-1		
9	E	0-0440-4-0015:36361	Task	*	1-50C-0000-CMM-00000053		PS9323 CARRIER	25-30	PME-1		
10	E	0-0440-4-0016:36361	Task	*	1-50C-0000-CMM-00000053		LARGE OVEN RACK	20-00	PME-1		
		<									

Figure 3.14 Maintain restoration work units

- 2. Enter the search criteria in the **Search Criteria** group box to retrieve parts. Click the **Search** pushbutton. The system retrieves all active parts and displays the details in the **Search Results** multiline.
- 3. Use the **Work Unit Type** drop-down list box to select the type of work unit. Work unit type can be "standard procedure" or "task".
- 4. Enter the number identifying the task or standard procedure that has to be performed to remove or attach the part in the **Work Unit #** field.
- 5. Enter any comments regarding the association of the work unit to the part in the **Comments** field.
- 6. Click the **Edit Work Units** pushbutton to update the work unit details for the part.

#### 3.1.18 Managing Schedule Adjustment for Components

You can set maintenance schedules adjustments for parts at various levels, such as Model, Position Code, Operator, Model & Operator and Operator & Position Based. This helps you to create variants of maintenance programs for specific model, position code or operator.

By means of this activity, you can make adjustments to the following for a part:

- Schedule Information
- Date based schedules
- Usage based schedules
- Aircraft Effectivity

The schedule adjustments incorporated in the part program can be inherited by the associated components, if currently attached to an aircraft. If the components are not attached to aircraft at the time of activation, the schedule adjustments will be incorporated in the component maintenance program upon attachment.

However, you cannot define more than a single schedule adjustment for a specific model, position code, operator, Model & Operator and Operator & Position Based.

1. Select the Manage Schedule Adjustments for Components under the Component Maintenance Program business component. The Manage Schedule Adjustments for Components page appears. *See Figure 3.15.* 

#### 69 | Maintenance Programs

								-
	ge Schedule Adjustments for Comp	onents				24	ē ⊄ ?	
Р	Component Schedules Adjustment	Part Description			PRG defined?	Schedule Adjustment At		
Search Criteria	Aircraft Model 🔎		Configuration C	lass	T	Position Code # 👂		
	Operator #	•	St	atus	▼ Court			
Schedule Adju	stment Details				Search			
4	[No records to display] 🕨 🕨 + 🗇	T Tx		7		7 10 14 🗡 Al	T	
🔲 Model #	# D Configuration Class	Position Code #	Operator #	Status	Schedule Info?	Date Based Sch.	Usage Based Sch.	
		*		*				
🗉 Model #		Position Code #	Operator #	Status	1		1	
<								>
rit Changes to At	tached Component Save		Activate			Inactivate		

Figure 3.15 Managing schedule adjustments for parts / components

- 2. Select **Part #** in the **Part Details** group box for which you want to create / update schedule adjustment.
- 3. In the Manage Part-Component Schedules Adjustment group box, use the Schedule Adjustment At drop-down list box to select the level at which you wish to adjust schedules for the part. The drop-down list box displays the following: Model, Position Based, Operator, Model :: Operator and Operator :: Position Based.
- 4. In the Schedule Adjustment Details multiline, enter Model #, Configuration Class, Position Code # and Operator # for the part.
- 5. To ensure that the part schedule adjustments are copied to the maintenance program of the components attached to aircraft, select the **Inherit Changes to Attached Component** check box. By default, the check box remains selected.
  - Note: Only Active schedule Adjustment will be inherited by the components attached to aircraft. However, for components not currently attached to aircraft, schedule adjustments will be carried upon component attachments in future.
- 6. To update schedule details for the part, click the data hyperlink in the **Schedule Info?** column to open the **Edit Schedule Information** page.
- 7. To update the date based schedules for the part, click the data hyperlink in the **Date Based Sch.** column to open the **Edit Date Based Schedule Information** page.
- 8. To update details of the usage based schedules for the part, click the data hyperlink in the **Usage Based Sch.** column to open the **Edit Usage Based Schedule Information** page.
- 9. To update aircraft effectivity details based on schedule adjustments, click the data hyperlink to open the **Define Aircraft Effectivity** page.
- 10. Click the Save pushbutton to save the entered details.
  - >> Note: The status of the record becomes Fresh upon saving the details.
- 11. Click the Activate pushbutton to activate the schedule adjusted maintenance program.
  - Note: You can activate an Adjustment record only after saving the record details. The status of the record becomes Active after it is made Active.
- 12. Click the **Inactivate** pushbutton to inactivate the schedule adjusted maintenance program.

#### Conditional availability of fields based on Schedule Adjustment

The fields in **Search Criteria** and the **Schedule Adjustment Details** multiline will be available / unavailable based on the selection you have made in the **Schedule Adjustment At** drop-down list box. The next table shows the field availability for each option in the **Schedule Adjustment At** drop-down list box.

ramco



Schedule Adjustment At option	Fields Displayed	Fields Hidden
Model	Model	Configuration Class Position Code #, Operator,
Position Based	Model #, Configuration Class, Position Code #	Operator
Operator	Operator	Model #, Configuration Class, Position Code #
Model :: Operator	Model #, Operator	Configuration Class, Position Code #
Operator :: Position Based	Model #, Configuration Class, Position Code #, Operator #	

# 4AIRCRAFT MAINTENANCE PROGRAM

The Aircraft Maintenance Program business component addresses the function of configuring and maintaining the standard maintenance routines of aircraft. The maintenance program identifies the scope (work packages, standard procedures and tasks) of the planned maintenance as well as the periodicity with which the maintenance needs to be performed.

A maintenance program is created for an aircraft model. Individual aircraft can later be associated to the maintenance program. The work units and the schedules that are defined for a specific aircraft maintenance program are later used for forecasting the maintenance activities that are due to be carried out on the aircraft, in the "Aircraft Maintenance Forecast" business component. Later, work orders are generated for carrying out the work units in the "Hangar Work Order" business component.

The following are the various stages and the statuses of an aircraft maintenance program:

Creation of an aircraft maintenance program – A maintenance program is created for an aircraft model and work units that must be carried out as part of the program are identified. The status of the maintenance program is set to "Fresh". The maintenance program can be modified or canceled only when it is in the "Fresh" status.



- Cancellation of the aircraft maintenance program A maintenance program can be cancelled when it is in the "Fresh" status. The status of the program is set to "Cancel".
- Confirmation of the aircraft maintenance program Once the program is confirmed, further modifications are not possible. Only a confirmed program can be activated. The status of the program is set to "Confirm".
- Activating, Returning or Inactivating the maintenance program:
  - Activation of the program The program is activated and the status of the program is set to "Active".

Or

 Returning the program – If the program is not activated it can be returned for further modification. The status of the program is set to "Return".

Or

- Inactivating the program An activated program can be inactivated. The status of the program is set to "Inactive".
- Revising the program A program that is in the "Active" or "Inactive" status can be revised. In such cases, a new revision of the program is created in the "Fresh" status. When this new revision of the program is activated, the previous active revision of the program is set to "Revised".

*See Figure 4. 1* to view the diagrammatical representation of the various stages and the statuses of an aircraft maintenance program:




# 4.1 Configuring the maintenance program for aircraft models

# 4.1.1 Creating quick codes

Quick Codes are user-defined values, used to categorize a set of details of identified behavior. These quick codes are later used in the process of retrieving or addressing the details by referring to the attached quick code.

Quick codes act as additional qualifiers for a business entity or document. Quick codes can assume user-provided values, which can be used to categorize/group an entity/document record to satisfy specific needs in a user organization's internal processes, especially with respect to unique reporting requirements.

The quick code type "Package Category" and "Program Category" are predefined in the system. Values can be defined for this quick code type.

1. Select the Create Quick Codes link under the Aircraft Maintenance Program business component.

*		Quick Code Information			<b>Ⅲ</b> 7\$		•	?	To K
- Q	uick (	Quick Code Type Use	Status for Verify						
44	•	1 - 2 / 2 🕨 🗰 🕂 🗖	* 0 G T T.			Ŧ			Q
#		Quick Code	Description						
1		Confirmed	AMP confirmed						
2			AMP Authorized						
3									
				Create Quick Codes					
								_	

The Quick Code Information page appears. See Figure 4. 2

Figure 4.2 Creating quick codes

- 2. Select the quick code type for which quick codes must be defined, in the **Quick Code Type** field.
- 3. Set the field to **Package Category** to define quick codes for work package.
- 4. Set the field to Program Category to define quick codes for aircraft maintenance program.
- 5. Set the field to **MPD Group** to define quick codes for Maintenance Planning Document (MPD) group.
- 6. Enter a unique identifier for the quick code in the **Quick Code** field in the **Quick Code Details** multiline.
- 7. Enter the textual description for the quick code in the **Description** field.
- 8. Click the Create Quick Codes pushbutton to create the quick codes.
  - Note: The system creates the quick codes and sets the status of the created quick codes as "Active".

A

# 4.1.2 Setting options for aircraft maintenance program

You can set default options for the various fields in the activities of the Aircraft Maintenance Program business component.

1. Select the **Set Options** link under the **Aircraft Maintenance Program** business component. The **Set Options** page appears. *See Figure 4. 3* 



* 🗎 Set Options	드 · · · · · · · · · · · · · · · · · · ·	← ? 🗔 🗖
	Date Format yyyy-dd-mm	
Option Setting Details     Sub Fleet Security Not Required		
Revision History Maintenance Policy Detailed 💌		
Work Center Entry For Task Optional 🔻		
Set Options	Decides whether previous	
Record Statistics     Last Modified by DMUSER	revisions of a program should be maintained or not	

Figure 4.3 Setting options for aircraft maintenance program

- 2. Enable or disable security check on modification of maintenance program associated to a sub fleet, by selecting one of the following options in the **Sub Fleet Security** field:
  - Set the field to Required to enable security check on modification of a maintenance program associated to an aircraft which is part of the sub fleet. When this option is selected, the user, who does not hold rights for a particular sub fleet, will not be allowed to modify the maintenance program of an aircraft belonging to the sub fleet.
  - Set the field to Not Required to disable security check on modification of a maintenance program associated to a sub fleet. When this option is selected, users will be allowed to modify the maintenance program of an aircraft belonging to a sub fleet, irrespective of whether they hold rights for the sub fleet or not.
- 3. Set the policy for maintaining previous revisions of a program, in the **Revision History Maintenance Policy** field.
  - Set the field to Summary if you wish to maintain a summary of the previous revisions of the program. When this option is selected, the system displays No in the Maintain Prev Rev.? field in the Approve Maintenance Program activity.
  - Set the field to Detailed if you wish to maintain all the details of the previous revisions of the program. When this option is selected, the system displays Yes in the Maintain Prev Rev.? field in the Approve Maintenance Program activity.
- 4. Select any one of the following options in the Work Center Entry for Task field:
  - Optional Indicates the work center entry for the task is optional.
  - Mandatory Indicates the work center entry for the task is mandatory.
- 5. Click the Set Options pushbutton to set the options for the maintenance program.

# 4.1.3 Maintaining maintenance events

You can define customized maintenance event codes and their description. You can also specify the status of the maintenance event.

1. Select Maintain Maintenance Events under the Common Master business component. The Maintain Maintenance Events page appears. *See Figure 4. 4* 

#### 75 | Maintenance Programs



	) I	Maintain Maintenan	ice Events						== ≭ = ₽ ← ? □	¢
- Ev	ent f	etails						Date Format yyyy-dd-mm		
			-0%0017						•	
		Event #	Event Description	Status		Created by	Created Date	Last Modified by	Last Modified Date	
		BELLY LANDING	Belly Landing	Active	*	DMUSER	2016-08-01	DMUSER	2016-08-01	
		BIRDHIT	Bird Hit	Active	*	DMUSER	2015-18-09	DMUSER	2015-18-09	
		MNT_1	mnt_1	Active	*	DMUSER	2015-25-08	DMUSER	2015-25-08	
				Active	~					
		<								
						Maintain Events				
						Haircair Evenes				

Figure 4.4 Maintaining maintenance events

In the Event Details multiline,

- 2. Enter the Event # and Event Description.
- 3. Set the **Status** of the maintenance event to "Active" or "Inactive", to specify whether the maintenance event is to be enabled or disabled for future reference.
- 4. Click the Maintain Events pushbutton at the bottom of the page to save the maintenance event details.

# 4.2 Creating an aircraft maintenance program

An aircraft maintenance program identifies the work package, standard procedure and tasks of the planned maintenance, as well as the periodicity with which the maintenance must be performed. Typically, aircraft of the same model will have similar maintenance needs. Therefore, the same maintenance program can be applied to all such similar aircraft. Besides routine maintenance, specialized maintenance activities such as corrosion prevention programs can also be configured in the maintenance program.

- 1. Select the **Create Aircraft Maint. Program** link under the **Aircraft Maintenance Program** business component. The **Create Maintenance Program** page appears. *See Figure 4. 5*
- 2. Enter a unique number that identifies the maintenance program in the **Maintenance Program #** field in the **Maintenance Program Information** group box.
- 3. Enter the textual description of the maintenance program, in the Program Desc. field.
- 4. Specify the aircraft model to which the maintenance program is applicable, in the **Primary Model #** field.
- 5. Select the configuration class for the maintenance program, in the **Configuration Class** field.

You can create a maintenance program by copying the details from an existing maintenance program.

- 6. Enter the maintenance program number, the details of which must be copied into the maintenance program that is currently being defined, in the **Maintenance Program #** field of the **Copy Details** group box.
  - Note: Ensure that the number entered here is different from the maintenance program number entered in the "Maintenance Program Information" group box.
- 7. Check the **Copy From** box to copy the details from the required maintenance program.
- 8. Check the **Copy All** box to copy all details of the maintenance program.
- 9. Check the **Schedule Information** box to copy only the schedule information to the maintenance program that is being created.
- 10. Check the Model Effectivity box to copy only the model effectivity details.
- 11. Check the Associated Aircrafts box to copy only the details of associated aircraft.

*	Cre	ate Maintenance Program	L						RAMCO	OU-ramco role 👻 😂 !		<b>←</b> ? [₀
									Date F	ormat mm-dd-yyyy		
	aintenanc	Prima	Program # AMP 2101 gram Desc	The status program is "Fresh" on	set to				class CA V			
	ork Scope	Maintenance Program # A320-	211 D Copy Fro		✔ All ✔ Mod		ichedule Information Issociated Aircrafts	I				
•			< O % T 7				× C X I	-14	10 14 % All	▼ Search		Q
#		MPD Group	Work Unit # O	Prog. Item Type		Default Exe. Priority	Initiated/ Reset by		Maint. Operator #	Execution Type		Job Type
1			0000-737-0006477	Block	~		Self Compliance	~		Major	~	Aircraft
2		¥	2-B737-0000-MOD-00004404	Base	~		Self Compliance	~		Major		Aircraft
3		~	1-A310-0000-CMM-00003926	As Required	~	*	Self Compliance	~		Major	~	Aircraft
4		~	1-A310-0000-CMM-00003926	Event Driven	~	~	Self Compliance	~		Major	~	Aircraft
5		~			~	~	Self Compliance	~		Major	~	Aircraft
		<										>
	eq #					Get Base Task						
Edit S	hedule Info	ormation	Ma	iintain Task Relationship			View Ta	sk				
<b>A</b>	oproval De	t <b>tails</b> Regulatory Authority Approval # Other Details	ANAC V							Description Date		
		Cre	eate Maintenance Program				Co	onfirm	Maintenance Progra	m		
	Create Maintenance Program     Confirm Maintenance Program       rcraft Model Effectivity Information     Associate Aircrafts     Edit Maintenance Event Information       derences     Upload Documents     View Associated Doc. Attachments											

#### Figure 4.5 Creating aircraft maintenance program

- 12. Select the Maintenance Planning Document (MPD) Group from the MPD Group drop-down list box
- 13. Enter the task, standard procedure; aircraft work package or component work package to be performed as part of the maintenance program, in the **Work Unit #** field.
- 14. Select any one of the following valid Program item types from the **Prog. Item Type** drop-down list box:
  - Block Indicates the task contains a group of tasks to be executed based on a schedule
  - Base Indicates the task to be executed is associated to a block.
  - Perpetual Indicates the task is to be executed while executing any other tasks on the aircraft.
  - As required Indicates the task will be executed whenever required.
  - Event driven Indicates the task will be executed based on the occurrence/ triggering of an event
  - Non-block Indicates the task to be executed is not associated to a block.
  - To be decided Indicates the classification of the program item type for the task will be decided later.
- 15. Select the default execution priority from the **Default Exe. Priority** drop-down list box.
- 16. Select any one of the following options from the Initiated/Reset by drop-down list box:

ramco

- Self-Compliance Indicates the task would be initiated/ reset on compliance of the same task. This implies that the compliance of this task is not dependent on any other task.
- Related Task Compliance Indicates the task, on compliance, would initiate/ reset another task available as part of the same program. The task, which has to be initiated on compliance of a task is based on the "Initiate Schedule" relationship defined for that task in the 'Maintain Task Relationship' activity under the "Maintenance Task" business component.
- 17. Set the **Execution Type** field to **Major**, to indicate that the work unit is a major check and must be sent to the hangar for maintenance. Set the **Execution Type** field to **Minor**, to indicate that the work unit must be executed as a part of line maintenance.
- 18. Set the **Job Type** field to **Aircraft** to indicate that the work unit must be performed on the aircraft. Set the **Job Type** field to **On-wing** to indicate that the work unit must be performed on components attached to the aircraft.
  - Note: If the job type is "Aircraft" or "On-wing" and if the work unit type is "Aircraft Work Package", then ensure that the work unit number is valid for the aircraft model selected in the "Primary Model #" field.
- 19. Specify the type of maintenance job that must be carried out as a part of the work unit in the **Maintenance Type** field. The type of the maintenance can be **Repair**, **Overhaul**, **Inspection** or **Others**.
- 20. Enter the position code in the aircraft where the work unit must be performed, in the **Position Code** field.
- 21. The number identifying the level at which the position code occurs in the configuration class attached to the primary model in the **Level Code** field.
- 22. Enter the engineering document as a part of which the work unit was defined along with its revision number, in the **Eng Doc#** and the **Revision #** fields.
- 23. Specify the status of the schedule, in the Schedule Status field.
  - Set the status of the schedule to **Fresh** when the work unit is first specified for the program.
  - Set the status of the schedule to **Active** if you wish to activate the work unit.
  - Set the status of the schedule to **Inactive** if you do not wish to carry out the work unit as a part of the maintenance program.
  - Set the status of the schedule to **Terminated**, to terminate the work unit. This status gets updated from an Engineering Document as a part of which the work unit was defined and is updated for the work unit and for a specific aircraft registration number.
- 24. Set the **Long Term?** field to **Yes** to indicate that the execution of the work unit is for long term. Otherwise, select the **No** option.
- 25. Specify the schedule execution of the work unit in the Sch. Exec Rule field.
  - Set the field to **Earliest** to indicate that when both date based and usage based schedules is defined for the work unit, the system must consider the earliest of the due dates calculated by these schedules.



- Set the field to **Latest** to indicate that when both date based and usage based schedules is defined for the work unit, the system must consider the latest of the due dates calculated by these schedules.
- 26. Specify the type of deferment policy for deferring or postponing the work unit when work orders are generated based on the maintenance program, in the **Deferment Policy** field.
  - Set the field to **Not Allowed** if you do not wish to allow deferment of the work unit.
  - Set the field to **Complete Work Unit**, option if you wish to allow the deferment of the work unit along with the tasks and procedures defined for it. For example, If WU1 is a work package and its contents are T1, a task and SP1, a procedure, and if T1 is marked for deferring the whole package WU1 gets deferred.
  - Set the field to **Constituent Unit**, if you wish to allow the deferment of the work unit separately. For example, If WU1 is a work package and its contents are T1, a task and SP1, a procedure, T1 and SP1 can be deferred separately without affecting the parent work package.
- 27. Select the Package Type of the aircraft.
- 28. Select the expense type of the work unit in the **Expense Type** field.
  - Set the field to **Revenue**, if the cost incurred on the work unit is for routine maintenance activity and must be recorded as revenue cost.
  - Set the field to **Capital**, if the cost incurred on the work unit must be recorded as a capital cost.
- 29. The work center where the work unit must be performed in the Work Center # field.
- 30. The sequence in which the work unit should be performed as part of the maintenance program in the **Seq #** field.
- 31. Click the Get Base Task pushbutton to retrieve the base tasks for the selected block task.
  - The system retrieves the tasks in the Work Scope Details multiline. These tasks would have been mapped to the selected parent Task # with the "Block Schedule'" relationship defined in the "Maintain Task Relationship" activity of the "Maintenance Task" business component.
- 32. Select the regulatory authority as a reference to the maintenance program in the **Regulatory Authority** field in the **Approval Details** group box.
- 33. Enter the approval number with reference to which the maintenance program is created, in the Approval # field.
- 34. Select the Create Maintenance Program pushbutton to create the maintenance program.
  - Note: The system creates the maintenance program and sets the status of the created maintenance program as "Fresh".

- 35. Select the **Confirm Maintenance Program** pushbutton to confirm the maintenance program.
  - Note: The system confirms the maintenance program and sets the status of the program to "Confirm". Further modifications to the program are not allowed. On confirmation, the system changes the 'Schedule Status' from "Fresh" to "Active".

To provide further details,

- $\tau$  Select the **Edit Schedule Information** link to specify the schedule information for the maintenance program.
- $\tau$  Select the **Edit Aircraft Model Effectivity Information** link to specify aircraft models on which the maintenance program is applicable.
- $\tau$  Select the **Associate Aircrafts** link to associate aircraft to the maintenance program.
- $\tau$  Select the **Edit Maintenance Event Information** link to specify the maintenance events that must be performed as a part of the maintenance program.
- $\tau$  Select the **Edit References** link to specify reference document details based on which the maintenance program was created.
- $\tau$  Select the **Upload Documents** link to upload documents associated with the maintenance program of the aircraft model.

# 4.2.1 Specifying schedules for carrying out the maintenance program

The periodicity of a maintenance activity is defined in the form of schedules. Schedules are defined for executing the various work units as a part of the maintenance program for the aircraft model. These schedules are defaulted to aircrafts that are associated to the maintenance program. You can then forecast the maintenance activities that are due for an aircraft in the "Aircraft Maintenance Forecast" business component. Work orders are generated based on these schedules in the "Hangar Work Order" business component.

Refer the "Aircraft Maintenance Program" Online Help for more details on schedules.

- 1. Select the Edit Schedule Information link in the Create Aircraft Maintenance Program or the Edit Aircraft Maintenance Program page.
- 2. The Edit Schedule Information page appears. See Figure 4. 6.
- 3. Select the work unit for which schedule information must be specified in the **Work Unit #** field in the **Work Unit Details** group box.
- 4. Click the **Get Details** pushbutton.
- 5. Specify whether the work unit must be executed only once or repeated at certain intervals, in the **Schedule Type** group box in the **Schedule Details** group box.
  - Set the field to One Time if the work unit must be executed only once.
  - Set the field to **Recurring** if the work unit must be repeated at certain defined intervals.
- 6. Select the mode by which the due dates for executing the work units must be calculated, in the **Update Basis** field.
  - Set the field to Schedule if the next due dates for carrying out the work units must be calculated based on a fixed interval.



Set the field to **Actual Completion** if the next due dates for carrying out the work units must be calculated based on its previous completion.

★ 🗎 Edit Schedule Information					<b></b>	다 좀 다	← ? 🗔
Maintenance Program Information				Date Format	yyyy-dd-mm		
Maintenance Program # //				Revision #			
Program Desc. /	A1 Weekly			Status F	Fresh		
	1-PME-000000001-1	Get Details					
Work Unit Description	Aircraft task for print task card			Work Unit Type	Task		
Execution Type	Major			Job Type	Aircraft		
- Schedule Details							
Schedule Type	Recurring 👻			Update Basis	Actual Completion 🛛 👻		
<ul> <li>Date Based Schedule Details</li> </ul>							
Time Unit	Days 🔻			Reference Date Basis	•		
Threshold Interval				Alert Interval			
Positive Tolerance				Negative Tolerance			
Repeat Fixed Interval				Repeat Floating Interval			
Terminating Interval							
<ul> <li>Usage Based Schedule Details</li> </ul>							
(4) 4 1 -1/1 > >> + = □ ≤ ∅ ∅	T T.			🗎 🎫 C 💶 🖷 💷	Al	T	Q
# Parameter UOM Threshold Repeat Fixed	Repeat Floating Alert Interva	Positive Negativ	e Terminating	Parameter Description			
1 FH 10000.00 20	1900.00	200.00	-200.00				
2 🗉							
Enter the parameter							
		Edit Schedules					

Figure 4.6 Editing schedule information for a maintenance program

To define date-based schedules,

- 7. Specify the time unit for the date-based schedule in the **Time Unit** field in the **Date Based Schedule Details** group box. The time unit can be **Days, Months, Month End**" or **"Years"**.
- 8. Specify the reference date based on which the schedules for the work unit must be ascertained, in the **Reference Date Basis** field.
  - Set the field to **Operational Date** if the schedules for executing the work units must be calculated based on the operational date of the aircraft.
  - Set the field to **Induction Dat**e if the schedules for executing the work units must be calculated based on the induction date of the aircraft.
- 9. Enter the period from the induction date or the operational date, for which the due dates for the work units must be calculated in the **Threshold Interval** field.
- 10. Enter the period at which an alert message must be displayed about an impending due date for performing the work unit in the **Alert Interval** field.
- 11. Enter the maximum time period subsequent to the next schedule date, within which the work unit must be performed, in the **Positive Tolerance** field.
- 12. Enter the maximum time period ahead of the next schedule date, within which the work unit could be performed, in the **Negative Tolerance** field.
- 13. Enter the interval after which the work unit must be repeated irrespective of whether its previous execution is completed or not, in the **Repeat Fixed Interval** field.
- 14. Enter the interval after which the work unit must be repeated, from the period of its last execution and completion in the **Repeat Floating Interval** field.
  - Note: You must specify either Fixed Interval or Repeat Floating Interval for a date-based recurring task. However, for a recurring task with both date-based and usage-based

schedules, you can specify.

15. Enter the period reaching which the work on the work unit must be terminated in the **Terminating Interval** field.

To define usage-based schedules,

- 16. Enter the parameter based on which the work unit must be performed in the **Parameter** field in the **Usage Based Schedule Details** multiline.
- 17. Enter the parameter value at which the first execution of the work unit is due in the Threshold Value field.
- 18. Specify the interval after which the work unit must be repeated irrespective of the completion of its last execution, in the **Repeat Fixed Interval** field.
- 19. Specify the interval after which the execution of the work unit must be repeated, from the period when it was last executed, in the **Repeat Floating Interval** field.
  - Note: If the usage-based schedule for a task comprises of many parameters, you must specify repeat fixed interval or the repeat floating interval for a minimum of one of the parameters. However, if the usage-based schedule comprises of single parameter, repeat fixed interval or repeat floating interval is mandatory for the lone parameter. For a task with both date-based and usage-based schedules, you must specify one of the below-listed,
    - 1. Repeat fixed interval or repeat floating interval for date based schedule
    - 2. Repeat fixed Interval or the repeat floating interval for at least one parameter comprising the usage-based schedule.
- 20. Enter the parameter value at which an alert message would be displayed about an impending work due on the work unit, in the **Alert Interval** field.
- 21. Enter the upper limit of the parameter value greater than the next schedule value, within which the work unit must be performed, in the **Positive Tolerance** field.
- 22. Enter the lower limit of the parameter value lesser than the next schedule value, within which the work unit could be performed, in the **Negative Tolerance** field.
- 23. Specify the parameter value, reaching which the work on the work unit must be terminated, in the **Terminating Value** field.
- 24. Click the Edit Schedules pushbutton.
  - > Note: The system saves the schedule details against the maintenance program.

### 4.2.2 Identifying the maintenance events to be carried out for the program

You can specify the various maintenance events that must be performed as a part of the maintenance program. You can select the maintenance event that must be performed and specify the various work units that must be executed as a part of the maintenance event for the maintenance program. A maintenance event can be a "Pre Flight Check", "Night Haul Check" etc.

1. Select the Edit Maintenance Event Information link in the Create Maintenance Program or the Edit Maintenance Program page. The Edit Maintenance Event Information page appears. *See Figure 4.* 7

#### 83 | Maintenance Programs

Maintenance Program Inform	nation											
	Maintenance Program #	A1 WEEKLY				F	levision # 0					
	Program Desc.	A 1 Weekly					Status Fresh					
Dependent Work Unit Details	Maintenance Event	MNT_1	Get Details									
		T <sub>x</sub>			<u>له</u> ا	0 x C 🖻 × 0		All		T		\$
🗉 Work Unit # 🔎	Work Unit Type	Work Center # 🔎	Priority		Est. Man Hrs.	Time Unit	Est. Time	Mandatory?		Work Un	t Desc	
3-874-00-MPD-15240	Task	0210	AOG	~				Yes	~	Flap actu	ator inspec	tion
3-B74-00-MPD-15243		·		~				Yes	~	Flap actu	ator inspec	tion
3-874-00-MPD-15246		·		~				Yes	~	Flap actu	ator inspec	tion
				*				Yes	*			
	carried out as event for the											

Figure 4.7 Identifying the maintenance events to be carried out for the program

- 2. Select the maintenance event for which dependent work units must be specified in the **Maintenance Event** field.
- 3. Click the **Get Details** pushbutton.
- 4. Specify the type of the work unit, in the Work Unit Type field. The work unit type can be Task, Standard Procedure, Aircraft Work Package or Component Work Package.
- 5. Specify the work center where the work unit must be performed in the **Work Center #** field.
- 6. Set the **Mandatory**? field to **Yes** to indicate that the execution of the work unit is mandatory in the work package. Otherwise, set the field to **No**.
- 7. Click the **Edit Maintenance Event** pushbutton to save the details of the work units for the maintenance program maintenance event combination.

# 4.2.3 Capturing document references for the maintenance program

1. Select the Edit References link in the Create Maintenance Program or the Edit Maintenance Program page. The Edit References page appears. *See Figure 4. 8* 

	Edit F	References					≣ <i>'</i> ,	-	₽ ←	?	Cō.	ĸ
-	lainte	nance Program Details										<u> </u>
		Maintenance F	Program # A1 WEEKLY			Revision #						
		Prog	gram Desc. A1 Weekly			Status Fresh						
	eferer	nce Document Details										
44	•	1 -1/1 > >> + - 🗇 🛠 🗘	A A T T				All					Q
#		Ref. Document Type	Ref. Document #	Document Name	File Name 🔎		Remarks					
1			Ref. Doc. 12345	AME guidelines	Trn-11A112AV-T'w	vay-Maintenance Program.ppt						
2		*										
		Select this link to									>	2
View	File	view the file.		Edit Reference	ies							_
-												-

#### Figure 4.8 Capturing document references for the maintenance program

- 2. Specify the type of the reference document for which the maintenance program is created in the **Ref. Document Type** field in the **Reference Document Details** multiline.
- 3. Enter the reference document number in the **Ref. Document #** field.
- 4. Click the Edit References pushbutton.

ramco

Note: The system saves the details of the reference documents against the maintenance program

# 4.2.4 Defining model effectivity for the maintenance program

1. Select the Edit Aircraft Model Effectivity Information link in the Create Aircraft Maint. Program or the Edit Maintenance Program page. The Edit Aircraft Model Effectivity Information page appears. *See Figure 4. 9* 

*		Edit Aircraft Model Effectiv	ity Information					≣ x;		+	?	¢ K
	Mainte	nance Program Information										
		Mainter	nance Program # A1 WEEKLY			Revision #						
			Program Desc. A1 Weekly			Status Fresh	1					
	Aircraf	t Model Details										
	•	1 -1/1 > >> + - 🗇	* © © T T,				All		•			Q
#		Aircraft Model # 🔎	Model Description		Model Type	Primary Program?						
1		A320-211	Airbus 320-211			No						*
2						No						¥
					Click this p	ushbutton to odifications						
-				Edit Aircraft Model Effectivity	~							_
-												

Figure 4.9 Defining model effectivity for the maintenance program

- 2. Enter the aircraft model for which the maintenance program is applicable in the **Aircraft Model #** field in the **Aircraft Model Details** multiline.
- 3. Set the **Primary Program?** field to **Yes** to assign the maintenance program as the primary program for the aircraft model.
- 4. Click the Edit Aircraft Model Effectivity pushbutton to save the model effectivity details for the program.

# 4.3 Associating aircraft to maintenance program

# 4.3.1 Associating multiple aircraft to the maintenance program

- 1. Select the Associate Aircrafts to Program link under the Aircraft Maintenance Program business component.
- 2. The Select Maintenance Program page appears.
- 3. Enter the program to which the aircraft must be associated and select the **Associate Aircrafts to Maintenance Program** link provided alongside.

Or

4. Search for the program and click the hyperlinked program number in the multiline. The **Associate Aircrafts to Maintenance Program** page appears. *See Figure 4. 10* 

	<b>Asso</b>	ciate Aircraft to Maintenar	nce Program			44 4 6 7	8 9 10 > >> 9 /39	) 📰 🖂		+	?	Cø
_	ainto	nance Program Details					Date Format yyyy-dd-mm					
	anne	-	nance Program # A330-C CHECK				Revision #					
			Program Desc. C-check maintenance				Status Active					
- 4	ircraf	t Model Details										_
	4	1 -1/1 > >> + - 🗇	* © © T T.				X2 🖡 🛱 💷 🗛		T			Q
#		Aircraft Reg # 🔎	Aircraft Model #	Model Description			Model Type	Sub Fleet #				
1		1101	A310									
2	E											
		Select this link to the program det										>
				Associate Aircraft	Confirm Maint. Program							
Edit A	rcraft	Specific Maint. Program	Track Maintenance C	ompliance History								

Figure 4.10 Associating multiple aircrafts to the maintenance program

5. Enter the registration number of the aircraft, which must be attached to the maintenance program in the **Aircraft Reg #** field in the **Aircraft Model Details** multiline.

> Note: Ensure that the aircraft registration number is not repeated in the multiline.

- 6. Specify the status of the aircraft association in the Association Status field.
  - Set the field to **Fresh** if you are associating the aircraft to the program for the first time.
  - Set the field to **Active** to activate the association of the aircraft to the program.
  - Set the field to **Inactive** to inactivate the association of the aircraft to the program.
  - Note that the system allows you to associate multiple aircraft maintenance programs to an aircraft. However, only one among these maintenance programs can be in "Active" status.
- 7. Set the **Primary Program?** field to **Yes** to set the program as the primary program for the aircraft. Otherwise, set the field to **No**.
- 8. Click the Associate Aircraft pushbutton to associate aircraft to the maintenance program.
  - Note: The system associates the aircraft to the maintenance program and copies the work units defined for the program along with its schedules and all the resources

defined for carrying out the program.

- 9. Click the Confirm Maint. Program pushbutton to confirm the maintenance program.
  - Note: You can confirm a maintenance program only when it is in the "Fresh" status. The system confirms the maintenance program and sets the status of the program to "Confirmed". Further modification of the program is not possible.
  - Note: On confirmation, the system changes the "Association status" from "Fresh" to "Active".

To provide further details,

- τ Select the Edit Aircraft Specific Maint. Program link to edit the program details
- $\tau$  Select the **View Compliance Details** link to view the compliance details of an aircraft maintenance program

### 4.3.2 Associating multiple programs to an aircraft

- 1. Select the **Associate Programs to Aircraft l**ink under the Aircraft Maintenance Program business component. The **Select Aircraft** page appears.
- 2. Enter the aircraft to which programs must be associated and select the **Associate Maintenance Programs** link provided alongside.

Or

- 3. Search for the aircraft and click the hyperlinked aircraft registration number in the multiline. The Associate Maintenance Programs to Aircraft page appears. *See Figure 4. 11*
- 4. Enter the maintenance program to be associated to the aircraft in the **Maintenance Program #** field in the **Maintenance Program Details** multiline.
  - Note that the system allows you to associate multiple aircraft maintenance programs to an aircraft. However, only one among these maintenance programs can be in "Active" status.

								1 2 3 4 5 🕨	- ,		
	0.0.1.1.							Date Format yyyy-d	d-mm		
Aircrat	ft Details	Aircraft Reg #	VE-AD-1					Aircraft Model # A 380			
		Model Description						Model Type A310-3	100		
		Sub Fleet #	-					Customer #	00		
Mainte	enance Program Details	SUD Fleet #	0100					Customer #			
( (	1 -7/7 <b>→ →→ + -</b> □		T <sub>x</sub>				노비명	x፼¤₽⊨Ⅲ	All	•	
	Maintenance Program # 🔎	Revision #	Status	Program Category	Association Stat.	is Associated By		Associated Date	Primary Program?	Program Des	s.
	1348	3	Fresh		Active	V DMUSER		2016-06-04	Yes	✓ program	
	1348	2			Fresh	*			Yes	*	
	6YJME_AMP_1				Fresh	*			Yes	*	
	737-200 SAMPLE				Fresh	<b>v</b>			Yes	*	
	767-200SPEC	1			Fresh	<b>~</b>			Yes	*	
	A330 (A-CHECK)	8			Fresh	*			Yes	*	
	A330-C CHECK				Fresh	*			Yes	*	
					Fresh	*			Yes	<b>~</b>	

Figure 4.11 Associating multiple programs to an aircraft

- 5. Specify the association status of the program to the aircraft in the Association Status field.
  - Set the field to **Fresh** if you are associating the program to the aircraft for the first time.



- Set the field to **Active** to activate the association of the program to the aircraft.
- Set the field to **Inactive** to inactivate the association of the program to the aircraft.
- 6. Set the **Primary Program?** field to **Yes** to set the program as the primary program for the aircraft. Otherwise, set the field to **No**.
- 7. Click the Associate Programs pushbutton to associate the maintenance programs to the aircraft.
  - Note: The system associates the aircraft to the maintenance program and copies the work units defined for the program along with its schedules and all the resources defined for carrying out the program.

To provide further details,

 $\tau$  Select the Edit Aircraft Specific Maint. Program link to edit the program details.

# 4.3.3 Modifying the maintenance program of an aircraft

1. Select the Edit Aircraft Specific Maint. Program link in the Associate Aircrafts to Maintenance Program page. The Edit Aircraft Specific Maintenance Program page appears. *See Figure 4. 12* 

* 1	Ed	it Aircraft	Specific N	Maintenance Pro	gram								4 1 ▶ ≫	RAMCO OU-ramco role	- x ∈	• 🛱 🗲	? [
													Date Format mm	ı-dd-уууу			
Air	craft De	Idits		Aircraft Reg #	10 101								rcraft Model # A32	0.011			
				Sub Fleet #									nt. Operator #	0-211			
				Sub Heet #	55-101							Pidi	ic operator #				
Mai	intenan	ce Program	Information														
			Ma	intenance Program #									Revision # 5				
				Program Desc.	prog1								Status Act	ive			
- Mai	intenan	ce Program	Details														
-	4	1 - 10/14	• • 1	* * *	<b>T</b> 7						2 🛛	) × C 🗙	Ŧ∋∎ m ti %	All	<ul> <li>Searce</li> </ul>	h	Q
#		MPD Group	Work U	Jnit #	O Pr	og. Item Type		Default Exe. Priority		Initiated/ Reset by		Maint. Operator	# Parent Item #	Execution Type		Job Type	
1	E		✓ 3-OPER-	000000001	Blo	ck	~		~	Self Compliance	~	CA		Major	~	Aircraft	
2			V 3-OPER-	06052013-01	Nor	n-Block	~		~	Self Compliance	~			Major	~	Aircraft	
3			✓ 3-OPER-	06052013-02	Nor	n-Block	~		×	Self Compliance	~			Major	~	Aircraft	
4			V 3-OPER-	06052013-03	Nor	n-Block	~		v	Self Compliance	v			Major	~	Aircraft	
5			V 3-OPER-	06052013-04	Nor	n-Block	~		¥	Self Compliance	×			Major	~	Aircraft	
6			✓ test-098		Nor	n-Block	~	AOG	v	Self Compliance	v			Minor	~	Aircraft	
7			✓ 0000-73	7-0006476	Eve	ent Driven	~		¥	Self Compliance	¥		MB85	Major	~	Aircraft	
8			✓ 0000-73	7-0006477	Eve	ent Driven	~		¥	Self Compliance	~		MB85	Major	~	Aircraft	
9			✓ 0000-A3	2-0006473	Eve	ent Driven	~		¥	Self Compliance	¥		MB85	Major	~	Aircraft	
10			✓ 0000-A3	2-0006474	Eve	ent Driven	~		¥	Self Compliance	v		MB85	Major	~	Aircraft	
		<															>
Re-Nu	umber							G	et B	ase Task							
/iew Ta	ısk					View Task (	Card					View	MMM Reference				
App	proval D	etails —															
				Authority AG54		•								Description		dette	
				proval #										Date		1111 1	
			Oth	er Details													
		(	Edit Mainten	ance Program				Confirm Maintenar	nce	Program				Cancel Maintenance	Program		
		formation						alue for Aircraft					eferences for Aircraf				
	-	evel Reference						ummary Information					in Task Relationship				
nitialize	e Mainter	nance Program	n			Upload Doc	uments					View	Associated Doc. Atta	chments			

Figure 4.12 Editing aircraft specific maintenance program

- 2. Select the Maintenance Planning Document (MPD) Group from the MPD Group drop-down list box.
- 3. Enter the task, standard procedure; aircraft work package or component work package to be performed as part of the maintenance program in the **Work Unit #** field.

🔌 Note: You can include only one instance of a task in an aircraft maintenance program

- 4. Select any one of the following valid Program item types from the **Prog. Item Type** drop-down list box:
  - Block Indicates the task contains a group of tasks to be executed based on a schedule
  - Base Indicates the task to be executed is associated to a block.

- Perpetual Indicates the task is to be executed while executing any other tasks on the aircraft.
- As required Indicates the task will be executed whenever required.
- Event driven Indicates the task will be executed based on the occurrence/ triggering of an event
- Non-block Indicates the task to be executed is not associated to a block.
- To be decided Indicates the classification of the program item type for the task will be decided later.
- 5. Select the default execution priority from the **Default Exe. Priority** drop-down list box.
- 6. Select any one of the following options in the Initiated/Reset by drop-down list box:
  - Self-Compliance Indicates the task would be initiated/ reset on compliance of the same task. This implies that the compliance of this task is not dependent on any other task.
  - Related Task Compliance Indicates the task, on compliance, would initiate/ reset another task available as part of the same program. The task, which has to be initiated on compliance of a task is based on the "Initiate Schedule" relationship defined for that task in the 'Maintain Task Relationship' activity under the "Maintenance Task" business component.
- 7. Set the **Execution Type** field to **Major** to indicate that the work unit is a major check and must be sent to the hangar for maintenance. Set the **Execution Type** field to **Minor** to indicate that the work unit must be executed as a part of line maintenance.
- 8. Set the **Job Type** field to **Aircraft** to indicate that the work unit must be performed on the aircraft. Set the **Job Type** field to **On-wing** to indicate that the work unit must be performed on components attached to the aircraft.
  - Note: If the job type is "Aircraft" or "On-wing" and if the work unit type is "Aircraft Work Package", then ensure that the work unit number is valid for the specified aircraft model.
- Specify the type of maintenance job that must be carried out as a part of the work unit in the Maintenance Type field. The type of the maintenance can be Repair, Overhaul, Inspection or Others.
- 10. Enter the position code in the aircraft where the work unit must be performed in the **Position Code** field.
- 11. The number identifying the level at which the position code occurs in the configuration class attached to the primary model in the **Level Code** field.
- 12. Enter the engineering document as a part of which the work unit was defined along with its revision number in the **Eng Doc #** and the **Revision #** fields.
- 13. Specify the status of the schedule in the **Schedule Status** field.
  - Set the status of the schedule to **Fresh** when the work unit is first specified for the program.
  - Set the status of the schedule to **Active** if you wish to activate the work unit.



- Set the status of the schedule to **Inactive** if you do not wish to carry out the work unit as a part of the maintenance program.
- Set the status of the schedule to **Terminated**, to terminate the work unit.
- 14. Set the Long Term? field to Yes to indicate that the execution of the work unit is for long term. Otherwise, select the No option.
- 15. Specify the schedule execution of the work unit in the Sch. Exec Rule field.
  - Set the field to Earliest to indicate that when both date based and usage based schedules is defined for the work unit, the system must consider the earliest of the due dates calculated by these schedules.
  - Set the field to **Latest** to indicate that when both date based and usage based schedules is defined for the work unit, the system must consider the latest of the due dates calculated by these schedules.
- 16. Specify the type of deferment policy for deferring or postponing the work unit when work orders are been generated based on the maintenance program in the **Deferment Policy** field.
  - Set the field to **Not Allowed** if you do not wish to allow deferment of the work unit.
  - Set the field to Complete Work Unit, option if you wish to allow the deferment of the work unit along with the tasks and procedures defined for it. For example, If WU1 is a work package and its contents are T1, a task and SP1, a procedure, and if T1 is marked for deferring the whole package WU1 gets deferred.
  - Set the field to **Constituent Unit**, if you wish to allow the deferment of the work unit separately. For example, If WU1 is a work package and its contents are T1, a task and SP1, a procedure, T1 and SP1 can be deferred separately without affecting the parent work package.
- 17. Select the **Package Type** of the aircraft
- 18. Select the expense type of the work unit in the **Expense Type** field:
  - Set the field to **Revenue**, if the cost incurred on the work unit is for routine maintenance activity and must be recorded as revenue cost.
  - Set the field to **Capital**, if the cost incurred on the work unit must be recorded as a capital cost.
- 19. Enter the work center where the work unit must be performed in the **Work Center #** field.
- 20. The sequence in which the work unit should be performed as part of the maintenance program in the **Seq #** field.
- 21. Click the Get Base Task pushbutton to retrieve the base tasks for the selected block task.



- The system retrieves the tasks in the **Maintenance Program Details** multiline. These tasks would have been mapped to the selected parent Task # with the "Block Schedule<sup>17</sup> relationship defined in the "Maintain Task Relationship" activity of the "Maintenance Task" business component.
- 22. Select the regulatory authority as a reference to the maintenance program in the **Regulatory Authority** field in the **Approval Details** group box.
- 23. Enter the approval number with reference to which the maintenance program is created in the **Approval #** field.
- 24. Select the **Edit Maintenance Program** pushbutton to edit the maintenance program pertaining to the aircraft.
- 25. Click the **Re-Number** pushbutton to rearrange the existing rows in the ascending order of the sequence number.

To confirm the maintenance program,

- 26. Click the Confirm Maintenance Program pushbutton to confirm the maintenance program.
  - Note: The system confirms the maintenance program and sets the status of the program to "Confirm". Further modifications to the program are not allowed. On confirmation, the system changes both the 'Association Status' and the 'Schedule Status', from "Fresh" to "Active".

To cancel the maintenance program,

- 27. Click the Cancel Maintenance Program pushbutton to cancel the maintenance program.
  - Note: The system cancels the maintenance program and sets the status of the maintenance program to "Cancel".

To provide further details,

- $\tau$  Select the **Maintain Task Relationship** link at the bottom of the page to maintain relationship between a task and the related tasks.
- $\tau$  Select the **View Aircraft Schedule Summary Information** at the bottom of the page to view aircraft schedule summary information.
- $\tau$  Select the **Edit Schedule Information** link to specify the schedule information for the maintenance program.
- $\tau$  Select the **Edit Schedule Dates & Value for Aircraft** link to edit the scheduled dates and values for the maintenance program.
- $\tau$  Select the **Edit References for Aircraft Program** link to specify reference document details based on which the maintenance program was created.
- τ Select the View Program Level References link to view the reference documents specified for the maintenance program.
- $\tau$  Select the **Upload Documents** link to upload documents associated with the maintenance program of the specific aircraft.

# 4.3.4 Modifying the schedules for an aircraft specific maintenance program

1. Select the Edit Schedule Information link in the Edit Aircraft Specific Maintenance Program page. The Edit Aircraft Maint. Schedule Information page appears. *See Figure 4. 13* 



- 2. Select the work unit for which schedule information must be specified in the **Work Unit #** field in the **Work Unit Details** group box.
- 3. Click the **Get Details** pushbutton.
- 4. Specify whether the work unit must be executed only once or repeated at certain intervals in the **Schedule Type** group box in the **Schedule Details** group box.
  - Set the field to **One Time** if the work unit must be executed only once.
  - Set the field to **Recurring** if the work unit must be repeated at certain defined intervals.
- 5. Select the mode by which the due dates for executing the work units must be calculated in the **Update Basis** field.
  - Schedule If the option is selected as "Scheduled", then NSD/NSV (Next Schedule Date/ Next Schedule Value) for executing the work units will be calculated based on a scheduled fixed interval.
  - Actual Completion If the option is selected as "Actual Completion", then NSD/ NSV will be calculated by adding the Repeat Fixed / Floating Interval value to the date or value at which the work unit is actually executed.
  - Time Window This option is applicable for the following conditions:
    - Within the Tolerance limits: If the option is selected as "Time Window", then NSD/NSV will be calculated based on a scheduled fixed interval
    - Beyond the Tolerance limits: If the option is selected as "Time Window", then NSD/NSV will be calculated by adding the Repeat Fixed / Floating Interval value to the date or value at which the work unit is actually executed.

To define date-based schedules,

- 6. Specify the time unit for the date-based schedule in the **Time Unit** field in the **Date Based Schedule Details** group box. The time unit can be **Days, Months, Month End** or **Years**.
- 7. Enter the period from the induction date or the operational date of the aircraft, for which the due dates for the work units must be calculated, in the **Threshold Interval** field.
- 8. Enter the period at which an alert message must be displayed about an impending due date for performing the work unit in the **Alert Interval** field.
- 9. Enter the interval after which the work unit must be repeated irrespective of whether its previous execution is completed or not in the **Repeat Fixed Interval** field.
- 10. Enter the interval after which the work unit must be repeated, from the period of its last execution and completion in the **Repeat Floating Interval** field.
  - Note: Ensure that only one of the fields, "Repeat Fixed Interval" or "Repeat Floating Interval", is entered. Data entry in the "Repeat Fixed Interval" or the "Repeat Floating Interval" fields is mandatory, if you are defining date-based schedules and if you have set the "Schedule Type" field to "Recurring".
- 11. Enter the maximum time period subsequent to the next schedule date, within which the work unit must be performed, in the **Positive Tolerance** field. Enter the maximum time period ahead of the next schedule

date, within which the work unit could be performed, in the Negative Tolerance field.

12. Enter the period reaching which the work on the work unit must be terminated in the **Terminating Interval** field.

Edit Aircraft Maint. Schedule Information						ē ⊄ ← ?	Ľø.
					Date Format yyyy-dd-m	m	
Aircraft Details     Aircraft Reg #     Sub Fleet #				Aircraft Model # (	ESSNA 173		
Maintenance Program Information     Maintenance Program #     Program Desc.     Work Unit Details	CESSNA 173 NEW CESSNA 173 ew			Revision # Status F	resh		
Work Unit # Work Unit Description Execution Type		Get Details		Work Unit Type 1 Job Type 4			
Schedule Details     Schedule Type     Date Based Schedule Details	Recurring 🔻			Update Basis	Actual Completion		
Time Unit Threshold Interval Repeat Fixed Interval Positive Tolerance Terminating Interval		1.00		Alert Interval Repeat Floating Interval Negative Tolerance	11.0	0	
	T			x 🖂 🖄 📽 📮 🖷 🖩	Al	v	Q
# 🖻 Parameter 👂 UOM		Repeat Fixed Interval	Repeat Floating Interval	Alert Interval	Positive Tolerance	Negative Tolerance	
1 APUC 2 APUH 3 DO 4 DM3 5 Schedules for depend work units.		5000.0	chedules	4500.0	0 2000.00		>
Edit Schedule Dates & Values for Arcraft		Eurs	ALL CALLED				

Figure 4.13 Editing aircraft maintenance schedule information

To define usage-based schedules,

- 13. Enter the parameter based on which the work unit must be performed in the **Parameter** field in the **Usage Based Schedule Details** multiline.
- 14. Enter the unit of measurement of the parameter in the **UOM** field.
- 15. Enter the parameter value at which the first execution of the work unit is due in the Threshold Value field.
- 16. Specify the interval after which the work unit must be repeated irrespective of the completion of its last execution in the **Repeat Fixed Interval** field.
- 17. Specify the interval after which the execution of the work unit must be repeated, from the period when it was last executed in the **Repeat Floating Interval** field.
  - Note: Ensure that only one of the fields, "Repeat Fixed Interval" or "Repeat Floating Interval", is entered. Note: Data entry in the "Repeat Fixed Interval" or the "Repeat Floating Interval" field is mandatory if you are defining usage-based schedules and if you have set the "Schedule Type" field to "Recurring".
- 18. Enter the parameter value at which an alert message would be displayed about an impending work due on the work unit in the **Alert Interval** field.
- 19. Enter the upper limit of the parameter value greater than the next schedule value, within which the work unit must be performed, in the **Positive Tolerance** field.
- 20. Enter the lower limit of the parameter value lesser than the next schedule value, within which the work unit could be performed, in the **Negative Tolerance** field.
- 21. Specify the parameter value, reaching which the work on the work unit must be terminated in the **Terminating Value** field.
- 22. Click the Edit Schedules pushbutton.



Note: The system saves the schedule details against the maintenance program.

To provide further details,

 $\tau$  Select the **Edit Schedule Dates & Values for Aircraft** link to edit the details of the scheduled date or values for the work unit in the program.

# 4.3.5 Modifying the scheduled dates and the values for the work units in the maintenance program

- 1. Select the Edit Schedule Dates & Value for Aircraft link in the Edit Aircraft Specific Maintenance Program or the View Aircraft Schedule Summary Information page. The Edit Schedule Date/Value page appears. *See Figure 4. 14*
- 2. Select the work unit for which the schedule dates and values must be specified in the **Work Unit #** field in the **Work Unit Details** group box.
- 3. Click the **Get Details** pushbutton.

To modify the date-based schedule details,

- 4. Enter the period from the induction date or the operational date, for which the due dates for the work units must be calculated, in the **Threshold Date** field in the **Date Based Schedule Details** group box.
- 5. Enter the date on which the work unit was last performed in the Last Performed Date field.
- 6. Enter the next scheduled date of the work unit in the Next Scheduled Date field.
- 7. Enter the date on which the work unit must be terminated in the Terminating Date field.
- 8. Use the **Change Type** drop-down list box and select the value 'Correction' or 'Re-baseline' to indicate whether NSD can be modified on modification of Threshold or interval.
  - Note: The system displays the computation basis controls such as Next Due Calc. On indicating computation basis of NSD, Calc. Ref. Date (reference date based on which NSD is computed) and Last Schedule Date.

★ 🗎 Edit Schedule Date/Value					≡ ≭ ≣		2		
Aircraft Details			Date	e Format yyyy-dd-mm					
Aircraft Reg #	VA-TE-6		Aircraft	Model # CESSNA 173					
Sub Fleet #	105								
Maintenance Program Details  Maintenance Program #	CESSNA 173 NEW		Re	evision #					
Program Desc.				Status Fresh					
- Work Unit Details									
Work Unit #	EO-000608-2016-4 The	type of dependent	Prog. It	em Type Non-Block					
Work Unit Type	acho	dule information	Work Unit De	scription sdf					
Execution Type     Schedule Details									
Schedule Type	One time	ed for the	Upda						
Sch Exec Rule	Earliest depe	endent work units	Initiate/F						
Date Based Schedule Details					starts.				
Last Performed Date	2016-01-05 00:00:00		Terminat	ing Date 2016-01-05 23:59	:59 🗰				
Earliest Schedule Date	2016-01-05 00:00:00		Latest Sched	:59 📖					
Alert Date				m. Days -11D					
Remarks	7			ige Type	•				
Next Due Calc. On	Eng.Doc. : Effec. From Da		Calc. Ref. Date 2016-20-04 00:00:00						
Last Schedule Date									
Usage Based Schedule Details									
44 4 1 -1/1 >>> + - 0 % 0 0 T	τ.		人画句文図画文の	# ₩ III All	<b>•</b>		Q		
# Parameter Unit Of Measurement Threshold Value		Earliest Latest Schedule			Calc. Ref. Value	Last Schedule			
1 🗉 FH	11.00		11.00 En	g.Doc. : Effec. From Date	0.00	Not Applicable			
2									
Parameters available f	or dependent								
and independent sche									
							>		
work unit are listed ser	Darately.								
		Edit Schedules							

Figure 4.14 Modifying scheduled dates and values for work units in the maintenance program

To modify the usage-based schedule details,

- 9. Enter the parameter value at which the first execution of the work unit is due in the **Threshold Value** field in the **Usage Based Schedule Details** multiline.
- 10. Enter the last parameter value at which the work unit must be performed in the **Last Performed Value** field.
- 11. Enter the parameter value at which the work unit is next due to be executed in the **Next Schedule Value** field.
- 12. Enter the parameter value, reaching which the work on the work unit must be terminated in the **Terminating Value** field.
- 13. Use the **Change Type** drop-down list box and select the value 'Correction' or 'Re-baseline' to indicate whether NSV can be modified on modification of Threshold or interval.
  - Note: The system displays Next Due Calc. On indicating computation basis of NSV, Calc. Ref. Value (reference date based on which NSV is computed) and Last Schedule Value.
- 14. Click the **Edit Schedules** pushbutton.
  - Note: The NSD / NSV computation basis controls are updated based on selection of 'Change Type' and values specified by user, on modifying the schedule details.

# 4.3.6 Inactivating the association of an aircraft to a maintenance program

#### Inactivate the association through the "Associate Aircrafts to Program" activity

- 1. Select the Associate Aircrafts to Program link under the Aircraft Maintenance Program business component.
- 2. The Select Maintenance Program page appears.
- 3. Enter the program to which aircrafts must be dissociated, and select the Associate Aircrafts to Maintenance Program link provided alongside.

Or

- 4. Search for the program and click the hyperlinked program number in the multiline. The **Associate Aircrafts to Maintenance Program** page appears.
- 5. The system retrieves the details of the aircraft already associated to the program in the Aircraft Model Details multiline.
- 6. Set the Association Status field to "Inactive", to inactivate the association of the aircraft to the program.
- 7. Click the **Associate Aircraft** pushbutton to inactivate the association of the aircraft.

#### Inactivate the association through the "Associate Programs to Aircraft" activity

- 1. Select the Associate Programs to Aircraft link under the Aircraft Maintenance Program business component. The **Select Aircraft** page appears.
- 2. Enter the aircraft registration number of the aircraft to which programs must be dissociated and select the **Associate Maintenance Programs** link provided alongside.

Or

- 3. Search for the aircraft and click the hyperlinked aircraft registration number in the multiline. The **Associate Maintenance Programs to Aircraft** page appears. *See Figure 2.24.* The system retrieves the details of the programs already associated to the aircraft in the **Maintenance Program Details** multiline.
- 4. Set the Association Status field to "Inactive", to inactivate the association of the program to the aircraft.
- 5. Click the **Associate Programs** pushbutton, to inactivate the association of the program to the aircraft.

1. Select the **Edit References** for Aircraft Program link in the Edit Aircraft Specific Maintenance Program page. The Edit References for Aircraft Program page appears. *See Figure 4. 15* 

	📱 Edit References for Aircraft Program 🔤 🕫 🖶 🗧												
-	ircra	ft Details											
			Aircrat	ft Reg # VA-TE-6		Aircraft I	Model # CES	SSNA 173					
		enance Program Details		Fleet # 105									
	Idiliit	enance Program Details		ogram # CESSNA 173 NEW		Re	vision #						
				m Desc. CESSNA 173 ew			Status Fre	sh					
- <b>- R</b>	efer	ence Document Details								_			
44	-	1 - 1 / 1 > >> +	* •				₽ ₩ 000					Q	
#		iter botanert i jpe		Ref. Document #	Document Name		File Name 🔎						
1	E			ARR 1001	References Doc			Trn-11A112AV-T'way-Maintenance Program.ppt					
2			*										
		<										>	
					Edit References								
View F	ile												

Figure 4.15 Editing the reference details for the aircraft maintenance program

- 2. Specify the type of the reference document for which the maintenance program is created in the **Ref. Document Type** field in the **Reference Document Details** multiline.
- 3. Enter the reference document number in the **Ref. Document #** field.
- 4. Click the Edit References pushbutton.
  - Note: The system saves the details of the reference documents against the maintenance program.



# 4.4 Activating a maintenance program

1. Select the Activate Aircraft Maint. Program link under the Aircraft Maintenance Program business component. The Activate Maintenance Program page appears. *See Figure 4. 16.* 

k 🗎 /	Activate Maintenan	ce Program	≣ ≭ 帚 ₽ ← ? G									
Search	Criteria											
- Search	Results	Maintenance Program # Program Desc. Aircraft Reg #	Search		Status Confirm  Program Category Aircraft Model #							
	1 - 6 / 6 🕨 👐 🕂											
# 8	Maintenance Program #	Revision #	Status	Maintain Prev. Rev.?	Program Desc.							
1	A330 (A-CNECK)	9	Confirm	Yes	<ul> <li>Airbus Maintenance Program</li> </ul>							
2	BOEING	1	Confirm	Yes	✓ BO							
3	CESSNA 173-1	Click here to modify	Confirm	Yes	✓ CESSNA 173-1							
4 🗉		Click here to modify	Confirm	Yes	✓ ER-002-Program							
5	MAS	the maintenance	Confirm	Yes	Maint. program for Malaysian Airlines							
	MP_001_16	program details	Confirm	Yes								
7				retain the	eld to "No" if you do not wish to previous revisions of the program ing the latest revision							
	<					>						
	Acti	ivate Program	Inactivate Program		Return Program							
ew Program	1 Information		Compare Progra	am Revisions								

Figure 4.16 Approving the maintenance program

- 2. Search to retrieve the maintenance programs to be approved, in the multiline.
  - Note: The system retrieves the maintenance programs only for those aircraft for which the record status is not "Frozen".
- 3. Set the **Maintain Prev Rev.?** field in the multiline to **Yes** if you wish to maintain all the previous versions of the maintenance program when this version of the program is activated. Otherwise, set the field to **No**.

To activate maintenance programs,

- 4. Select the programs to be activated and click the Activate Program pushbutton to activate the programs.
  - Note: Ensure that the selected programs are not already in the "Active" or "Inactive" status. The system activates the selected programs and sets the status of the programs to "Active". If a previous revision of the program exists in the "Active" status, the system sets the status of the previous revision to "Revised". In addition, if the "Maintain Prev Rev.?" field is set to "Yes" for the selected program, the system maintains all previous revisions of the programs on activating the selected revision.

To inactivate maintenance programs,

- 5. Select the programs to be inactivated and click the **Inactivate Program** pushbutton.
  - Note: You cannot inactivate the maintenance program, which is already in "Inactive" or "Confirmed" status.

The system sets the status of the maintenance program to "Inactive".

To return maintenance programs,

- 6. Select the programs to be returned and click the **Return** pushbutton.
  - Note: Ensure that the maintenance program is in the "Confirm" status. The system sets the status of the maintenance program to "Return". The returned program can be modified further in the "Edit Aircraft Maint. Program" activity and confirmed for activating it at a later date.

To compare various revisions of a program,

τ Select a specific revision of a program and select the **Compare Program Revisions** link to compare the selected revision of the maintenance program with other revisions.

### 4.4.1 Comparing the revisions of a maintenance program

You can compare the various revisions of a maintenance program. You can select a specific revision of the maintenance program and compare it with the other revisions of the program.

The following comparison details are retrieved:

- The level at which the revision change has taken place, whether it is at the program level or at the aircraft specific program level.
- The entity due to which the revision has taken place. It could be either the changes in the work units or to the aircraft associated to the program.
- The value of the entity that was affected.
- The action due to which the program was revised.
- The type of information that was affected due to the revision.
- 1. Select the Activate Aircraft Maint. Program link under the Aircraft Maintenance Program business component. The Activate Maintenance Program page appears. *See Figure 4.21.*
- 2. Use Search Criteria to retrieve the maintenance program, that you wish to compare with other versions.
- 3. Select a specific revision of a program and then click **Compare Program Revisions** link to compare the selected revision of the maintenance program with other revisions. The **Compare Program Versions** page appears. *See Figure 4. 17*

	aintenance Program Deta	ile			Date	e Format yyyy-dd-mm				
	aintenance Program Deta	Maintenance Program # A330	(A-CHECK)		Re	evision # 9				
		Program Desc. Airbus	s Maintenance Program			Status Confirm				
		Compare with Rev #		aile						
C	mparison Details		occoca							
	<pre></pre>	T T,				. # # Ⅲ AI 💌				
#	Level	Entity Affected	Affected Entity Value	Aircraft Reg #	Action	Information Affected				
L	PROGRAM	Task	1-1-50C-2000-CMM-00005049		ADDED	BASIC CONTENT				
2	PROGRAM	Task	1-1-A330-0000-9X-00005051		ADDED	BASIC CONTENT				
3	PROGRAM	Task	1-1-A330-0000-9X-00005052		ADDED	BASIC CONTENT				
4	PROGRAM	Task	1-1-A330-0000-AD-00005053		ADDED	BASIC CONTENT				
5	PROGRAM	Task	1-1-A330-0000-CMM-00005050		ADDED	BASIC CONTENT				
5	PROGRAM	Task	1-1-A330-0000-CMM-00005054		ADDED	BASIC CONTENT				
7	PROGRAM	Task	1-3-A33-00-MPD-14579		ADDED	BASIC CONTENT				
3	PROGRAM	Task	1-Aircraft EO Task		ADDED	BASIC CONTENT				
9	AIRCRAFT	MODEL	A330		ADDED	MODEL EFFECTIVITY				
10	AIRCRAFT	AIRCRAFT CODE	RPT-1	RPT-1	ADDED	AIRCRAFT ASSOCIATION				
	<									

Figure 4.17 Comparing the revisions of a program

The system displays the specific revision of the maintenance program that must be compared in the **Maintenance Program #** and the **Revision #** fields in the **Maintenance Program Details** group box.

- 4. Select the specific revision number of the program to which the selected maintenance program revision must be compared, in the **Compare with Rev #** field in the **Maintenance Program Details** group box.
- 5. Click the **Get Details** pushbutton.

The system retrieves the following details of the revision in the **Comparison Details** multiline.

Value displayed	Indicates
Program	Revision details are displayed for the
	maintenance program
Aircraft	Revisions details are displayed for the
	maintenance program associated to a
	specific aircraft

**Column 2 - Entity Affected -** The type of entity that is affected by the revision.

Value displayed	Indicates
Task / Procedure / Aircraft Work	The type of the work unit in the
Package / Component Work Package	maintenance program, which has undergone revision
Aircraft	Aircraft effectivity details of the
	program have been modified

Column 3 - Affected Entity Value - The entity that is affected by the revision.

Value displayed	Indicates
Work unit number	Changes have been made to the work units of the maintenance program
Work unit number along with its schedule ID	Schedules are defined for the work units in the program
Work package number along with the work unit number and its schedule ID Aircraft registration number	The work unit has been added or deleted from the work package Associated or dissociated from the program
Aircraft model number	Added or deleted from the model effectivity details of the program

**Column 4 - Aircraft Reg # -** The registration number of the aircraft, if the revision change is due to the change in the aircraft effectivity.

Column 5 – Action - The action performed, which has resulted in the revision.

Value displayed	Indicates
Added / Activated / Inactivated /	Actions performed on the work units
Terminated	defined as a part of the program
Added / Deleted / Updated	Actions performed on the dependent
	work units in the program
Updated	Changes made to the scheduled
	details of the work units as well as to
	the dependent scheduled details for
	the work unit in the program
Added/Deleted	Changes made to the work units that
	are part of the work packages
	specified for the program
Added / Activated / Modified	Changes made to the aircraft
	associated to the program
Added / Deleted	Changes made to the aircraft models
	associated to the program

**Column 6** - **Information Affected -** The type of information that is affected due to the revision.

Value displayed	Indicates
Basic Content	Changes made to the work units
	specified for the maintenance
	program
Dependency Definition	Dependent work units are added,
	deleted or updated
Schedule Definition	Changes made to the schedules that
	are defined for the work units
Dep Schedule Definition	Changes made to the dependent
	schedules
Work Content / Seasonal Variations	Work units in the work packages have
	undergone change, or when seasonal
	work units have been defined
Aircraft Association	Aircrafts have been associated or
	dissociated from the program
Model Effectivity	Changes made to the model
	effectivity details of the program

# 4.5 Revising a maintenance program

You can revise an aircraft maintenance program that is in the "Active" or "Inactive" status. You can modify the program and create a new revision in the "Edit Maintenance Program" page. The status of the newly created revision of the program is set to "Fresh". On confirmation and activation of the newly created revision, the status of the previous revision, which is in the "Active" status, is set to "Revised".

# 4.5.1 Revising the program

1. Select the Revise Aircraft Maint. Program link under the Aircraft Maintenance Program business component.

The Select Maintenance Program page appears.

2. Search for the program and click the hyperlinked program number in the multiline. The Edit Maintenance **Program** page appears. *See Figure 4. 18.* 

* [	🛨 🗎 Edit Maintenance Program										41 4 1 2 3 4 5 🕨 🖬 1 /20 📰 🕮 🛱 🗲 ? 🗔								
Date Format     yyyy     Date Format											yyyy-dd-mm	,							
	Maintenance Program # 6YJME_AMP_1											Revi	sion #	1	_	The revision	on		
				Program	n Desc.	AIRCRAFT MAI	NTENANC PR	OGRAN						Status	Inactive	<u> </u>	number of		
				Program Ca		•							Primary M	odel #	A320-211				
				Configuratio	n Class	AI-707 💌					Model Effectivity Single program is retriev						retrieved		
Wo	ork S	cope Details	5																-
		1 - 4 / 4	h hh	+-0**	Q T	T,					7	10 V V	🗎 😒 C 👎	9 U	I All		<b>v</b>		ρ
#		MPD Group		Work Unit # 🔎	Prog.	Item Type		Default Exe	Priority	Initiated/ Reset by		Parent Item #	Execution Type		Job Type		Maintenance Type	Position Code 🔎	
1			*	36 HOUR CHECK	Base		*	AOG	*	Self Compliance	*		Major	*	Aircraft	*	*		
2			~	A-CHECK	Block			NRM	*	Self Compliance	*		Major	*	Aircraft	*	~		
3			~	DAILY CHECK	As Rec	quired		WS1	*	Self Compliance	*		Major	*	Aircraft	¥	*		
4			~					A1	*	Self Compliance	*		Major	~	Aircraft	•	*		
5			~				*	Vlow	*	Self Compliance	*		Major	*	Aircraft	*	~		_
		<																>	۲.
Re-S	eq #									Get Ba	ase Task								
Edit Sch	edule	Information					Maint	tain Task Rela	ationship				View Task						
	DEON	al Details —																	
	prov		atory A	uthority	Ŧ												Description		
				oroval #													Date		ß
			Other	r Details															
- Ot	her D	Details																	
				Revision Cor	nments														
							🗹 Сору	to Associated	d Aircrafts										
			Edit N	laintenance Program					Confir	rm Maintenance Progr	ram				Cance	el Mai	intenance Program		
Edit Air	raft I	Model Effectiv	ity				Asso	ciate Aircraft	s				Edit Maintenar	nce Ever	nt Informatio	'n			
Edit Re	ferenc	ces					Uploa	ad Documents	;				View Associate	ed Doc.	Attachments	;			
- Re	cord	Statistics –																	
				Crea	ated by	DMUSER							Created	d Date	2014-05-06				
				Last Mod	fied by	DMUSER							Last Modified	d Date	2014-05-06				
					Source								(	Owner					

Figure 4.18 Revising maintenance program

You can modify the details of the maintenance program in this page. Refer to the topic "Creating aircraft maintenance program" for more details.

# Index

### A

Access Panel #, 11, 29, 30 Access Panel Description, 11 Access Panel Task #, 9, 10 creating, 9 specifying resources, 10 associating access panels, 11 specifying parts, 10 Action, 99 Activate Program, 96 Activating tasks, 33 Additional Repair Details, 45 Affected Entity Value, 98 Aircraft Maintenance Program status flow, 72 Aircraft Model #, 9, 11, 28, 30, 84 Aircraft Reg #, 98 Alert Interval, 81, 82, 91, 92 AP Task Description, 10 AP Task Type, 10 Applicability, 28 Approval #, 90 Approval Regd, 10, 18 Approve & Activate Maintenance Program, 96, 97 Associate Aircrafts to Program, 85 Associate Programs to Aircraft, 86 Associate work units on opportunity check, 62 Associating access panels to AP task, 11 Association Status, 86, 87 Active, 85, 87 Fresh, 85, 86 Inactive, 85, 87 ATA #, 39 Authorize component maintenance program, 62 part maintenance process, 59 position based schedule, 65

#### В

Base Model #, 45

### С

Cancel Maintenance Program, 90 Child Mfr. Part #, 45 Child Part #, 45 Child Position #, 45 Close Access Panel Task #, 11 Comparing the revisions of a maintenance program, 97 Component effectivity, 29 Component maintenance program, 52 authorize, 62 opportunity check, 62 Configuration Class, 76 Confirm Maintenance Program, 90 Copying part maintenance program to multiple parts, 58 Create Maintenance Program, 76 Create component maintenance program, 59 date based schedule, 63 part maintenance, 54 position based schedule, 63 usage based schedule, 64 Creating an aircraft maintenance program work package, 76 Creating standard task from non-standard task, 16

# D

Deactivating a task, 40 Default Applicability, 28 Default Numbering Type, 13 Deferment Policy, 79, 89 Define aircraft effectiveness, 65 date based schedule, 57 usage based schedule, 57 Defining maintenance event, 74 Dependent System Condition, 32 Dependent System, 32 DSC #, 32 DSC Description, 32

### Ε

Edit Aircraft Model Effectivity Information, 84 Edit Aircraft Specific Maint. Program, 87 Edit Maintenance Event Information, 82 Edit Maintenance Program, 90, 100 Edit References for Aircraft Program, 95 Edit References, 83 Edit Schedule Date/Value, 93 Edit Schedule Information, 80 Editing aircraft mod #, 28 Effectivity Control, 44 Eng. Doc#, 78, 88 Entity Affected, 98 Est. Elapsed Time, 39 Est. Man Hrs., 39 Execute With, 62 **Execution Facility**, 45 **Execution Phase**, 14 Execution Type, 78, 88 Expense Type, 79, 89

# G

Generate Non-Standard Tasks, 12

# I

Ideal Time, 14 Identifying maintenance events, 82 Inactivate Program, 97 Inactivating the association of an aircraft to a maintenance program, 94 Information Affected, 99

# J

Job Type, 78, 88

### L

Last Performed Value, 94 Level, 98

### Μ

Maint. Object #, 44 Maint. Type, 45 Maintain Prev Rev.?, 96 Maintaining activated tasks, 39, 41 Maintaining repair scheme, 43 Maintenance Event Defining, 74 identifying, 82 maintenance program approving, 96 Maintenance Task, 13 Maintenance Type, 60, 78, 88 Managing repair scheme definition, 42 Managing unified task sequence, 34 Managing unified task sequence, 34 Mapping parts to sequence control, 35 tasks to sequence control, 35 Mfr. #, 45 Model effectivity, 10, 27 MPD Item #, 15 Multiple Model Effectivity, 12

# Ν

Negative Tolerance, 81, 82, 92 Next Schedule Value, 94 Non-standard task converting to standard task, 16 Numbering Type Non-Standard Tasks, 12

# 0

On Wing, 61 Open Access Panel Task #, 11 Operations Type, 13 Operator Code, 44

# Ρ

Package Category, 73 Parameter, 19, 57, 82, 92 specifying details for task/sub-task, 31 Part #, 17, 29 Part program Copying to multiple parts, 58 Creating part maintenance program, 54 Defining usage based schedule, 57 Defning date based schedule, 57 Parts for access panel tasks, 10 for task, 17 Position based schedule Activating a position based schedule, 65 aircraft effectivity, 65 Date based, 63 Usage based, 64 Position Code, 78, 88 Positive Tolerance, 81, 82, 92 Primary Model #, 76 Primary Program ? , 85, 87 Prime Part Group Appl.?, 45 Program Category, 73

# Q

Quick Code Type, 73 Quick codes, 11, 73 Component maintenance program, 53

# R

Re-activating a task, 40 Ref. Document Type, 83, 95 Reference Date Basis, 81 Reference Doc Type, 30 Regulatory Authority, 79, 90 Rel. Task Lag (Days), 45 Related Task #., 45 Repair Agency #, 45 Repair Classification, 45 Repair Scheme Details, 45 maintaining, 43 Repeat Fixed Interval, 81, 82, 91, 92 Repeat Floating Interval, 81, 82, 91, 92 Repeat Interval (Days), 19 **Re-Sequence Multiplication Factor**, 13 Re-Sequence, 37 Reset on Attachment?, 55 Resource #, 10, 18 Resource Group, 10, 50 Resource Type, 10 Resources for access panel task, 10 specifying for task, 18 Return maintenance programs, 96 Return Task, 34 **Revising a task** without a new revision, 39 Revising an activated task with a new revision, 39

Revising the program, 100 Revision #, 13, 78, 88, 97 Revision History Maintenance Policy, 74 Routing Req.?, 45 Routing Sheet Updates, 45

# S

Sch. Exec Rule, 78, 89 Schedule Relationship, 45 Schedule Status, 78, 88 Active, 78, 88 Fresh, 78, 88 Inactive, 78, 89 Terminated, 78, 79 Schedule Type, 19, 80, 91 Schedule, 19 Seq. Control #, 37 Seq. Control Desc., 37 Sequence Control Generation Mode, 13 Set Options, 73 Setting options Task, 12 Shelf Life Expiry, 61 Sign Off, 10 Sign-off information specifying for task, 18 Specifying schedules maintenance program, 80 Specifying parameter details for task/sub-task, 31 Standard task creating from non-standard task, 16 Status, 71 Active, 72 Cancel, 72 Confirm, 72 Fresh, 71 Inactive, 72 Return, 72 Revised, 72 Std. Disposition, 45 Sub Fleet Security, 74 Sub Tasks, 10, 15 Sub-task specifying parameter details, 31

# Т

Task Applicability, 13 Task Control #, 45 Task Description, 13 Task references, 30 activating, 33 authoring, 9 creating, 13 deactivating, 40 entering work area / zone information, 30 maintain activated tasks, 39, 41 quick codes, 11 reactivating, 41 returning, 34 revising, 40 setting options, 12 specifying component effectivity, 29 specifying model effectivity, 10 specifying parameter details, 31 specifying parts, 17 specifying resources/sign-off details, 18 specifying schedule, 19 specifying task references, 30 Terminating Value, 82, 92 Task execution, 19 Threshold Interval, 81, 91 Threshold Value, 82, 92

# U

UOM, 10, 93 Updating work units in existing maintenance program, 67 Updation Basis, 57, 81, 92 Actual Completion, 82 Schedule, 81

# W

WBS Code, 13, 39 Work Area #, 9 Work Area Description, 9 Work area, 30 creating, 9 Work Center #, 84, 90 Work Center, 46 Work Location Reference, 15 Work package updating existing program, 67 Work Unit #, 88, 92, 94 Work Unit Type, 84 WorkArea / Zone #, 30 Wrench Time, 14

# Ζ

Zone, 30 Zones, 9

# ramco

### **Corporate Office and R&D Center**

Ramco Systems Limited, 64, Sardar Patel Road, Taramani, Chennai – 600 113, India Office + 91 44 2235 4510 / 6653 4000 Fax +91 44 2235 2884 Website - www.ramco.com